## WS02-CFDC1-E DeviceNet Configurator 3G8F5-DRM21 ISA Board 3G8E2-DRM21 PCMCIA Card **DeviceNet Configurator** Ver. 2.

# **OPERATION MANUAL**



AUDIN - 8, avenue de la malle - 51370 Saint Brice Courcelles Tel : 03.26.04.20.21 - Fax : 03.26.04.28.20 - Web : http: www.audin.fr - Email : info@audin.fr

## WS02-CFDC1-E DeviceNet Configurator 3G8F5-DRM21 ISA Board 3G8E2-DRM21 PCMCIA Card DeviceNet Configurator Ver. 2.

## **Operation Manual**

Revised November 2001

## Notice:

OMRON products are manufactured for use according to proper procedures by a qualified operator and only for the purposes described in this manual.

The following conventions are used to indicate and classify precautions in this manual. Always heed the information provided with them. Failure to heed precautions can result in injury to people or damage to property.

- **DANGER** Indicates an imminently hazardous situation which, if not avoided, will result in death or serious injury.
- **WARNING** Indicates a potentially hazardous situation which, if not avoided, could result in death or serious injury.
- **Caution** Indicates a potentially hazardous situation which, if not avoided, may result in minor or moderate injury, or property damage.

## **OMRON Product References**

All OMRON products are capitalized in this manual. The word "Unit" is also capitalized when it refers to an OMRON product, regardless of whether or not it appears in the proper name of the product.

The abbreviation "Ch," which appears in some displays and on some OMRON products, often means "word" and is abbreviated "Wd" in documentation in this sense.

The abbreviation "PC" means Programmable Controller and is not used as an abbreviation for anything else.

## Visual Aids

The following headings appear in the left column of the manual to help you locate different types of information.

**Note** Indicates information of particular interest for efficient and convenient operation of the product.

- **Reference** Indicates supplementary information on related topics that may be of interest to the user.
  - *1,2,3...* 1. Indicates lists of one sort or another, such as procedures, checklists, etc.

## Trademarks and Copyrights

DeviceNet is a registered trademark of the Open DeviceNet Vendor Association, Inc.

Windows, Windows 95, Windows 98, Windows NT, and Windows 2000 are registered trademarks of the Microsoft Corporation.

Other product names and company names in this manual are trademarks or registered trademarks of their respective companies.

The copyright of the DeviceNet Configurator software belongs to OMRON Corporation.

#### © OMRON, 2000

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form, or by any means, mechanical, electronic, photocopying, recording, or otherwise, without the prior written permission of OMRON.

No patent liability is assumed with respect to the use of the information contained herein. Moreover, because OMRON is constantly striving to improve its high-quality products, the information contained in this manual is subject to change without notice. Every precaution has been taken in the preparation of this manual. Nevertheless, OMRON assumes no responsibility for errors or omissions. Neither is any liability assumed for damages resulting from the use of the information contained in this publication.

## **TABLE OF CONTENTS**

PRF	CA	UTIONS
	1	Intended Audience
-	2	General Precautions
-	3	Safety Precautions
2	4	Operating Environment Precautions
:	5	Application Precautions
SEC	TI	ON 1
Ove	rvie	ew
	1-1	Introduction
	1-2	Confirming Product Contents
	1-3	Operating Environment
SEC	TI	ON 2
Inst	alla	tion
	2-1	Installation Procedure
-	2-2	ISA Board Installation
	2-3	PCMCIA Card Installation
	2-4	Configurator Installation
<b>GEO</b>		
SEC		UN 3
Basi	c U	peration
-	3-1	Starting the Configurator and the Main Window
-	3-2	Menu Commands
-	3-3	Connecting to the Network
-	3-4	Operation Flowcharts
SEC	TI	ON 4
Edit	ing	Device Parameters
4	4-1	Adding Devices to the Virtual Network
4	4-2	Setting Slave Parameters
2	4-3	Adding Slaves to the Master
2	4-4	Setting Master Properties
2	4-5	Editing Master Parameters
2	4-6	Parameter Wizard
2	4-7	Master Parameter Editing Details (Tab Descriptions)
2	4-8	Manual I/O Allocations.
2	4-9	Advanced Settings (Connection, Communication Cycle Time, Slave Function Settings, Etc.)
4	4-10	Creating and Editing I/O Comments
2	4-11	Displaving Device Properties
		r 7 6

## **TABLE OF CONTENTS**

Online	Operation	10
5-1	Switching between Online and Offline	1(
5-2	Downloading the Network Configuration/Device Parameters to Devices	1
5-3	Uploading and Verifying Device Parameters.	11
5-4	Monitoring Devices	1
SECTI	ON 6	
Manip	ulating Files 1	12
6-1	Saving and Reading Files	13
6-2	EDS File Management.	13
6-3	Printing	13
6-4	Using General-purpose Tools to Set Devices	13
6-5	Exporting Data Created in Configurator	13
6-6	Importing Files Created in Version 1 of the Configurator	14
6-7	Optional Functions	14
Appen	dices	
A	Error Messages	14
В	Troubleshooting	14
С	Error History Information	15
D	Unit Replacement	1.
E	Replacing the C200HW-DRM21 on a CS-series CPU Rack with the CS1W-DRM21	1:
F	Replacing the C200HW-DRT21 with the CS1W-DRM21's Slave Functions	1
G	Dimensions of PCMCIA Card and Unit	16
Index	1	6
muca	· · · · · · · · · · · · · · · · · · ·	.0
Povisio	n History 1	16

## About this Manual:

This manual describes the installation and operation of version 2 of the DeviceNet Configurator and includes the sections described below.

Please read this manual and all manuals for related products carefully and be sure you understand the information provided before attempting to install and operate the DeviceNet Configurator. **Be sure to read the precautions provided in the following section.** 

*Section 1* Section 1 describes the features, specifications, operating conditions, and interfaces of the DeviceNet Configurator.

**Section 2** Section 2 explains how to install the ISA Board or PCMCIA Card hardware in the computer, make the necessary hardware settings and computer settings, and install the DeviceNet Configurator software in the computer.

Section 3 Section 3 explains the basic operation of the DeviceNet Configurator.

**Section 4** Section 4 explains how to enable remote I/O communications by creating device parameters for a virtual network constructed with the DeviceNet Configurator.

**Section 5** Section 5 explains the online operation of the DeviceNet Configurator, such as data downloading to or uploading from devices and device monitoring.

**Section 6** Section 6 provides information on manipulating files, including saving and reading procedures.

The *Appendices* provide information on error messages, troubleshooting, the error history, Unit replacement, replacing the C200W-DRM21 on a CS1-series PC with the CS1W-DRM21, replacing the C200W-DRT21 with the CS1W-DRM21's slave functions, and the dimensions of PCMCIA Card and Unit.

Manual	Products	Contents	Cat. No.
DeviceNet Configurator Ver. 2.□ Operation Manual (This manual)	WS02-CFDC1-E DeviceNet Configurator 3G8F5-DRM21 ISA Board 3G8E2-DRM21 PCMCIA Card	Information on using the Configurator.	W382
CS/CJ-series DeviceNet Unit Operation Manual	CS1W-DRM21 CJ1W-DRM21 DeviceNet Unit	Information on CS/CJ-series DeviceNet Units.	W380
DeviceNet (CompoBus/D) Operation Manual	CVM1-DRM21-V1 DeviceNet Master Unit C200HW-DRM21-V1 DeviceNet Master Unit	Information on C200H-series, CVM1, and CV-series DeviceNet Units, as well as general DeviceNet communications specifications and wiring methods.	W267

The following manuals provide information on the DeviceNet and OMRON DeviceNet products.

WARNING Failure to read and understand the information provided in this manual may result in personal injury or death, damage to the product, or product failure. Please read each section in its entirety and be sure you understand the information provided in the section and related sections before attempting any of the procedures or operations given.

## PRECAUTIONS

This section provides general precautions for using the DeviceNet Configurator and related devices.

The information contained in this section is important for the safe and reliable application of the DeviceNet Configurator and personal computer. You must read this section and understand the information contained before attempting to set up or operate a DeviceNet Configurator and personal computer.

1	Intended Audience	xii
2	General Precautions	xii
3	Safety Precautions	xii
4	Operating Environment Precautions	xiii
5	Application Precautions	xiv

## 1 Intended Audience

This manual is intended for the following personnel, who must also have knowledge of electrical systems (an electrical engineer or the equivalent).

- Personnel in charge of installing FA systems.
- Personnel in charge of designing FA systems.
- Personnel in charge of managing FA systems and facilities.

#### 2 General Precautions

The user must operate the product according to the performance specifications described in the operation manuals.

Before using the product under conditions which are not described in the manual or applying the product to nuclear control systems, railroad systems, aviation systems, vehicles, combustion systems, medical equipment, amusement machines, safety equipment, and other systems, machines, and equipment that may have a serious influence on lives and property if used improperly, consult your OMRON representative.

Make sure that the ratings and performance characteristics of the product are sufficient for the systems, machines, and equipment, and be sure to provide the systems, machines, and equipment with double safety mechanisms.

This manual provides information for installing and operating the DeviceNet Configurator. Be sure to read this manual before operation and keep this manual close at hand for reference during operation.

**WARNING** It is extremely important that a PC and all PC Units be used for the specified purpose and under the specified conditions, especially in applications that can directly or indirectly affect human life. You must consult with your OMRON representative before applying a PC System to the above-mentioned applications.

## 3 Safety Precautions

**WARNING** Never attempt to disassemble a Board or Card or touch the inside of a Board or Card while power is being supplied. Doing so may result in serious electrical shock or electrocution.

- WARNING Provide safety measures in external circuits, i.e., not in the Programmable Controller (CPU Unit including associated Units; referred to as "PC"), in order to ensure safety in the system if an abnormality occurs due to malfunction of the PC or another external factor affecting the PC operation. Not doing so may result in serious accidents.
  - Emergency stop circuits, interlock circuits, limit circuits, and similar safety measures must be provided in external control circuits.
  - The PC will turn OFF all outputs when its self-diagnosis function detects any error or when a severe failure alarm (FALS) instruction is executed. As a countermeasure for such errors, external safety measures must be provided to ensure safety in the system.
  - The PC outputs may remain ON or OFF due to deposition or burning of the output relays or destruction of the output transistors. As a counter-

1

measure for such problems, external safety measures must be provided to ensure safety in the system.

- When the 24-VDC output (service power supply to the PC) is overloaded or short-circuited, the voltage may drop and result in the outputs being turned OFF. As a countermeasure for such problems, external safety measures must be provided to ensure safety in the system.
- WARNING The CPU Unit refreshes I/O even when the program is stopped (i.e., even in PROGRAM mode). Confirm safety thoroughly in advance before changing the status of any part of memory allocated to I/O Units, Special I/O Units, or CPU Bus Units. Any changes to the data allocated to any Unit may result in unexpected operation of the loads connected to the Unit. Any of the following operation may result in changes to memory status.
  - Transferring I/O memory data to the CPU Unit from a Programming Device.
  - Changing present values in memory from a Programming Device.
  - Force-setting/-resetting bits from a Programming Device.
  - Transferring I/O memory files from a Memory Card or EM file memory to the CPU Unit.
  - Transferring I/O memory from a host computer or from another PC on a network.
  - **Caution** Confirm safety at the destination node before transferring a program to another node or changing contents of the I/O memory area. Doing either of these without confirming safety may result in injury.

#### 4 Operating Environment Precautions

Do not install the Unit in any of the following locations.

- Locations subject to direct sunlight.
- Locations subject to temperatures or humidities outside the range specified in the specifications.
- Locations subject to condensation as the result of severe changes in temperature.
- Locations subject to corrosive or flammable gases.
- Locations subject to dust (especially iron dust) or salt.
- Locations subject to exposure to water, oil, or chemicals.
- Locations subject to shock or vibration.

Provide proper shielding when installing in the following locations:

- Locations subject to static electricity or other sources of noise.
- Locations subject to strong electromagnetic fields.
- Locations subject to possible exposure to radiation.
- Locations near to power supply lines.
- ▲ Caution The operating environment of the PC System can have a large effect on the longevity and reliability of the system. Improper operating environments can lead to malfunction, failure, and other unforeseeable problems with the PC System. Be sure that the operating environment is within the specified conditions at installation and remains within the specified conditions during the life of the system.

## 5 Application Precautions

Observe the following precautions when using the DeviceNet Configurator.

- **WARNING** Failure to abide by the following precautions could lead to serious or possibly fatal injury. Always heed these precautions.
  - Always connect to a class-3 ground (100  $\Omega$  or less) when installing the Units.
  - Caution Failure to abide by the following precautions could lead to faulty operation or the PC or the system or could damage the PC or PC Units. Always heed these precautions.
    - Enable the scan list to before operating the system.
    - When adding a new node to the network, make sure that the baud rate is the same as other nodes.
    - Use specified communications cables.
    - Do not extend connection distances beyond the ranges given in the specifications.
    - Always turn OFF the power supply to the personal computer, Slaves, and Communications Units before attempting any of the following.
      - Attaching or detaching the DeviceNet Board or Card.
      - Assembling the Units.
      - Setting rotary switches.
      - Connecting or wiring the cables.
      - Connecting or disconnecting connectors.
    - Be sure that the communications cables and other items with locking devices are properly locked into place.
    - Be sure that all Board mounting screws, connector screws, and cable screws are tightened to the torque specified in this manual.
    - Always use the power supply voltage specified in this manual.
    - Double-check all the wiring and connection of terminal blocks and connectors before mounting the Units.
    - Use crimp terminals for wiring. Do not connect bare stranded wires directly to terminals.
    - Observe the following precautions when wiring the communications cable.
      - Separate the communications cables from the power lines or high-tension lines.
      - Do not bend the communications cables.
      - Do not pull on the communications cables.
      - Do not place heavy objects on top of the communications cables.
      - · Be sure to wire communications cable inside ducts.
      - Use appropriate communications cables.
    - Before touching the PCI Board, be sure to first touch a grounded metallic object in order to discharge any static build-up. Not doing so may result in malfunction or damage.

- Take appropriate measures to ensure that the specified power with the rated voltage and frequency is supplied in places where the power supply is unstable. An incorrect power supply may result in malfunction.
- Install external breakers and take other safety measures against short-circuiting in external wiring. Insufficient safety measures against short-circuiting may result in burning.
- Double-check all the wiring and switch settings before turning ON the power supply.
- Check the user program for proper execution before actually running it on the Unit. Not checking the program may result in an unexpected operation.
- After replacing Units, resume operation only after transferring to the new CPU Unit and/or Special I/O Units the contents of the DM Area, HR Area, and other data required for resuming operation. Not doing so may result in an unexpected operation.
- When transporting or storing the product, cover the PCBs with electrically conductive materials to prevent LSIs and ICs from being damaged by static electricity, and also keep the product within the specified storage temperature range.
- Do not touch the mounted parts or the rear surface of PCBs because PCBs have sharp edges such as electrical leads.
- Do not attempt to disassemble, repair, or modify any Units.

## SECTION 1 Overview

This section describes the features, specifications, operating conditions, and interfaces of the DeviceNet Configurator.

1-1	Introdu	ction	2
	1-1-1	Connecting to the DeviceNet Network	2
	1-1-2	Models	3
	1-1-3	Overview	3
	1-1-4	Configurator Specifications	4
	1-1-5	Upgrade History for DeviceNet Configurator Ver. 2. $\Box$	5
	1-1-6	Files Created with Configurator	7
	1-1-7	Importing Version 1 Configurator Files	7
	1-1-8	Functions of the Configurator	7
	1-1-9	Comparison with Previous Versions of the Configurator	11
1-2	Confiri	ning Product Contents	12
1-3	Operati	ing Environment	13
	1-3-1	Configurator Precautions	13
	1-3-2	PCMCIA Card Precautions.	13

## 1-1 Introduction

The Configurator is a software package that allows the user to construct, set up, and maintain a multivendor DeviceNet network through user-friendly graphic displays.

The Configurator internally constructs a virtual network in the Network Configuration Window, making it possible to set up and monitor DeviceNet devices.

The Configurator is referred to as simple the Configurator in the rest of this manual.

#### 1-1-1 Connecting to the DeviceNet Network

The Configurator is connected to the DeviceNet network using one of the following three methods. The same online functions are supported by all three methods.

Connection from Dedicated Board/Card Installed in Computer The computer running the Configurator is connected to the DeviceNet network by installing an OMRON DeviceNet Board in the computer.



The Configurator is treated as a single DeviceNet node.

## Serial Connection from COM port of Computer

The computer running the Configurator is connected to the DeviceNet network by connecting from the COM port of the computer to the serial communications port (peripheral port) of a CS/CJ-series CPU rack containing a CS1W-DRM21 or CJ1W-DRM21 DeviceNet Unit.



The Configurator is not treated as a single DeviceNet node.

#### Ethernet Connection

The computer running the Configurator is connected to the DeviceNet network by connecting from the Ethernet port of the computer to a CS/CJ-series Ethernet Unit.

Section 1-1



#### 1-1-2 Models

Product	Model	Contents	Method of connecting to DeviceNet network	Personal computer	Applicable OS
Configurator (Ver. 2.□)	WS02-CFDC1-E	Installation disk (CD-ROM)	<ul> <li>Either one of the following methods</li> <li>Dedicated Board/Card (See note.)</li> <li>Serial connection from COM port of computer</li> <li>Ethernet connection from Ethernet port of computer</li> </ul>	IBM PC/AT or compatible	Windows 95, 98, Me, 2000 or NT4.0

#### **Note** Use the following dedicated Boards and Card.

Model	Contents	Personal computer	Applicable OS
3G8F5-DRM21	Dedicated ISA Board and Configurator (Ver. 2.□) installation disk	IBM PC/AT or compatible	Windows 95, 98 or NT4.0
3G8E2-DRM21	Dedicated PCMCIA Card and Configurator (Ver. 2.□) installation disk		Windows 95 or 98

#### 1-1-3 Overview



**Note** Master device parameters used in a scan list are created with either of the following methods.

- a) Using the Parameter Wizard
- b) Setting all parameters

## 1-1-4 Configurator Specifications

Item			Specification	
Operating environment	Hardware	Personal computer: IBM PC/AT or compatible CPU: Pentium 166 MHz or higher Memory: 32 Mbytes Hard disk: A minimum of 15 Mbytes Windows 95, 98, Me, or NT4.0, or 2000 (Some limitations exist for the Dedicated Boards.)		
	OS			
Network connect	ion method	Dedicated Board/Card	3G8F5-DRM21: Dedicated ISA Board 3G8E2-DRM21: Dedicated PCMCIA Card	
		Serial connection	Connect from the COM port of the computer to the DeviceNet network, via the following serial communi- cations port (Toolbus or Host Link) on a CS/CJ-series PLC. • Peripheral port of CPU Unit	
			RS-232C port of CPU Unit	
			RS-232C port of Serial Communications Board/Unit	
			Note The CS/CJ-series DeviceNet Unit (CS1W- DRM21 or CJ1W-DRM21) is necessary to con- nect to the DeviceNet network using serial communications.	
		Ethernet connection	Connect from the Ethernet port of the computer to the DeviceNet network, via an Ethernet Unit for a CS/CJ-series PLC.	
			Note The CS/CJ-series Ethernet Unit (CS1W- ETN□□ or CJ1W-ETN□□) and CS/CJ-series DeviceNet Unit (CS1W-DRM21 or CJ1W- DRM21) are necessary to connect to the DeviceNet network using Ethernet.	
Node address used	Dedicated Board/Card	A single node address is used.		
	Serial connection	No node address is used.		
Connectable number of Boards/ Cards		One/network		
DeviceNet masters		OMRON CS1W-DRM21, CJ	1W-DRM21, CVM1-DRM21-V, or C200HW-DRM21-V1	
		Note Device parameters of file format for open n DeviceNet.	f OMRON DeviceNet Units are output in the DRM_UNIT etwork controllers or the NX-Server file format for the	

Item		Specification
Main functions	Setup functions	Master device parameter settings for OMRON's Master
		<ul> <li>Remote I/O master user-set allocations (with a scan list) The node address order can be set as desired. Two output blocks and two input blocks can be allocated. (See note.)</li> </ul>
		Remote I/O slave user-set allocations
		Setting master remote I/O communications connections.
		<ul> <li>Setting slave remote I/O communications connections.</li> </ul>
		• Enabling or disabling device data checks through remote I/O communications (checks on slave vendor, device type, and product code data).
		<ul> <li>Setting an explicit message monitor timer list.</li> </ul>
		<ul> <li>Setting a COS/cyclic heart beat timer value.</li> </ul>
		<ul> <li>Setting the communications cycle time.</li> </ul>
		Note 1. A device Parameter Wizard is supported for the Master.
		<ol> <li>Using remote I/O master user-set allocations eliminates restrictions on node addresses. Furthermore, a number of masters can be mounted to the PC with no allocation area duplication.</li> </ol>
		Setting parameters for slaves (including other manufacturers' slaves).
		Setting node addresses and baud rates.
		Setting I/O comments (for slave I/O data).
	Monitoring functions	Listing information on devices connected to the network (in node address order or remote I/O configuration order, for example).
		Monitoring status of OMRON Master Unit.
		• Monitoring error history of OMRON Master Unit (time, error code, and error condi- tion).
		Monitoring communications cycle time.
		Monitoring slave status and parameters.
	Storage functions	• Saving the parameters of devices connected to the network as network configuration files.
		• Saving the parameters of each device, in device units, as device parameter files.
	File export functions	• Exporting the network configuration list (contents displayed in the Detailed Display Mode) in the CSV file format.
		Exporting I/O comments in CSV file format.
		• Device parameters of OMRON's Master Units are exported in the DRM_UNIT (vir- tual unit) file format for open network controllers.
		• Device parameters of OMRON's Master Units are exported in the NX-Server file for- mat for the DeviceNet.
	Other functions	Reading/preparing EDS files.
		Printing master/slave device parameters.
		<ul> <li>Setting/reading parameters with explicit messages.</li> </ul>

**Note** To use the Configurator to construct a DeviceNet network or make master or slave settings, the EDS file is required. If other manufacturers' slaves are used, obtain the correct EDS files.

#### 1-1-5 Upgrade History for DeviceNet Configurator Ver. 2.

Changes Made from Ver. 2.01 to Ver. 2.02

The following new devices were supported.

- DeviceNet-compatible 1-axis positioner (3F88M-DRT141)
- I/O Link Unit (CPM1A-DRT21) for CPM2A/CPM1A

When changing the I/O size of a slave with **Device**, **Property** and the **I/O Information** Tab, the new version made it possible to select whether or not to automatically update the I/O size registered in the master scan list. (This can be switched from the Option Menu. It is set not to automatically update as the default.) (Refer to 4-4-2 CVM1-DRM21-V1 or C200HW-DRM21-V1.)

	<ul> <li>When setting the I/O size of a slave with the <i>Device, Property</i> and the I/O Information Tab, the new version made it possible to attain the I/O size from the master scan list. This is possible, however, only with OMRON Master Units, and the I/O size cannot be attained with the slave function of the CS1W-DRM21. (Refer to 4-4-2 CVM1-DRM21-V1 or C200HW-DRM21-V1.)</li> <li>When the slave I/O size and the I/O size registered in the master scan list do not match, an exclamation mark (!) was made to appear next to the slave icon. The new version made it possible to send explicit requests other than Get Attribute Single and Set Attribute Single with <i>Setup Parameters</i> in the Tools</li> </ul>
Changes Made from Ver	Menu.
2.02 to Ver. 2.10	C.I-series DeviceNet Unit (C.I1W-DRM21)
	Fiber Amplifier Communications Unit (E3X-DRT21)
	With a CS/CJ-series DeviceNet Unit and an Ethernet Unit both mounted, the new version made it possible for the Configurator computer to connect directly to the Ethernet network to make DeviceNet settings via Ethernet. This became possible by selecting the CS/CJ Ethernet Unit interface with <b>Select Interface</b> in the Option Menu. Refer to 3-3 Connecting to the Network.
	<ul> <li>Changes were made so that when the detailed display of the Network Configuration Window is shown, the master/slave relationship is clarified. It was also made so that the explanations set for each device would be shown as comments in the detailed display. Refer to <i>3-1-2 Network Configuration Window</i>.</li> <li>The new version made it possible to save CVM1-DRM21-V1, C200HW-DRM21-V1, CS1W-DRM21, and CJ1W-DRM21 error histories in the CSV file format. Refer to <i>5-4-2 Monitoring Devices</i>.</li> </ul>
Changes Made from Ver. 2.10 to Ver. 2.11	When connected via a serial communications port, the new version made it possible to connect to the DeviceNet network across three network levels. Refer to 3-3 Connecting to the Network.
	When connected via Ethernet, the new version made it possible to connect to the DeviceNet network across three network levels. Refer to 3-3 Connecting to the Network.
	The new version made it possible to remove the hardware list for use in a sep- arate window. Refer to 3-3 Connecting to the Network.
	When downloading parameters to the Master Unit with the PLC not in the Pro- gram mode, a confirmation dialog box was made to appear to confirm whether or not to switch to Program mode. Refer to 5-2-2 Downloading Device Param- eters.
	In the Network Configuration Window, the WD30-S DeviceNet Wireless Unit (slave) was made to be displayed also in the DeviceNet network configuration under the Master Unit.
	The monitor screen was made easier to see.
	The new version made it possible to compare the entire network configuration (including each device parameter). Refer to 5-3-3 Verifying the Network Configuration.

#### 1-1-6 Files Created with Configurator

It is r	possible to	o create th	ne following	files with	the Cor	nfigurator.

Files	Description	Extension
Network configuration files	The parameter file for all devices (master and slave devices) on a single virtual network (Network Configuration Window)	.npf
Device parameter files	Parameters for each device (master or slave)	.dvf
EDS files	A common device definition file on the DeviceNet network	.eds

Note In addition, files can be exported and saved in the following formats.

- The network configuration (contents displayed in the Detailed Display Mode) can be saved in the CSV file format.
- I/O comments can be exported to a CSV file
- Device parameters of OMRON Master Units can be saved in the DRM\_UNIT (virtual unit) file format for open network controllers.
- Device parameters of OMRON Master Units can be saved in the NX-Server file format for the DeviceNet.

#### **1-1-7** Importing Version 1 Configurator Files

Version 2 of the Configurator can import files created with version 1 of the Configurator.

Files	Description	Extension
Network configuration files	The parameter file for all devices (master and slave devices) on a single virtual network (Network Configuration Window).	.ntf
Master parameter files	Parameters for each master device	.dsf
Slave parameter files	Parameters for each slave device	.dpf

#### 1-1-8 Functions of the Configurator

The DeviceNet Configurator has 2 modes: Online and offline.

Online: The Configurator is connected to the DeviceNet network for network communications.

Offline: The Configurator is not connected to the DeviceNet network.

The Configurator has the following functions that allow the user to design, set up, and control the DeviceNet network.

Windows 95, 98, Me, and NT4.0 or 2000 Compatibility

Other applications can be run simultaneously while monitoring the DeviceNet network's status.

The Configurator will operate under the Windows operating systems.

**Note** The ISA Board is compatible with Windows 95, 98, and NT4.0 only. The PCMCIA Card is also compatible with Windows 95 and 98 only.

**Graphic Displays** The Configurator has Windows-style displays that are easy to understand and easy to use.

Required master and slave settings are made just by selecting the icons. It is not necessary to input complicated commands. Even a beginner can make settings and monitor operation with ease.

Network Management	The Configurator constructs a virtual network in the Network Configuration Window and adds necessary devices to the network. The construction of the network is thus possible offline. It is possible for the virtual network (i.e., the Network Configuration Window) to read the network configuration from a file or actual network. Furthermore, the network configuration can be downloaded from the virtual network (i.e., the Network Configuration Window) and all settings can be reg- istered with actual devices.
Online Connections to DeviceNet Network	The online functions of the Configurator are executed by connecting the Con- figurator to the DeviceNet network through a dedicated Board or Card. (In that case, the Configurator will be treated as a single node on the DeviceNet net- work.) Connection can also be made from the COM port of the computer to the serial communications port of a CS/CJ-series CPU Unit (or the Serial Communica- tions Board/Unit) via Toolbus or Host Link. In this case, a CS/CJ-series DeviceNet Unit (CS1W-DRM21 or CJ1W-DRM21) is necessary for connection to the DeviceNet network. Similarly, connection can be made from the Ethernet port of the computer via a CS/CJ-series Ethernet Unit. In this case also, a CS/CJ-series DeviceNet Unit (CS1W-DRM21 or CJ1W-DRM21) is necessary for connection to the DeviceNet network.

Device Settings	The following parameter	settings are for	other manufactu	irers' slaves as well
	as OMRON's master and	slave devices.		

Setting	Description	CS1W-DRM21	C200HW-DRM21-V1 and CVM1-DRM21-V1
Master functions enable/dis- able	Enables or disables the master functions for remote I/O communications.	Yes	No
I/O allocations	Slaves are allocated to the master in the desired areas and order.	Yes	Yes
	If the I/O allocations are simple, the Parameter Wizard can be used to allocate I/O very easily.		
	I/O allocations are possible as well by just drag- ging and dropping the slave to the allocation area.		
Communication cycle time	Sets the communication cycle time. (The interval for I/O communications with slaves.)	Yes	Yes
Connection setting	During remote I/O communications, the user can make connection settings for up to two connec- tions per slave. Automatic connection settings can be selected as well.	Yes	No
Device data check	Determines whether slave device data items (i.e., the vendor, device type, and product code) are checked during remote I/O communications.	Yes	No
Slave functions enable/dis- able	Enables or disables the slave functions for remote I/O communications.	Yes	No
Slave functions I/O allocation	I/O allocations can be made for slaves of the CS1-series DeviceNet Unit. (I/O can be allocated freely.)	Yes	No

A comment can be added to the I/O data of each device.

Setting	Description	CS1W-DRM21	C200HW-DRM21-V1 and CVM1-DRM21-V1
Remote I/O communications startup	Determines whether or not to start remote I/O communications when the master is started.	(See note 1.)	Yes
Explicit message communi- cations set/reset	Setting or resetting the function to enable explicit message communications without remote I/O communications.	(See note 2.)	Yes

**Note** 1. The same function can be achieved with the master functions enable/disable setting.

2. When not registered in the scan list, explicit message communications are possible without remote I/O communications.

**Device Monitoring** The status, error log, etc., of a device can be monitored, provided that the device supports the monitor function. The monitor function is supported by OMRON's Master Units and some Special Slaves.

#### **Master Status Monitor**

The status of the specified OMRON master is displayed. The status of a slave will be displayed as well if it is in remote I/O communications with the Master.

#### **Slave Status Monitor**

The status of the specified MULTIPLE I/O TERMINAL (i.e., DRT1-COM Communications Unit) will be displayed. The configuration of the I/O Unit connected to the Communications Unit will be displayed as well.

#### Master Error Log Display

The error log in the specified OMRON master can be read, displayed, or cleared.

#### Communication Cycle Time Display

The present communication cycle time of the specified OMRON master will be displayed. The maximum communications cycle time and minimum communications cycle time counted from the moment previous values are refreshed will be displayed as well. The maximum communications cycle time and minimum communications cycle time can be reset any time.

#### **File Management**

The Configurator saves or loads the following data in files.



- The network configuration file includes the parameters for all devices (master and slave devices). The parameter file for each master and slave device contains the individual device parameters. Parameter files can be managed independently from network configuration files.
- Files from version 1 of the Configurator are not compatible with those of version 2. Files from version 1 of the Configurator can be read, however, by using the import function.
- **Note** Device parameter files of the CS-series CS1W-DRM21 DeviceNet Unit are compatible with Unit setup files that can be backed up in the Memory Card of the CS1-series CPU Unit.
- Device (EDS File)The construction of a network with the Configurator requires the appropriate<br/>EDS files. Information on each device is obtained from the EDS file.

If no EDS file is provided, use the hardware (EDS file) management function to install or create an EDS file.

 Other Functions
 DeviceNet vendor definition files, device type definition files, and slave setup files for the MULTIPLE I/O TERMINALs can be set for the Configurator, with which additional DeviceNet vendors and device types can be used.

Furthermore, the Expansion Module can be installed so that edit and monitoring functions can be added for other manufacturers' devices (masters and slaves) or OMRON devices that will be released in the future.

**Note** When the dedicated Board is used to connect to the DeviceNet network, the Configurator will be a single node on the DeviceNet network. Be sure that the node address of the computer is not duplicated with that of any other node.

## 1-1-9 Comparison with Previous Versions of the Configurator

The following table lists the difference in function between version 1.20 and version  $2.\Box$  of the Configurator.

Item		Version 2.	Version 1.20	
Basic configuration/operations		Dragging and dropping offline make it possible to create a virtual network (for display) in the Configurator.	Each device must be set online, except the scan list of the master.	
		also be set online.		
Network connection		<ul> <li>Dedicated ISA Board</li> <li>Dedicated PCMCIA Card</li> <li>From the COM port of the computer to the serial port of the PLC. (Pos- sible with CS/CJ-series PLCs only.)</li> <li>From the Ethernet port of the com- puter to the Ethernet Unit of the PLC. (Possible with CS/CJ-series PLCs only.)</li> </ul>	<ul> <li>Dedicated ISA Board</li> <li>Dedicated PCMCIA Card</li> </ul>	
Supported DeviceNet masters	OMRON masters	CS1W-DRM21, CJ1W-DRM21, CVM1-DRM21-V1 or C200HW- DRM21-V1 Note Output is possible in file for- mats compatible with open network controllers or NX- Servers for DeviceNet.	CVM1-DRM21-V1 or C200HW- DRM21-V1 Note The CS1W-DRM21, open network controllers, or NX- Servers for DeviceNet cannot be used.	
	Slave remote I/O allocations	Possible with CS/CJ-series PLCs only.	Impossible	
	Master remote I/O communications settings	Possible with CS/CJ-series PLCs only.	Impossible	
	Slave remote I/O communications settings	Possible with CS/CJ-series PLCs only.	Impossible	
	Enabling or disabling device data checks in remote I/O communication	Possible with CS/CJ-series PLCs only.	Impossible	
	Explicit message monitoring timer list settings	Possible with CS/CJ-series PLCs only.	Impossible	
	COS/Cyclic heart beat timer settings	Possible with CS/CJ-series PLCs only.	Impossible	
	I/O comment settings (for slave I/O data)	Possible	Impossible	
Monitor functions	Display of list of data on net- work devices	Possible	Possible	
	Unit status monitoring	Possible with CS/CJ-series PLCs only.	Impossible	
	Slave function monitoring	Possible	Impossible	
Storage	Network configuration files	Possible (extension: .npf)	Possible (extension: .ntf)	
functions	Device parameter files	Possible (extension: .dvf)	Possible (master: Extension: .dsf slave: Extension: .dpf)	
	EDS files	Possible (extension: .eds)	Impossible	
	Compact flash memory	Stored in compact flash memory and downloaded to the master. (Possible with CS/CJ-series PLCs only.)	Impossible	

Item			Version 2.	Version 1.20
File export functions	I/O command of file format	output in CCV	Possible	Impossible
	Device parameter of OMRON masters	Output for open network controllers in DRM_UNIT file format	Possible	Impossible
		Output for NX- Servers for DeviceNet in NX-Server file format	Possible	Impossible
	EDS file creation		Possible Device discrimination data and I/O data files are created. Device param- eter data files cannot be created.	Conditionally possible Tentative EDS files (with device discrimination data only) are created. Device I/O data or parameter data files cannot be created.
	I/O allocation d check between	uplication Masters	Impossible	Possible

## **1-2 Confirming Product Contents**

Please check the contents of your product as soon as if arrives.

Product	Contents	Quantity
WS02-CFDC1-E Configurator	Configurator installation disk (CD-ROM)	1
	3G8F5/3G8E2 setup disk (driver setup tool on CD-ROM)	1
	Operation manual (this manual)	1
	User registration card and software license	1
3G8F5-DRM21 ISA Board	Dedicated ISA board	1 (with 1communications connector included)
	Configurator installation disk (CD-ROM)	1
	3G8F5/3G8E2 setup disk (driver setup tool on CD-ROM)	1
	Operation manual (this manual)	1
	User registration card and software license	1
3G8E2-DRM21 PCMCIA Board	Dedicated PCMCIA board	1 (with 1communications connector included)
	Configurator installation disk (CD-ROM)	1
	3G8F5/3G8E2 setup disk (driver setup tool on CD-ROM)	1
	Operation manual (this manual)	1
	User registration card and software license	1

## 1-3 Operating Environment

The	Configurator	operates	in the	following	environment.
-----	--------------	----------	--------	-----------	--------------

Compatible personal computer	IBM PC/AT or compatible personal computer
OS	Windows 95, 98, Me, NT4.0, or 2000
CPU	Pentium 166 MHz or better
Available hard disk space	15 Mbytes min.
Memory	32 Mbytes min.
CD-ROM drive	1 (used for software installation)
Display	VGA or better
Hardware for network connec- tion	One of the following is required for online connection to the DeviceNet network.
	OMRON DeviceNet Board/Card Dedicated PCMCIA Card (3G8E2-DRM21) Dedicated ISA Board (3G8F5-DRM21)
	Ethernet port
	COM port

#### **1-3-1** Configurator Precautions

Observe the following precautions when using the Configurator.

- The only DeviceNet masters that can be handled by the Configurator are OMRON's CS1W-DRM21, CJ1W-DRM21, CVM1-DRM21-V1, and C200HW-DRM21-V1.
- When a dedicated Board or Card is used, the Configurator is connected to the DeviceNet network as a single node. The Configurator cannot be connected to the network unless a node address is available. No node address is required if the Configurator is connected via Ethernet or a serial communications port.
- Do not use the Configurator in a location with too much electromagnetic noise. If noise is too extreme, the computer may run out-of-control, although there will be no negative effects on the DeviceNet network even if noise causes the computer to run out-of-control.

#### 1-3-2 PCMCIA Card Precautions

• Use the PCMCIA Card Unit on a level surface, as shown in the following diagram.



- Do not bend or pull the cable that connects the PCMCIA Card and the PCMCIA Card Unit.
- The PCMCIA Card is a portable peripheral device. Do not leave it connected to the network.

• Attaching ferrite cores between the PCMCIA Card and the PCMCIA Card Unit can reduce noise from the network.

Attach one ferrite core near the Unit and another near the card, as shown in the following diagram.



## SECTION 2 Installation

This section explains how to install the ISA Board, PCI Board, or PCMCIA Card hardware in the computer, make the necessary hardware settings and computer settings, and install the DeviceNet Configurator software in the computer.

2-1	I Installation Procedure		16
2-2	ISA Board Installation		
	2-2-1	Preparation for Installation	16
	2-2-2	ISA Board Components and Functions	19
	2-2-3	DIP Switch Settings	19
	2-2-4	Installation in the Computer	20
	2-2-5	3G8F5/3G8E2-DRM21 Driver Setup Software Installation	20
2-3	PCMC	IA Card Installation	24
	2-3-1	Preparation for Installation	24
	2-3-2	PCMCIA Card Components and Functions	25
	2-3-3	PCMCIA Card Setup	25
	2-3-4	3G8F5/3G8E2-DRM21 Driver Setup Software Installation	28
2-4	Config	urator Installation	32

## 2-1 Installation Procedure

The following flowchart shows the steps involved in installing the Configurator.



## 2-2 ISA Board Installation

This section explains how to set the dedicated ISA Board used by the Configurator and install the Board in the computer.

## 2-2-1 Preparation for Installation

The dedicated ISA Board used by the Configurator is not a plug-and-play board. Before installing the Board, check the items listed in the following table.

Item	Description
Windows OS	Verify that the OS is Windows 95, 98, or NT4.0.
Unused expansion slot	Verify that there is an unused expansion slot (ISA slot) in the computer.
Available memory	The ISA Board for the Configurator occupies 16 KB of memory between the memory range of C0000 and DFFFF as a system resource. Verify that the computer has 16 KB of available memory between the memory range of C0000 and DFFFF and select a leading address. The leading address is needed during the driver installation.
Unused I/O port address	The ISA Board for the Configurator occupies 8 port addresses between the memory range of 200 and 2FF as a system resource. Verify that the computer has 8 addresses available and select a leading I/O address. The leading I/O address is needed when setting the ISA Board's DIP switch.

**Note** The dedicated ISA Board used by the Configurator cannot be used with Windows Me or 2000.

When Windows 95/98 is being used, follow this procedure to verify the available memory and I/O port addresses and reserve these addresses for the Configurator's ISA Board. *1,2,3...* 1. Click the **Start** Button and select **Control Panel** from the Settings Menu.



- 2. Double-click the System Icon in the Control Panel Window.
- 3. Select the **Device Manager** Tab from the System Properties Window.
- 4. Select *Computer* and click the **Properties** Button at the bottom of the window.

System Properties ? × General Device Manager Hardware Profiles Performance
View devices by type     View devices by connection
CDROWS     CDROW     CDROW     Disk drives     Display adapters     Floppy disk controllers     Hard disk controllers     Get      Monitor     Mouse     Monitor     Ports (COM & LPT)     System devices
Properties Refresh Remove Print
OK Cancel

- 5. Select the View Resources Tab.
- 6. Select **Input/output (I/O)** from the View Resources Page and click the **OK** Button. The current I/O port allocations will be displayed. Find an unused I/O port address that can be used for the ISA Board.

<ul> <li>Interrupt reque</li> <li>Input/output (</li> </ul>	est (IRQ) C <u>D</u> irect memory access (DMA) [/O] C <u>M</u> emory	
Setting	Hardware using the setting	<b></b>
💻 00F0 - 00FF	Numeric data processor	
🗟 0170 - 0177	Opti Dual PCI IDE Controller	_
🔁 0170 - 0177	Secondary IDE controller (single fifo)	
🚭 01F0 - 01F7	Opti Dual PCI IDE Controller	
🚭 01F0 - 01F7	Primary IDE controller (single fifo)	
J 02F8 - 02FF	Communications Port (COM2)	1
🖳 0370 - 0371	System board extension for PnP BIOS	L
5 ····		l N

7. Select the **Reserve Resources** Tab.

Computer Properties	? 🗙
View Resources Reserve Re	esources
<ul> <li>Interrupt request (IRQ)</li> <li>Input/output (I/O)</li> </ul>	C Direct memory access (DMA) C Memory
Setting	
	Add Modify Remove
	OK Cancel

8. Selecting *Input/output (I/O)* from the Reserve Resources Page, click the Add Button, and register the range of I/O port addresses for the ISA Board as shown in the following diagram.

In this example, I/O port addresses 250 through 257 (hexadecimal) are reserved.

Computer Properties	?	×
View Resources Reserve	e Resources	
C Interrupt (equest (IRC C Input/gutput (I/D) Setting	Edit Resource Setting     ? ★       Enter the beginning and ending values of the input/output range you would like to reserve.       Start value:     250 **       End value:     257 **       OK     Cancel	
	Add Modify Remove	
	0K Cancel	

- 9. Follow the same procedure in steps 5 to 8 to reserve 16-Kbytes of memory.
- Click the **OK** Button at the bottom of the Computer Properties Window to accept the changes and close the window. A pop-up window will appear; restart the computer as required.

When Windows NT 4.0 is being used, the operating system does not monitor the status of memory and I/O port addresses. Refer to the user's manuals for the computer and peripheral devices being used and locate an unused address for the Configurator's ISA Board.

Windows NT 4.0 Procedure

#### 2-2-2 ISA Board Components and Functions

The following diagram shows the main components of the Configurator's ISA Board and their functions.



#### 2-2-3 DIP Switch Settings

Before connecting the ISA Board to the computer, set the I/O port address that the Board will use with the Board's DIP switch.

**Caution** Do not touch the ISA Boards connectors or components. Take precautions against static electricity when handling the ISA Board.

Set the I/O port address with the DIP switch as shown in the following table. Be sure to set addresses that aren't being used by another device in order to avoid a conflict. (A pin setting of 0 is OFF and 1 is ON.)

I/O port addresses DIP switch pin settings					Remarks		
(Hexadecimal)	1	2	3	4	5	6	
200 to 207	0	0	0	0	0	0	Used for the game port.
208 to 20F	1	0	0	0	0	0	
210 to 217	0	1	0	0	0	0	
218 to 21F	1	1	0	0	0	0	
220 to 227	0	0	1	0	0	0	
228 to 22F	1	0	1	0	0	0	
230 to 237	0	1	1	0	0	0	
238 to 23F	1	1	1	0	0	0	
240 to 247	0	0	0	1	0	0	
248 to 24F	1	0	0	1	0	0	
250 to 257	0	1	0	1	0	0	Factory default setting.
258 to 25F	1	1	0	1	0	0	

I/O port addresses	DIP switch pin settings			settii	ngs	Remarks	
(Hexadecimal)	1	2	3	4	5	6	
260 to 267	0	0	1	1	0	0	
268 to 26F	1	0	1	1	0	0	
270 to 277	0	1	1	1	0	0	
278 to 27F	1	1	1	1	0	0	Used for LPT2.
280 to 287	0	0	0	0	1	0	
288 to 28F	1	0	0	0	1	0	
290 to 297	0	1	0	0	1	0	
298 to 29F	1	1	0	0	1	0	
2A0 to 2A7	0	0	1	0	1	0	
2A8 to 2AF	1	0	1	0	1	0	
2B0 to 2B7	0	1	1	0	1	0	
2B8 to 2BF	1	1	1	0	1	0	
2C0 to 2C7	0	0	0	1	1	0	
2C8 to 2CF	1	0	0	1	1	0	
2D0 to 2D7	0	1	0	1	1	0	
2D8 to 2DF	1	1	0	1	1	0	
2E0 to 2E7	0	0	1	1	1	0	
2E8 to 2EF	1	0	1	1	1	0	Used for COM port 4.
2F0 to 2F7	0	1	1	1	1	0	
2F8 to 2FF	1	1	1	1	1	0	Used for COM port 2.

Note There are some boards marketed that cannot distinguish between I/O port addresses 200 through 2FF and addresses 600 through 6FF. If one of these boards is allocated address 2 , it will occupy address 6 as well. To avoid any possible conflicts when assigning an address between 600 and 6FF, be sure to assign an address that corresponds to an unused address between 200 and 2FF. (For example if 6A0 to 6A7 are selected, make sure that 2A0 to 2A7 are unused.)

#### 2-2-4 Installation in the Computer

The ISA Board must be installed into one of the computer's expansion slots (ISA slots). When installing the Board, turn OFF the computer's power and take precautions against static electricity. Refer to the computer's user's manual for detailed installation procedures.

If there are no problems with the I/O port settings, the computer should recognize the ISA Board when the computer is turned ON and the ISA Board should be usable. If the ISA Board has been installed successfully, proceed to 2-2-5 3G8F5/3G8E2-DRM21 Driver Setup Software Installation.

#### 2-2-5 3G8F5/3G8E2-DRM21 Driver Setup Software Installation

The operations and displays shown in the following procedure may differ slightly depending on the version of Windows software being used. The displays for Windows 98 are shown here.

- 1,2,3... 1. Exit all other Windows-based programs.
  - 2. Insert 3G8F5/3G8E2-DRM21 setup disk into the CD-ROM drive.
  - 3. Double-click **Setup.exe**. The 3G8F5/3G8E2-DRM21's setup program will start.

The following window will be displayed when the setup program has been loaded. Click the **Next** Button after reading the precautions.

InstallShield Wizard	×
	Welcome to the InstallShield Wizard for 3G8F5,3G8E2-DRM21 Setup The InstallShield® Wizard will install 3G8F5,3G8E2-DRM21 Setup on your computer. To continue, click Next.
	< <u>Back</u> <u>Next&gt;</u> Cancel

4. Specify the destination directory for the driver files. If the default directory shown in the window is acceptable, click the **Next** Button. To specify a different directory, click the **Browse** Button, specify the desired directory, and click the **Next** Button. If a new directory is specified, the software will create it automatically.

Inst	allShield Wizard
Ch	oose Destination Location Select folder where Setup will install files.
	Setup will install 3G8F5,3G8E2-DRM21 Setup in the following folder.
	To install to this folder, click Next. To install to a different folder, click Browse and select another folder.
Insta	Destination Folder C:\\OMRON\3G8F5,3G8E2-DRM21 Setup Browse
Trista	< <u>B</u> ack <u>Next&gt;</u> Cancel

5. Specify the name of the 3G8F5/3G8E2-DRM21 program folder where the Software icons will be registered. If the default program folder is acceptable, click the **Next** Button. To specify a different folder, select an existing folder or enter a new folder name and then click the **Next** Button.
Note The program folder name specified here will be registered under DeviceNet Tools in the Start Menu.

InstallShield Wizard
Select Program Folder Please select a program folder.
Setup will add program icons to the Program Folder listed below. You may type a new folder name, or select one from the existing folders list. Click Next to continue.
Program Folders:
3G8F5,3G8E2-DRM21 Setup
Existing Folders: Accessories Administrative Tools DeviceNet Tools Startup
InstallShieldCancel

- 6. The installation program will be started and the 3G8F5/3G8E2-DRM21 files will be copied to the destination directory.
- 7. A program group folder will be created automatically when the 3G8F5/ 3G8E2-DRM21 Driver Setup Software installation has been completed.
- 8. A pop-up window will be displayed asking whether you want to setup the device driver. If you click the **No** Button, the installation program will end without performing the setup for the ISA Board or PCMCIA Card.

Card Setup 🛛 🕅						
?	Do you want to setup the 3G8F5/3G8E2-DRM21 Card now?					
	<u>Yes</u> <u>N</u> o					

- Note The setup for the ISA Board or PCMCIA Card can be performed later by clicking the **Start** Button, then selecting **Program**, **DeviceNet Tools**, **3G8F5/3G8E2-DRM21 Setup**, **3G8F5/3G8E2-DRM21-Setup**.
- 9. The setup program for the device driver will start. There is already a card driver called Driver250 registered, so click the **Edit** Button. The New selection is used to add a driver for the ISA Board or PCMCIA Card.

Card List About	×
Driver250	<u>N</u> ew <u>R</u> emove <u>E</u> dit
0	K Cancel

The *Remove* selection is used to delete the selected driver.

- Note a) The card name registered here will be used on the Configurator to access the ISA Board or PCMCIA Card.
  - b) Register a name for each ISA Board and PCMCIA Card.
- 10. Set the I/O port address, memory address, and interrupt (IRQ) for the ISA Board or PCMCIA Card being used with the Configurator. (Refer to the screen in step 11.)
- 11. Input the I/O port address set on the board's DIP switch and input a memory address that is available in the computer. Set the interrupt to None.

The Card Name and Driver Name can be changed if necessary.

Make sure that all of the settings are correct and click the **OK** Button. The Configuration Window shown in step 9 will be displayed.

Card Setup X Card Name Driver250
Configuration I/O Port Memory Address 250 Y D0000 V Interrupt None V
OK Cancel

Note The Configurator does not use interrupts. Set the interrupt level to None.

12. Click the **OK** Button to exit the setup program.



With Windows 95/98, the Configurator can be run immediately. Click the **OK** Button to start the Configurator. With Windows NT, a pop-up window will be displayed asking whether or not to restart Windows.

13. This completes installation of the 3G8F5-DRM21 or 3G8E2-DRM21 Driver Setup Software.

# 2-3 PCMCIA Card Installation

This section explains how to set the PCMCIA Card used by the Configurator and install the card in the computer.

# 2-3-1 Preparation for Installation

Before installing the PCMCIA Card, check the items listed in the following table.

ltem	Description		
Windows OS	Verify that the OS is Windows 95 or 98. The PCMCIA Card can only be used in this environment.		
Unused card slot	Verify that there is an unused card slot (type II) in the computer.		
Available memory	The PCMCIA Card used by the Configurator occupies 16 KB of memory as a system resource. Verify that the computer has 16 KB of available memory.	The PCMCIA Card memory and I/O port combi- nations used by the Configurator are set auto- matically by the Windows 95/97 Plug & Play feature. Use one of the automatically set combi-	
Unused I/O port address	The PCMCIA card used by the Configurator occupies 3 ports as a system resource. Verify that the computer has 3 I/O port addresses available.	nations.	

**Note** The PCMCIA Card for the Configurator can't be used with the Windows Me, NT4.0, or 2000 operating systems.

**PCMCIA Card Installation** 

## 2-3-2 PCMCIA Card Components and Functions

The following diagram shows the main components of the Configurator's PCMCIA Card (Card and Unit sections) and their functions.



### 2-3-3 PCMCIA Card Setup

The operations and displays shown in the following procedure may differ slightly depending on the version of Windows software being used. Use the following procedure for setup.

**Before Starting the Setup** Before starting the setup procedure, remove any PCMCIA Cards (including those used by the Configurator) that are currently being used. Since the PCM-CIA Card can't be used temporarily during the setup, it may not be possible to continue the setup operation midway through the operation.

### PCMCIA Card Setup

*1,2,3...* 1. Insert the 3G8F5/3G8E2-DRM21 setup disk into the CD-ROM drive.

2. Install the PCMCIA Card when the Windows system is active. The PCM-CIA Card will be recognized as new hardware and the Add New Hardware Wizard Window will be displayed. Click the **Next** Button.

Add New Hardware Wiz	zard
	This wizard searches for new drivers for:
	Omron-3G8E2-DRM21
	A device driver is a software program that makes a hardware device work.
8	
	< Back Next > Cancel

3. Select **Search for the best driver for your device** as the driver search method, then click the **Next** Button.



4. Select *CD-ROM drive* as the driver location to be searched, then click the **Next** Button.

Add New Hardware Wi	zard
	Windows will search for new drivers in its driver database on your hard drive, and in any of the following selected locations. Click Next to start the search.
	Eloppy disk drives
	CD-ROM drive
🌯 🚴	Microsoft Windows Update
	Specify a location
I ⇒ 1	A:\
	Browse
	< <u>B</u> ack Next > Cancel

5. The driver file will be recognized as shown below. Click the **Next** Button to install the driver.



6. After the driver has been installed, click the Finish Button.



### **Verifying Driver Resources**

 Check that the driver has been installed correctly, as follows: Click the Start Button, select Control Panel from the Settings Menu, double-click the System Icon in the Control Panel, and select the Device Manager Tab from the top of the System Properties Window (refer to page 24). Check that the SST Device has been added, then click the + to the left of SST Devices to check that x or ! is not displayed at PCMCIA Card driver 5136-DN-PCM(n), where n is the PCMCIA slot number.

stem Pr	operties					?
ieneral	Device Ma	inager Hardw	are Profil	es Perfo	ormance	
⊙ Vie	w devices t	y type C	View d	evices by	<u>c</u> onnecti	on
	mputer Disk drive Display ar Floppy dis Hard disk Keyboard Monitors Mouse PCMCIA: POMCIA: SST Dev SST Dev System dr	s dapters k controllers controllers m & LPT) deo and game ces DN-PCM (0) avices	controller	s		
P <u>r</u> c	perties	Re <u>f</u> resh		R <u>e</u> move		Pri <u>n</u> t
					OK	Cancel

 Verify the resources currently being used, as follows: Double-click 5136-DN-PCM(n) to display the 5136-DN-PCM(n) Properties Window and select the Resources Tab from the top of the window. The automatically set Input/Output Range and Memory Range values will be displayed.

5136-DN-PCM (0) Properties	? ×
General Driver Resources	
5136-DN-PCM (0)	
Le utomatic settings	
Setting based on:	<b>v</b>
Resource type Setting	
Memory Range 0250 - 0253 Memory Range 000D 4000 - 000D 7FFF	
Change Setting	
Conflicting device list:	
No conflicts.	×
OK	Cancel

This completes installation of the PCMCIA Card driver. Write down the values of Input/Output Range and Memory Range because these will be needed later when installing the Configurator.

# 2-3-4 3G8F5/3G8E2-DRM21 Driver Setup Software Installation

The operations and displays shown in the following procedure may differ slightly depending on the version of Windows software being used. The displays for Windows 98 are shown here.

- 1,2,3... 1. Exit all other Windows-based programs.
  - 2. Insert 3G8F5/3G8E2-DRM21 setup disk into the CD-ROM drive.

### 3. Double-click Setup.exe.

The 3G8F5/3G8E2-DRM21's setup program will start.

The following window will be displayed when the setup program has been loaded. Click the **Next** Button after reading the precautions.



4. Specify the destination directory for the driver files. If the default directory shown in the window is acceptable, click the **Next** Button. To specify a different directory, click the **Browse** Button, specify the desired directory, and click the **Next** Button. If a new directory is specified, the software will create it automatically.

InstallShield Wizard		X
Choose Destination Location Select folder where Setup will	<b>n</b> install files.	
Setup will install 3G8F5,3G8E	2-DRM21 Setup in the followin	ig folder.
To install to this folder, click N another folder.	lext. To install to a different fol	ter, click Browse and select
Destination Folder C:\\OMRON\3G8F5,3G8	E2-DRM21 Setup	Browse
Installometa	< <u>B</u> ack	Next> Cancel

5. Specify the name of the 3G8F5/3G8E2-DRM21 program folder where the Software icons will be registered. If the default program folder is acceptable, click the **Next** Button. To specify a different folder, select an existing folder or enter a new folder name and then click the **Next** Button.

Note The program folder name specified here will be registered under DeviceNet Tools in the Start Menu.

InstallShield Wizard
Select Program Folder Please select a program folder.
Setup will add program icons to the Program Folder listed below. You may type a new folder name, or select one from the existing folders list. Click Next to continue.
Program Folders:
3G8F5,3G8E2-DRM21 Setup
Existing Folders:
Accessories Administrative Tools DeviceNet Tools Startup
Install6hield

- 6. The installation program will be started and the 3G8F5/3G8E2-DRM21 files will be copied to the destination directory.
- 7. A program group folder will be created automatically when the 3G8F5/ 3G8E2-DRM21 Driver Setup Software installation has been completed.
- 8. A pop-up window will be displayed asking whether you want to setup the ISA Board or PCMCIA Card. Click the **Yes** Button. If you click the **No** Button, the installation program will end without performing the setup for the ISA Board or PCMCIA Card.

Card Setup 🛛 🕅					
2	Do you want to setup the 3G8F5/3G8E2-DRM21 Card now?				
	<u>Yes</u> <u>N</u> o				

- Note The setup for the ISA Board or PCMCIA Card can be performed later by clicking the **Start** Button, then selecting **Program**, **DeviceNet Tools**, **3G8F5/3G8E2-DRM21 Setup**, **3G8F5/3G8E2-DRM21-Setup**.
- 9. The setup program for the device driver will start. There is already a card driver called Driver250 registered, so click the **Edit** Button. The New selection is used to add a driver for the ISA Board or PCMCIA Card.

뿔출 3G8F5/3G8E2-DRM21 Setup	X
Card List About	
Driver250	<u>N</u> ew
	<u>R</u> emove
	<u>E</u> dit
	 Cancel

The *Remove* selection is used to delete the selected driver.

- Note a) The card name registered here will be used on the Configurator to access the ISA Board or PCMCIA Card.
  - b) Register a name for each ISA Board and PCMCIA Card.
- 10. Set the I/O port address, memory address, and interrupt (IRQ) for the ISA Board or PCMCIA Card being used with the Configurator. (Refer to the screen in step 11.)
- Input the values that were confirmed in the PCMCIA Card setup for the I/ O Port, Memory Address, and Interrupt (IRQ).
   The Card Name and Driver Name can be changed if necessary.
   Make sure that all of the settings are correct and click the **OK** Button. The Configuration Window shown in step 9 will be displayed.

Card Setup	×
Card Name	
Configuration	
I/O Port	Memory Address
250 💌	D0000 💌
Interrupt None	
OK	Cancel

Note The Configurator does not use interrupts. Set the interrupt level to None.

12. Click the **OK** Button to exit the setup program.



With Windows NT, a pop-up window will be displayed asking whether or not to restart Windows.

13. This completes installation of the 3G8F5-DRM21 or 3G8E2-DRM21 Driver Setup Software.

# 2-4 Configurator Installation

This section explains how to install the Configurator. The Configurator is installed by executing the setup program of the Configurator.

**Preparation for Installation** Prior to upgrading to a newer Configuration version, be sure to uninstall the previous Configuration version. (Uninstalling is not necessary, however, when the update module is used.)

If the update module was used to upgrade in the past, the update module must also be uninstalled.

When upgrading from a Ver. 1.  $\Box$  Configurator to a Ver. 2.  $\Box$  Configurator, install the driver setup tool first after uninstalling the Ver. 1.  $\Box$  Configurator.

Note Select Settings, Control Panel, and Add/Remove Programs from the Start Menu for uninstalling.

Installation The operations and displays shown in the following procedure may differ slightly depending on the version of Windows software being used. The displays for Windows 98 are shown here.

- 1,2,3... 1. Exit all other Windows-based programs.
  - 2. Insert the setup disk (CD-ROM) into the CD-ROM drive.
    - Double-click Setup.exe. The setup program of the Configurator will start. A pop-up window will indicate the progress of the setup as the program is loaded.

InstallSh	ield Wizard	
DeviceNet Configurator Setup is preparing the InstallShield® Wizard, which will guide you through the rest of the setup process. Please wait.		
	Cancel	

4. The following window will be displayed when the setup program has been loaded. Click the **Next** Button after reading the precautions.

InstallShield Wizard	×
	Welcome to the InstallShield Wizard for DeviceNet Configurator
	The InstallShield® Wizard will install DeviceNet Configurator on your computer. To continue, click Next.
	< Back Mext> Cancel

 The license agreement will be displayed. Read the license agreement. Click the Yes Button if you agree to all the contents. If No Button is selected, the setup procedure will be canceled.

InstallShield Wizard 🛛 🛛 🔀
License Agreement
Please read the following license agreement carefully.
Press the PAGE DOWN key to see the rest of the agreement.
× 
Do you accept all the terms of the preceding License Agreement? If you choose No, the setup will close. To install DeviceNet Configurator, you must accept this agreement.
InstaliShield
< <u>B</u> ack <u>Y</u> es <u>N</u> o

6. Specify the destination directory for the Configurator files. If the default directory shown in the window is acceptable, click the **Next** Button. To specify a different directory, click the **Browse** Button, specify the desired directory, and click the **Next** Button. If a new directory is specified, the software will create it automatically.

InstallShield Wizard	×
Choose Destination Location Select folder where Setup will install files.	B
Setup will install DeviceNet Configurator in the following folder.	
To install to this folder, click Next. To install to a different folder, click Browse and select another folder.	
Destination Folder C:\Program Files\DMRON\DeviceNet Configurator InstallBhfeld	]
< Back Next > Cancel	

 Specify the name of the interface to be used. If the default interface is acceptable, click the Next Button.

InstallShield Wizard Setup Type Choose the setup type that best suits your needs.	
Click the type of Setup you prefer. ISA, PCMCIA, CS/CJ I/F Port, CS/CJ Ethernet Unit PCI, ISA, PCMCIA, CS/CJ I/F Port, CS/CJ Ethernet Unit	Description Use 3G8F7-DRM21, 3G8F5-DRM21, 3G8E2-DRM21, CS/CJ/I/F Port, CS/CJ Ethemet Unit.
Instell5hield	K Next > Cancel

8. The installation program will be started and the Configurator files will be copied to the destination directory. If the default program folder is acceptable, click the **Next** Button. To specify a different folder, select an existing folder or enter a new folder name and then click the **Next** Button.

Note The program folder name specified here will be registered under DeviceNet Tools in the Start Menu.

InstallShield Wizard			×
Select Program Folder Please select a program folder.			
Setup will add program icons to the Program Fo name, or select one from the existing folders list	older listed below. t. Click Next to c	You may type ontinue.	a new folder
Program Folders:			
DeviceNet Configurator			
Existing Folders:			
Accessories Administrative Tools Startup			
InstallShield	< <u>B</u> ack	<u>N</u> ext >	Cancel

9. The installation program will be started and the Configurator files will be copied to the destination directory.

InstallShield Wizard	×
Setup Status	
DeviceNet Configurator Setup is performing the requested operations.	
Installing:	
C:\\DMRDN\DeviceNet Configurator\Program\DeviceIdentity.dll	
44%	
InstallShield-	
	Cancel

10. The following window will be displayed when the installation of the Configurator completes.

InstallShield Wizard		
	InstallShield Wizard Complete Setup has finished installing DeviceNet Configurator on your computer.	
	< Back Finish Cancel	

11. Click the **OK** Button to exit the setup program. This completes installation of the Configurator.

# **SECTION 3 Basic Operation**

This section explains the basic operation of the DeviceNet Configurator.

3-1	Starting	g the Configurator and the Main Window	38
	3-1-1	Hardware List	39
	3-1-2	Network Configuration Window	40
3-2	Menu C	Commands	42
	3-2-1	Main Menu	42
	3-2-2	Tool Bar Icons	45
	3-2-3	Status Bar	46
3-3	Connec	ting to the Network	46
	3-3-1	Connecting to the DeviceNet Network	46
	3-3-2	Designating the Interface	48
3-4	Operati	on Flowcharts	59
	3-4-1	Offline Operation	59
	3-4-2	Downloading Saved Parameters	59
	3-4-3	Saving the Parameters for the Entire Existing Network	60

# 3-1 Starting the Configurator and the Main Window

Select *Program*, *DeviceNet Tools*, and *DeviceNet Configurator*, from the Start Menu if the default program folder name is used. The Configurator will start and the following Main Window will be displayed.



The following description provides information on the Main Window of the Configurator.

The Main Window consists of the hardware list and Network Configuration Window.

💐 Untitled - DeviceNet Configurator		
Network Edit View Device EDS File Tools	s <u>O</u> ption <u>H</u> elp	
] 🗅 📽 🔜 🗏 💂 🍇 🍇 📚 📎	⊕   X ha G X   <mark>1</mark> = ∰   X h G X   <b>2</b> = ∭   X h G X   <b>2</b> = ∭	X E
Image: State of the state		-2
Heady	_Ethemet:FIN5NodeNo:#U  UnitNo:UUU.UUU.UT IP:U.U.U.U:9600   O Off-lineNUM	

Hardware list A list of devices that can be added to the network. Network Configuration Window A virtual network.

# **3-1-1 Hardware List**

The hardware list shows the devices in which EDS files have been installed. To simplify searching, it is divided into Vendor and Device Type groups.

The same devices are found in each of these groups, and can be selected from whichever one is easiest to search.

Devices can be added to the network by selecting them from the hardware list and dragging them into the Network Configuration Window.



Vendor Group When the Configurator is first installed, the Vendor group contains only OMRON hardware. If products (containing EDS files) from other manufacturers are installed, their list will automatically be built.

Device Type Group

The Device Type group consists of the following types of devices.

Device type	Description
Generic Device	General-purpose I/O devices. Many OMRON slaves currently belong to this category.
Communications Adapter	These serve as a network gateway. OMRON's DeviceNet Master Units, MULTIPLE I/O TERMINAL Communications Units, Digital Fiber Amplifiers, and Wireless Units currently belong to this category.
Human-Machine Interface	These devices function as human-machine interfaces. OMRON's PT Interface Unit currently belongs to this category.
AC Drives	These are AC drives. Inverters currently belong to this cate- gory.
Position Controller	These are devices with positioning functions, such as servo- motors and stepping motors. The Single-axis Positioner cur- rently belongs to this category.

Devices will not be displayed in the hardware list unless they have had an EDS file installed.

- **Note** The hardware list can be moved left/right and up/down, removed for use as a separate window, or closed.
  - To move or remove for use as a separate window Drag the top of the hardware list to move it. To remove it for use as a separate window, drag it to the center of, or outside, the Network Configuration Window.
  - To close
    - Click the Close button to close the window with the hardware list.
  - To display

Activate Hardware List from the View Menu.

### 3-1-2 Network Configuration Window

The Network Configuration Window is a virtual network. Devices dragged from the hardware list and dropped in this window will be added to the virtual network, where the devices can be setup or monitored.

It is possible to select either one of the following display modes in the Network Configuration Window.

- Network display: Large icons are displayed in the image of network connection devices.
- Detailed display: Device data is displayed in a report format.

The display mode is selected with the tool bar or from the View Menu.

### **Network Display** While the Network Display Mode is selected, the following Network Configuration Window will be displayed.

In Detailed Display Mode, click the Large Icons Icon or select Large Icons from the View Menu to display the network configuration.



Detailed Display In Network or solo

In Network Display Mode, click the **Detailed Display** Icon or select **Details** from the View Menu to display the following Network Configuration Window.

Node address								
Product name			Re	egis	stered node address	Devic	e type	
💐 Untitled.npf - DeviceNet Configurator								
<u>N</u> etwork <u>E</u> dit <u>V</u> iew <u>D</u> evice ED <u>S</u> File <u>T</u> ools	<u>O</u> ption <u>H</u> elp				Vendor name			Revision
] 🗅 😂 🖬 📕 👼 😽 🖓 💝	😂   X 🖻 I	a×∣•₌[		× I	। च च ⊨ ♦ 🖗 🖉		📽 🛛 🖬	24.   👪
×	Product Name	1	¥   :	##	Vendor	Device Type		Rev
⊡	📴 CS1W-DRI	M21 #	<b>#</b>		OMRON Corporation	Communicatio	ons Adapter	1.01
COMPON Corporation	dri 🧭	1-0D16 #	ŧ ‡	<b>#</b>	OMRON Corporation	Generic Devi	ce	1.03
Generic Device     Human Machine Interface	an the second se	1-MD16C #	<b>#</b> ‡	<b>#</b>	OMRON Corporation	Generic Devi	ce	1.03
	d DR1	11-ID16T #	<b>#</b> ‡	<b>#</b>	OMRON Corporation	Generic Devi	се	1.03
AC Drives	🗊 DRI	[1-MD32ML =	<b>#</b> ‡	<b>#</b>	OMRON Corporation	Generic Devi	се	1.03
e Generic Device ⊕ Generic Device ⊕	C DR1	[1-AD04 #	# i	Ħ	OMRON Corporation	Generic Devi	ce	1.03

**Note** By clicking the right mouse button in the window in network display or Detailed Display Mode, all the functions in the Edit or Device Menu can be accessed. Furthermore, by moving the mouse pointer to a device, an explanation for the device will be shown in a pop-up message.



# 3-2 Menu Commands

The following table describes all of the commands in the menus of the Configurator.

# 3-2-1 Main Menu

## Network Menu

Command	Description	Offline	Online
New	Creates a new network configuration.	OK	OK
Open	Opens a network configuration file.	OK	OK
Save	Saves the displayed network configuration.	OK	OK
Save As	The Save as command is the same as Save, but the Filename Specification Window is always displayed.	ОК	ОК
Connect	Connects the Configurator to the network	ОК	No
Disconnect	Disconnects the Configurator from the network	No	OK
Move to Upper Network	Displays the upper network.	No	ОК
Move to Lower Network	Displays the lower network.	No	ОК
Upload	Uploads device data from the network to the Configurator.	No	ОК
Download	Downloads device data from the Configurator to network devices.	No	ОК
Compare Struc- ture	Compares the current network configuration with the actual configuration of the network to which the Configurator is connected.	No	ОК
Export	Outputs a file in the CSV file format containing the contents of the detailed display.	ОК	ОК
Import	Imports network configuration files created on version 1 of the Configurator.	ОК	ОК
Print	Prints device parameters and the I/O comment list.	OK	OK
Setup Printer	Sets up the printer.	OK	OK
Exit	Exits the Configurator program.	OK	ОК

## Edit Menu

Command	Description	Offline	Online
Cut	Cuts devices to paste them to the clipboard.	OK	OK
Сору	Copies devices to the clipboard.	OK	OK
Paste	Copies devices from the clipboard to the cursor position.	OK	OK
Delete	Deletes selected devices.	OK	OK
All	Selects all devices.	OK	OK

# View Menu

Command	Description	Offline	Online
Tool Bar	Turns the tool bar ON or OFF.	OK	OK
Status Bar	Turns the status bar ON or OFF.	OK	OK
Large Icons	Changes to Network Display Mode.	OK	OK
Details	Changes to Detailed Display Mode.	OK	OK
Hardware List	Displays or hides the hardware list.	OK	OK

### **Device Menu**

Com	mand	Description	Offline	Online
Parameter	Wizard	Sets desired parameters for supported devices in the wizard format.	ОК	ОК
	Edit	Edits device parameters.	OK	OK
	Open	Reads parameters from device parameter files.	OK	OK
	Save As	Saves device parameters in files.	ОК	OK
	Upload	Uploads device parameters from network devices.	No	OK
	Download	Downloads device parameters to network devices.	No	OK
	Compare	Compares network devices and device parameters.	No	OK
Monitor		Monitors supported network devices.	No	OK
Reset		Resets supported network devices.	No	OK
Register to Device	the other	Registers devices with other devices.	ОК	ОК
Export		Exports I/O comments or device parameters in other file formats used for supported devices only.	ОК	ОК
Import		Imports device parameter files created on version $1.\square\square$ of the Configurator.	ОК	ОК
Change No	de Address	Changes device node addresses.	ОК	ОК
Change Device Com- ment		Changes device comments.	ОК	ОК
Edit I/O Cor	nment	Edits I/O comments.	ОК	ОК
Property		Displays device properties.	OK	ОК

**Note** By clicking the right mouse button in the Network Configuration Window, some of the functions in the Device Menu and Edit Menu can be accessed.

### EDS File Menu

Command	Description	Offline	Online
Install	Adds new EDS files to the EDS list.	OK	OK
Create	Creates and adds new EDS files to the EDS list.	OK	OK
Delete	Deletes EDS files from the EDS list.	OK	OK
Save As	Saves EDS files in the EDS list with new EDS file names.	OK	OK
Find	Searches for specified EDS files in the EDS list.	OK	OK
Add to Network	Adds hardware from the EDS list to the DeviceNet network.	OK	OK
Property	Displays EDS file properties.	OK	OK

**Note** By clicking the right mouse button in the Hardware List Window, all the functions in the EDS File Menu can be accessed.

# <u>Tools Menu</u>

Command	Description	Offline	Online
Setup Parameters	Makes parameter settings using explicit messages.	OK	OK
Setup Node Address/Baud rate	Makes node address and transmit rate settings for network devices.	ОК	ОК

## **Option Menu**

Command	Description	Offline	Online
Select Interface	Selects the interface used for the Configurator.	OK	OK
Edit Configuration File	Edits a variety of set files.	ОК	ОК

### Menu Commands

## Section 3-2

Command	Description	Offline	Online
Setup Monitor Refresh Timer	Sets the monitor refresh timer (screen refresh intervals in device monitoring operation).	ОК	ОК
Install Plug-in Module	Installs an Expansion Module.	ОК	ОК
Parameter Auto Update when I/O Size Changes	Automatically changes the I/O size of a slave registered in the master scan list when the size is changed. The default setting is OFF (not update), and should normally be kept OFF.	ОК	ОК

## Help Menu

Command	Description	Offline	Online
Торіс	Searches for help topics.	OK	OK
About	Displays version information on the Configurator.	OK	OK

# 3-2-2 Tool Bar Icons

Frequently used functions can be easily selected with the following icons.

lcon	Description	Equivalent menu command
D	Creates a new network configuration.	Network-New
<b>F</b>	Opens a network configuration file.	Network-Open
R	Saves the displayed network configuration.	Network-Save
垦	Connects the Configurator to the network.	Network-Connect
₩.	Removes the Configurator from the network.	Network-Disconnect
4.50 1.50	Displays the upper network.	Move to Upper Network
	Displays the lower network.	Move to Lower Network
	Uploads device data from the network to the Configurator.	Network–Upload
-	Downloads device data from the Configurator to network devices.	Network-Download
8	Prints.	Network-Print
ж	Cuts devices.	Edit-Cut
Ē	Copies devices.	Edit–Copy
Ê	Pastes devices.	Edit-Paste
×	Deletes devices.	Edit-Delete
<u>a</u> <u>a</u>	Displays the screen in network configuration mode.	
	Displays the screen in detailed report mode.	
Ň	Device parameter wizard.	Device-Parameter-Wizard
	Edits device parameters.	Device-Parameter-Edit
<b>F</b>	Reads parameters from device parameter files.	Device-Parameter-Open
	Saves device parameters in files.	Device-Parameter-Save As
۲	Uploads device parameters.	Device-Parameter-Upload
	Downloads device parameters.	Device-Parameter-Download
	Monitors supported network devices.	Device-Monitor
	Edits I/O comments.	Device-Edit I/O Comment
F	Displays device properties.	Device-Property
<u>_</u>	Adds new EDS files.	EDS File-Install
×	Deletes EDS files.	EDS File-Delete

Icon	Description	Equivalent menu command
	Saves new EDS files with new EDS file names.	EDS File-Save As
<u><u>م</u>ر</u>	Adds the selected device to the Network Configuration Window.	EDS File-Add to Network
熱	Searches for EDS files.	EDS File-Find
3	Displays EDS file properties.	EDS File-Property

# 3-2-3 Status Bar

The status bar displays menu explanations and the status of the Configurator.

### When Connecting Via DeviceNet Dedicated Board



Indicator color	Status
Gray	Offline
Blue	Online
Red	Bus OFF error

**Note** If a Bus OFF error is detected, set the Configurator offline and set it online again.

# **3-3** Connecting to the Network

# 3-3-1 Connecting to the DeviceNet Network

The Configurator can be connected to the DeviceNet network online through either of the following interfaces.

1,2,3...1. Dedicated Board/Card3G8F5-DRM21: Dedicated ISA Board3G8E2-DRM21: Dedicated PCMCIA Card



 Connection can also be made from the COM port of the computer to the serial communications port (peripheral port, RS-232C port) of a CS/CJ-series CPU Unit or to the serial communications port (RS-232C port or RS422A/485 port) of a Serial Communications Board/Unit. In this case, a CS/CJ-series DeviceNet Unit (CS1W-DRM21 or CJ1W-DRM21) is necessary for connection to the DeviceNet network.



If the Configurator is Ver. 2.11 or newer, it is possible to connect to the DeviceNet network across (up to three) network levels via serial communications, as shown below.



3. Connection can be made from the Ethernet port of the computer via a CS/ CJ-series Ethernet Unit. In this case also, a CS/CJ-series DeviceNet Unit (CS1W-DRM21 or CJ1W-DRM21) is necessary for connection to the DeviceNet network.



If the Configurator is Ver. 2.11 or newer, it is possible to connect to the DeviceNet network across (up to three) network levels via Ethernet, as shown below.



## 3-3-2 Designating the Interface

Connecting to the Network

Use the following procedure to designate the interface.

Note The interface is designated together with online connection settings.

- 1,2,3...1. Select *Select Interface* from the Option Menu. (The interface presently selected is indicated with a check mark.)
  - 2. Select the interface from those displayed on the menu.
    - Select DeviceNet IF Card if a dedicated Board or Card is used.
    - Select SYSMAC CS1 IF Port for a serial port connection.
    - Select SYSMAC CS/CJ Ethernet Unit IF for an Ethernet connection.
    - Note The interface cannot be changed when the Configurator is online. Select **Network** and **Cancel Connection** first. Then select the interface offline.
  - 3. Select Network and Connection from the menu bar.

The window corresponding to the designated interface will be displayed.

### Selecting DeviceNet IF Card

4. The DeviceNet Board Selection Window will open.



 Select the DeviceNet interface to be used and click the OK Button. The Setup Interface will appear.

Setup Interface 🛛 🗙
Card Name
Driver250
MAC ID
0 *
Baudrate
C 125K Bit/s C 250K Bit/s C 500K Bit/s
OK Cancel

For the 3G8E5-DRM21 or 3G8E2-DRM21

Set the following items.

### Card

The card name that was set with the driver setup tool is specified.

### MAC ID

Specify the node address of the Configurator. Set a unique node address.

#### **Baud Rate**

Set the same baud rate as the one set for the DeviceNet network. A network communications error may result if the same baud rate is not used.

6. Click the **OK** Button.

A confirmation window will appear. Click the **OK** Button. The Configurator will be connected to the DeviceNet network. When the Configurator is connected online, the color of the status indicator on the status bar will change to blue and *Online* will be displayed.

Selecting SYSMAC CS/CJ IF Port

- Perform steps 1 through 3 of the above procedure.
- 4. The Setup Interface Dialog Box will appear if a SYSMAC CS/CJ IF Port is specified.

Setup Interface	X
Interface :	Toolbus 🔽
Network Address :	0 -
Node Address :	0 .
Unit No. :	0 -
COM Port :	COM1
Baud Rate :	115200 Bit/s
Data Length :	8Bits 💌
Parity :	No
Stop Bit :	1Bit 💌
ОК	Cancel

Set the following items.

### Interface

Select the interface from the following settings for the serial communications mode at the serial communications port of the CS/CJ-series PLC.

- Host Link
- Toolbus

### **Network Address**

Input the network address of the DeviceNet Unit to be connected. Input if crossing networks past the serial communications of the CS/CJ-series CPU Unit. If not crossing networks, input 0.

### **Node Address**

Input the node address of the DeviceNet Unit to be connected. Input if crossing networks past the serial communications of the CS/CJ-series CPU Unit. If not crossing networks, input 0.

### Unit No.

Input the unit number within the following range for the CS1W-DRM21 DeviceNet Unit as a CPU Bus Unit. The unit number must be the same as that set with the thumbwheel switches on the front panel of the CS1W-DRM21.

• 0 to 15

### COM Port

Select the COM port of the computer running the Configurator within the following range.

• Select from the list of usable COM ports.

### Baud Rate

Select the baud rate for the serial communications port of the CS/CJ-series PLC from the following settings.

- 9,600 bps, 19,200 bps, 38,400 bps, or 115,200 bps
  - Note The Tool bus and Host Link are different to each other in available baud rate. For details, refer to the *CS/CJ Operation Manual*.

#### Data Length

Select the data length for the serial communications port of the CS/CJ-series PLC from the following settings if Host Link is used.

• 8 or 7 bits

#### Parity

Select the parity for the serial communications port of the CS/CJ-series PLC from the following settings if Host Link is used.

• None, even, or odd

### **Stop Bits**

Set the number of stop bits for the serial communications port of the CS/CJseries PLC from the following settings if Host Link is used.

- 1 bit or 2 bits
- 5. Click the **OK** Button.

A confirmation window will appear. Click the **OK** Button. The Configurator will be connected to the DeviceNet network. When the Configurator is connected online, the color of the status indicator on the status bar will change to blue and *Online* will be displayed.

**Note** When using a CS1W-CN226/626 or XW2Z-200S-CV/500S-CV Connecting Cable to make a serial connection to a CS-series CPU Rack containing a CS1W-DRM21, Host Link is selectable as the interface, but connection will not be possible if it is selected. In this case, select Toolbus.

However, when using an XW2Z-200S-CV/500S-CV Connecting Cable or CS1W-CN114 + CQM1-CIF02, connection is possible by selecting Host Link as the interface.



**Note** When Host Link is selected, it may take several minutes to download from the network. Selecting Toolbus is therefore recommended for serial connections.

Selecting SYSMAC CS/CJThe Configurator can be connected directly to the DeviceNet network viaEthernet Unit IFEthernet through a CS/CJ-series Ethernet Unit and a CS/CJ-series DeviceNet<br/>Unit.

**Note** The Ethernet connection can be made only with the CS/CJ-series Ethernet Unit and CS/CJ-series DeviceNet Unit combination. Connection by the combination of other PLC-series Units is not possible. Also, if multiple PLCs (using Ethernet Units and DeviceNet Units) are connected via Ethernet, connection can be made to a designated DeviceNet network by switching the connection target. The IP address of the Ethernet Unit and the unit number of the DeviceNet Unit are designated to register the DeviceNet network to be connected.



- Note 1. DeviceNet networks can be switched by designating the registered name for the network to be connected. The name of the DeviceNet network to be connected can be registered as follows.
  - Designate the DeviceNet Unit network address, node address, and Link Unit number.
  - Designate the FINS node address of the personal computer containing the Configurator.
  - 2. Connection is not possible through the combination of a CVM1/CV Ethernet Unit and CVM1/CV DeviceNet Unit. It is also not possible through the combination of a C200HX/HG/HE PLC Card Unit (Ethernet Set) and C200HX/HG/HE DeviceNet Master Unit.
  - 3. Configurator Ver. 2.10 allows Ethernet connection to the DeviceNet network across one network level only. DeviceNet networks cannot be connected to across multiple network levels. (For example, configurations such as Ethernet  $\rightarrow$  Controller Link  $\rightarrow$  DeviceNet, or Host Link (Toolbus) Ethernet  $\rightarrow$  DeviceNet are not possible.)

With Configurator Ver. 2.11, it is possible to cross network levels as long as FINS message communications are supported by the networks.

For connection via Ethernet, the DeviceNet network(s) to be connected must be registered in advance. Up to 20 DeviceNet networks can be registered, using the following procedure.

1,2,3... 1. Select Network and Connect.

Designate the IP address and UDP port number.

**Registering DeviceNet** Networks to Be Connected

2. The Setup Interface Dialog Box shown below will appear.

Setup Interface	×
- Host(PC) Information	
Host Name :	PC1
IP Address :	10.3.74.142
Network Address :	000
Node Address :	142
- Remote Information -	
Registration Name : — DeviceNet Unit	<b></b>
Network Address :	000
Node Addres :	000
Unit No. :	01
Ethernet Unit	
Port No. :	9600
IP Address :	10.3.74.37
	<u>S</u> etup
OK	Cancel

### Setup Interface Dialog Box

Host (PC) Infor- mation	The settings for the computer running the Configurator will be displayed.		
	Host Name	The name of the computer will automati- cally appear.	
	IP Address	The IP address of the computer will auto- matically appear.	
	Network Address	The FINS network address set for the computer will appear. (The value set with the Registration of the connection Dialog Box, after clicking the Setup Button as described in following step (step 3), will appear.)	
	Node Address	The FINS node address set for the com- puter will appear. (The value set with the Registration of the connection Dialog Box, after clicking the <b>Setup</b> Button as described in the following step (step 3), will appear.)	

3. Click the **Setup** Button. The Registration of the connection Dialog Box shown below will be displayed.

Registration of the connection	
Registration Name : Host(PC) Information Network Address : Node Address : 0	
Remote Information DeviceNet Unit Network Address : 0 * Node Address : 0 * Unit No. : 0 *	
Ethernet Unit Port No. : 9600 IP Address : 0 . 0 . 0 . 0	
Add Delete Register List Registratio Node Unit Port I	
Close	

Registration Name	The registration name for the DeviceNet network to be connected is set here. Up to 20 networks can be set, each up to 25 characters in length.		
Host (PC) Information	This is the setting for the Configurator computer.		
	Network Address	The user inputs the FINS network address for the computer here. Set the same address as the one set for the Ethernet Unit. Input 0 if no other address is going to be set.	
	Node address	The user inputs the FINS node address for the com- puter here.	
Remote Infor- mation	These are the settings for the DeviceNet Unit and Ethernet Unit that are used to connect with the DeviceNet network.		
	DeviceNet Unit	Network Address	Input the FINS network address for the DeviceNet to be connected. Input if crossing networks past the direct Ethernet connection. If not crossing networks, input 0.
		Node Address	Input the node address for the DeviceNet to be connected. Input if crossing networks past the direct Ethernet connection. If not crossing networks, input 0.
		Unit No.	Input the unit No. of the DeviceNet to be connected, for use as a CPU Bus Unit.
	Ethernet Unit	Port No.	Input the Fins UDP port No. for the Ethernet Unit.
		IP Address	Input the IP address for the Ethernet Unit.

#### **Registration of the connection Dialog Box**

### Setting the Network Address in Host (PC) Information:

Set the FINS network address for the computer. The computer running the Configurator uses OMRON's FINS communications service to connect to the DeviceNet network via Ethernet. Thus, the computer must have not only an IP address, but also a FINS network address and node address. The relationship between the FINS node address and the IP address is specified by the address conversion in the Ethernet Unit.

Obtain the FINS node address corresponding to the IP address (See note.) of the computer from the address conversion of the Ethernet Unit. Refer to *Information on Address Conversion* on page 57 for details.

#### Setting the Node Address in Host (PC) Information:

Set the FINS node address for the computer.

Obtain the FINS node address corresponding to the IP address (See note.) of the computer from the address conversion of the Ethernet Unit. Refer to *Information on Address Conversion* on page 57 for details.

**Note** The computer's IP address is automatically displayed in the Host (PC) Information setting in the Setup Interface Dialog Box.



# Setting the Network Address in the DeviceNet Unit Section Inside Remote Information

Set the FINS network address of the DeviceNet Unit on the DeviceNet network to be connected. Input if crossing networks past the direct Ethernet connection. If not crossing networks, input 0.

# Setting the Node Address in the DeviceNet Unit Section Inside Remote Information

Set the node address of the DeviceNet Unit on the DeviceNet network to be connected. Input if crossing networks past the direct Ethernet connection. If not crossing networks, input 0.

# Setting the Port No. in the Ethernet Unit Section Inside Remote Information

Set the UDP port No. for FINS communication service by the Ethernet Unit. Set it to the same value that is set in the CPU Bus Unit System Setting Area inside the CPU Unit to which the Ethernet Unit is mounted. Normally, this setting is 9600.

# Setting the Unit. No in the DeviceNet Unit Section Inside Remote Information

Set the unit No. (0 to F) of the DeviceNet Unit for CPU Bus Unit use on the DeviceNet network to be connected.

# Setting the IP Address in the Ethernet Unit Section Inside Remote Information

Set the IP address for the Ethernet Unit. For the CS-series Ethernet Unit, the value that is set with the rotary switch on the back of the Unit is input here. For the CJ-series Ethernet Unit, the value that is set in the Allocated Words in the DM Area, or in the System Setup for CPU Bus Units, is input here.

- **Note** For details on setting the IP address for the Ethernet Unit, refer to the SYS-MAC CS/CJ-series Ethernet Units Operation Manual (catalog No. W343).
  - Click the Add Button. The set value is registered and displayed in the Register List.
    - Registration Name: The registration name for the DeviceNet network to be connected to.
    - Node: The FINS network address and FINS node address. (The third digit is always 0.)
    - Unit: The FINS network address, FINS node address, and unit number of the DeviceNet Unit.
    - Port: The FINS UDP port No. for the Ethernet Unit.

Selecting the Registration

Information on Address

Conversion

Name

- IP Address: The IP address for the Ethernet Unit.
- 5. To exit the registration procedure, click the **Close** Button. The Setup Interface Dialog Box will be displayed again.

Select the DeviceNet network to be connected to from among those in the Registration Name column that were set with the Setup Interface Dialog Box.

- 1. Select the desired Registration Name from the Registration Name dropdown list in Remote Information. The following parameters are displayed for the selected Registration Name in Remote Information.
  - Network Address: The FINS network address for the DeviceNet Unit.
  - Node Address: The node address for the DeviceNet Unit.
  - Unit No.: The unit No. DeviceNet Unit.
  - Port No.: The FINS UDP port No. for the Ethernet Unit.
  - IP Address: The IP address for the Ethernet Unit.
- 2. Click the OK Button.

A confirmation window will appear. Click the **OK** Button. The Configurator will be connected to the DeviceNet network. When the Configurator is connected online, the color of the status indicator on the status bar will change to blue and *Online* will be displayed.

#### Ethernet Unit Address Conversion Method

The FINS communications service is used inside the PLC. Because there is no IP address information added to the FINS communications service, the IP address is determined from the FINS node address in order to send a FINS message (command or response) with the PLC's Ethernet Unit.

The Ethernet Unit address conversion is done by either of the three following methods (set in the System Setup for CPU Bus Units of the CPU Unit to which the Ethernet Unit is mounted).

- IP address table
- Automatic generation
- Combined method

The computer's FINS node address calculation for each address conversion method is described below.

Address conversion method	Computer's FINS node address calculation
IP address table	The FINS node address is set according to the computer's IP address based on a preset correspondence table called an IP address table.
Automatic generation	The FINS node address is automatically generated from the computer's IP address and the subnet mask.
Combined method	If the computer's IP address has been set in the IP address table, the FINS node address in the table is set. If the computer's IP address has not been set in the IP address table, a FINS node address that is automatically generated from the computer's IP address, or from the subnet mask, is set.

**Note** For details on address conversion, refer to the SYSMAC CS/CJ-series Ethernet Units Operation Manual (catalog No. W343).
## Obtaining the FINS Node Address from the Automatic Generation Method of Address Conversion

The FINS node address corresponding to the computer's IP address is obtained from the rightmost byte of the computer's IP address and the rightmost byte of the network address as follows:

FINS node address = The rightmost byte of (the Ethernet Unit's IP address AND the Ethernet Unit's subnet mask) XOR the rightmost byte of the computer's IP address.

This can be easily calculated using the Windows Calculator.

**Note** Address conversion from the Ethernet Unit's automatic generation method is obtained as follows.

Computer's IP address = (the Ethernet Unit's IP address AND the Ethernet Unit's subnet mask) OR FINS node address.

Accordingly, the FINS node address can be determined by taking the XOR of the network address and the computer's IP address.



Calculation example:

Computer's IP address = 192.168.10.100 Ethernet Unit's IP address = 192.168.10.119 Ethernet Unit's subnet mask = 255.255.255.224 FINS node address = (119 AND 224) XOR 100 = 4



## **3-4 Operation Flowcharts**

The following flowcharts show the flow of Configurator operations.

## 3-4-1 Offline Operation

Prior to constructing the actual network, it can be designed and the master and slave parameters can be set with the Configurator.

The set parameters can be saved as a network configuration file. The network configuration file contains the parameters of all of the devices on the network.



## 3-4-2 Downloading Saved Parameters

The parameters for all of the devices on the network can be set (downloaded) from a network configuration file saved in advance.



## **3-4-3** Saving the Parameters for the Entire Existing Network

The parameters for all of the devices on the network can be saved for use in maintenance.



## SECTION 4 Editing Device Parameters

This section explains how to enable remote I/O communications by creating device parameters for a virtual network constructed with the DeviceNet Configurator.

4-1	Adding	Devices to the Virtual Network
	4-1-1	Adding Devices
	4-1-2	Changing Node Addresses
	4-1-3	Reading the Network Configuration from the Network
	4-1-4	Editing Devices
4-2	Setting	Slave Parameters
	4-2-1	Editing Slave Parameters
	4-2-2	Checking and Setting I/O Size
4-3	Adding	Slaves to the Master
	4-3-1	Automatic I/O Area Allocation with Registration       7
	4-3-2	Adding Slaves
4-4	Setting	Master Properties
	4-4-1	CS1W-DRM21
	4-4-2	CVM1-DRM21-V1 or C200HW-DRM21-V1
4-5	Editing	Master Parameters
4-6	Parame	ter Wizard
4-7	Master	Parameter Editing Details (Tab Descriptions)
	4-7-1	Edit Device Parameters on CS1W-DRM21/CJ1W-DRM21 8
	4-7-2	Editing Device Parameters on CVM1-DRM21-V1 and
		C200HW-DRM21-V1
	4-7-3	Canceling Slave Registration with the Master
	4-7-4	Automatic Allocation with Registration
4-8	Manual	I/O Allocations
	4-8-1	I/O Allocation Tab Page
	4-8-2	Changing the First Address of Output/Input Block
	4-8-3	Allocating I/O 8
	4-8-4	Slave Information
4-9	Advanc	ed Settings (Connection, Communication Cycle Time,
	Slave F	unction Settings, Etc.)
	4-9-1	Advanced Settings
	4-9-2	Communication Cycle Time Settings
	4-9-3	Setting Message Timers
	4-9-4	Slave Function Settings
	4-9-5	Setting/Canceling Explicit Message Communications
	4-9-6	Starting Remote I/O Communications
4-10	Creating	g and Editing I/O Comments
4-11	Display	ing Device Properties 10
	4-11-1	Property Window Common to All Devices
	4-11-2	I/O Information Inherent to Each Slave Device
	4-11-3	Information for Master

## 4-1 Adding Devices to the Virtual Network

This section describes how to add devices (master and slave devices) to the virtual network in the Network Configuration Window of the Configurator.

## 4-1-1 Adding Devices

To add a device, drag the device from the Hardware List Window and drop it on the Network Configuration Window.

Select the device from the Hardware List Window on the left and then drag and drop it on the Network Configuration Window on the right.

#### Adding a Master



#### **Adding Slaves**

Drag and drop the slaves in sequence in the same way.



#### Example of Addition by Dragging

💭 Untitled - DeviceNet Configurator						×
Network Edit View Device EDS File Tools Q	option <u>H</u> elp					
D 😂 🖬   £ 👼   🌤 🏷   ⊕   X	ħ <b>C</b> X  ₽ <u>₽</u> #	i   🖄 🗉   🗉	⊛ ⊕   ,	*   #   #   4	9 X 🖬   2,	3
	#00 CS1W-DRM21	#01 DRT1-OD16	#02 DRT1-MD16C-1	#03 DRT1-ID16-1	#04 DRT1-ID08	
				o	O	

**Note** A device can be also added by selecting the device in the hardware list on the left pane, clicking the right mouse button, and selecting *Add to Network* or by selecting *Add to Network* from the EDS File Menu.

## 4-1-2 Changing Node Addresses

Device - Change Device ID Unused node addresses between 0 and 63 will be allocated in sequence to the respective devices that have been added.

To change the allocated node addresses, use the following procedure.

- *1,2,3...* 1. Select the device for which the node address is to be changed.
  - 2. Select *Device* and *Change Device ID* or click the right mouse button and select *Chance Device ID* from the pop-up menu.

The following window will be displayed.

T	arget Device	×
	Target Node Address : 🚺 📺 Setup Range 0 - 63	
	OK Cancel	

3. Click the **OK** Button to change the node address.

Repeat the above steps and add all the necessary devices to the Network Configuration Window.

## 4-1-3 Reading the Network Configuration from the Network

#### Network - Upload

Use the following procedure to read the network configuration from the actual network.

- 1,2,3... 1. Place the Configurator online.
  - 2. Select *Network*.
  - 3. Select Upload.

The upload operation will start and the following window indicating the uploading progress will be displayed.



The user can interrupt the upload operation by clicking the **Cancel** Button.

4. The network devices will be displayed in sequence in the Network Configuration Window.



5. After the upload operation completes, add any additional devices are required using the procedure explained in *Adding Devices*.

## 4-1-4 Editing Devices

## **Deleting Devices**

Edit - Delete

1,2,3...

To delete network devices, use the following procedure.

- . 1. Select the devices. (A number of devices can be selected at one time.)
  - 2. Select *Edit*.
  - 3. Select Delete.
  - 4. A confirmation window will be displayed. Click the **OK** Button to delete the devices.

If slaves registered with the master are deleted, the slave registration with the master will be lost and the slaves will be purged from the scan list automatically.

## **Copying Devices**

Edit - Copy

To copy the network devices, use the following procedure.

- 1,2,3... 1. Select the devices. (A number of devices can be selected at a time.)
  - 2. Select Edit.
  - 3. Select Copy.

The devices including the set parameters will be copied to the clipboard in the Configurator.

#### Pasting Devices

Edit - Paste

To paste a device from the clipboard in the Configurator, use the following procedure.

- *1,2,3...* 1. Select the network device existing at the insertion position.
  - 2. Select *Edit*.
  - 3. Select Paste.

An unused node address will be allocated to the network device that was added.

#### **Changing Device Comments**

**Device - Change Device Comment** A device comment can be added to each network device that has been registered. The product name is by default allocated as the device comment of each network device. To change the device comment, use the following procedure.

- *1,2,3...* 1. Select the network device.
  - 2. Select Device.
  - 3. Select Change Device Comment.

The following window will be displayed.

Change Device Comment	×
C200HW-DRT21	
OK Cancel	

4. Input the device comment and click the OK Button.

The device comment will be displayed by moving the mouse pointer to the position of the device in the Network Configuration Window.

**Note** In the Network Display Mode, the device comment will be displayed when the mouse pointer is moved to the icon. In the Detailed Display Mode, the device comment will be displayed in the Comment column.

#### **Changing Device Positions**

To change the display position of a network device, use the following procedure.

- *1,2,3...* 1. Drag the network device with the right mouse button.
  - 2. Drop the network device at the desired position occupied by another network device.
  - 3. The dragged network device will move to the specified position.

• The following example shows a network device with node address #00 is dragged and dropped on node address #02.



• The following example shows a network device with node address #01 is dragged and dropped on node address #02.



## 4-2 Setting Slave Parameters

## 4-2-1 Editing Slave Parameters

## Device - Parameter - Edit

The parameters of some slaves can be set. However, doing so may cause the I/O size to change. Because of this, the slave parameters must all be set before setting the master's parameters.

EDS files are required for setting the parameters. If there are no EDS files available, they must be obtained from the device manufacturer.

**Note** Slaves must be added to the Network Configuration Window to edit the parameters. Creating the network configuration will be explained later in this section.

To edit the parameters, use the following procedure.

- *1,2,3...* 1. Select the device.
  - 2. Select *Device*, *Parameter*, and *Edit*.
  - 3. The following warning will be displayed if there are no editable parameters existing.

Slaves for which this message is displayed do not require parameter setting.

DeviceNe	t Configurator 🛛 🛛 🔀
	Setup possible parameters not exist,
	ОК

The following window will be displayed if editable parameters exist.

dit Device Parameters		×	
Parameter Group : All parame	eters 💌		<ul> <li>Parameter group selection</li> </ul>
Parameter Name	Value		
0001 Set Point	OEU		
0002 Alarm 1	OEU		Developmenter
0003 Alarm 2	0 EU		- Parameter names
0004 Alarm 3	0 EU		
0005 ProportionalBand	10.0 %FS		— Parameter set values
0006 Integral Time	233 Sec		i alameter set values
0007 Derivative Time	40 Sec		
0008 Cool Coefficient	1.00		
0009 Dead Band	0.00 %FS		
0010 Manual Reset ノ	50.0 %	-	
Setting Range: SP Setting Low to SP Setting High	▲ Default : 0 EU Min : -1999 EU Max : 9999 EU		<ul> <li>Parameter help</li> </ul>
Upload Download	Compare	<u>H</u> eset 🗕	<ul> <li>Device reset</li> <li>Usable only when</li> </ul>
D <u>e</u> fault Setup	ОК	Cancel	onine.
	Compare paramete	rs Usable o	nly when online.
Downloa	ad parameters Usable c	only when or	line.
Upload parameter	Usable only when onlin	e.	

Reset parameters to default settings Usable only when online.

4. Select the parameter and press the **Enter** Key or double-click the parameter.

If a push-pin icon is displayed next to a parameter name, it is a read-only parameter and cannot be edited.

Parameter Name
🧐 0007 State
🧐 0008 Drive Running Forward
🧐 0009 Drive Running Reverse
🕼 0010 Drive Ready
🖉 0011 Drive Fault

The set value will be changed as shown below according to the parameter input type.

### Numerical Input within Certain Range

0004 Port1 header code	02 (STX)
0005 Port1 Delimiter code	03 (ETX)
0006 Port1 Data size	0
0007 Port2 character format	Data=7,Parity=Even ,Stop=2
0008 Port2 frame format	XXXXX100

Set the value between the maximum and minimum values.

#### **Selection from Limited Items**

0005 Port1 Delimiter code	03 (ETX)
0006 Port1 Data size	0 byte
0007 Port2 character format	Data=7,Parity=Even ,Stop=2
0008 Port2 frame format	Data=7,Parity=odd ,Stop=1
0009 Port2 baud rate	Data=7,Parity=non_Stop=1
0010 Port2 header code	Data=7,Parity=Even ,Stop=2
- Help	Data=7,Parity=non ,Stop=2
	Data=8,Parity=even ,Stop=1
Set the character format	Data=8,Parity=odd ,Stop=1
	Data=8,Parity=non ,Stop=1
	Data=8,Parity=non_Stop=2

Select the set value from the set value list.

### **ON/OFF Settings**

0006 Port1 Data size	0 byte
0007 Port2 character format	Data=7,Parity=Even ,Stop=2
0008 Port2 frame format	Header code disable Bit8
0009 Port2 baud rate	Delimiter code disab
0010 Port2 header code	Flow control disable Bit10
	🗖 Bit3 🗖 Bit11
Help	🗖 Bit4 🗖 Bit12
Enable (0) or disable (1) header,	🗖 Bit5 🗖 Bit13
trailer, now control processing.	🗖 Bit6 🗖 Bit14
	🗖 Bit7 🗖 Bit15

Select the item and turn it ON or OFF.

- 5. Press the **Enter** Key to input the set value. Press the **ESC** Key to cancel the change.
- 6. Click the **OK** Button when all the items have been edited.

- **Note** There is a special setting window for the following slaves. For information on setting parameters for these slaves, refer to their respective operation manual.
  - DRT1-COM MULTIPLE I/O TERMINAL
  - WD30-M DeviceNet Wireless Units
  - E3X-DRT21 Fiber Amplifier Sensor Communication Units

## 4-2-2 Checking and Setting I/O Size

### **Checking I/O Size**

Device - Property - I/O Information

General 1/0 Info	erty ormation				×
IN[Bit-Strobe] a	llocated to #(	)0. ( 3332 : B	it00 )	-	A. Y
Connection	Out/In	Size	Help		
Poll	Out	0 Bytes			
	In	0 Bytes			
🖉 Bit-Strobe	Out	0 Bytes			
	In	2 Bytes			
COS	Out	0 Bytes			
	In	0 Bytes			
Cyclic	Out	0 Bytes			
	In	0 Bytes			
<u> </u>	<u>G</u> et	/O Size from	the Sc	anlist.	
				Clo	se

The I/O size of this slave can be checked with the **I/O Information** Tab of the Property Window. Check to make sure that the I/O size listed is correct. To change it, use the following procedure.

## Setting the I/O Size

For slaves in which the I/O size can be changed by switches, or by tools other than the Configurator, the Configurator cannot be used to confirm that the I/O size is correct.

For these slaves, use the following procedure to directly input the I/O size.

- 1,2,3... 1. Select the desired slave, then select Device and Property.
  - 2. Select the I/O Information Tab.
  - 3. Click the Edit Button. The Edit I/O Size Dialog Box will be displayed.
  - 4. Input the correct I/O size for each connection. If the connection is not known, input the correct I/O size in the *Poll* settings.
  - 5. Click the **OK** Button.

Note The following are OMRON slaves.

- DRT1-AD04 Analog Input Terminal Input (4-word/2-word) can be changed by DIP switch.
- GT1-AD08MX Analog Input Unit for MULTIPLE I/O TERMINAL. Input (8-word/4-word) can be changed by DIP switch.
- NT-DRT21 Programmable Terminal DeviceNet (CompoBus/D) Interface Unit

The I/O data area can be set from 1 to 64 words with the PT system menu.

Use the same kind of input procedure for slaves of other manufacturers that allow the I/O size to be changed.

## **Obtaining I/O Size Information from the Master Scan List**

When connecting the Configurator to a network that is already operating, for example using OMRON Master Units with fixed allocation, the slave I/O size can be obtained from the master scan list for use in setting.

However, this is possible only with OMRON masters. Also, the slave I/O size cannot be obtained from the slave function of the CSW-DRM21.

Use the following procedure to obtain the I/O size from a master scan list.

- 1,2,3... 1. Select the desired slave, then select Device and Property.
  - 2. Select the I/O Information Tab.
  - 3. Click the Get I/O Size from the Scanlist Button.

#### Parameter Auto Update When I/O Size Changed

The function to automatically update the I/O size registered on the master scan list when the I/O size is changed can be switched on or off.

To automatically update the I/O size registered on the master scan list, select *Option* and *Parameter Auto Update When I/O Size Changed*.

When this function is enabled, the **v** mark will be displayed in the menu. The default is set to not automatically update.



**Note** If the function to automatically update the I/O size is enabled and there is no EDS file, some operations will cause the I/O size in the master scan list to be updated to zero. For this reason, the function to automatically update the I/O size should normally be disabled.

## 4-3 Adding Slaves to the Master

## 4-3-1 Automatic I/O Area Allocation with Registration

When a slave is registered to the master, it can automatically be allocated to the memory block set for I/O allocation.

The allocation, for both Out Size and In Size, will be made in the order of registration beginning with the unoccupied area of memory block 1. When memory block 1 is filled, the allocation will be made to memory block 2. Before adding slaves, set the area and range of the allocation memory block.

**Note** The allocation area can be changed later if desired.

Setting the Allocation Memory Block

- *1,2,3...* 1. Select the master, then select *Device*, *Parameter*, and *Edit*. The Edit Device Parameters Dialog Box is then displayed.
  - 2. Select the I/O Allocation (OUT) Tab.

Communica General	ition Cycle	Time   1/0 Alloc	Me ation(O	ssage Timer UT)	S  /0 A	lave Function llocation(IN)
# Pri	oduct	Size	Ch		C	Auto
🎾 #01 3F	88M-D	8 Byte	3200	:Bit00		Delete
						<u>E</u> dit
						Information
						Thomadon
Memory Bloc	k1		_	- Memory Blo	ck 2	
Ch	Proc	luct Name	•	Ch	Pro	duct Name 🔺
🂷 3200:Bit	00 #01	3F88M		💷 3400:Bi	t00	
🂷 3200:Bit	08 #01	3F88M		💷 3400:Bi	t08	
🂷 3201:Bit	00 #01	3F88M		💷 3401:Bi	100	
🂷 3201:Bit	08 #01	3F88M		💷 3401:Bi	108	
🂷 3202:Bit	00 #01	3F88M		💷 3402:Bi	100	
🂷 3202:Bit	08 #01	3F88M		💷 3402:Bi	108	
💷 3203:Bit	00 #01	3F88M		💷 3403:Bi	100	
🂷 3203:Bit	08 #01	3F88M		💷 3403:Bi	t08	
🂷 3204:Bit	00			💷 3404:Bi	t00	
🎹 3204:Bit	08		_	💷 3404:Bi	108	
🗰 3205-Rit	nn			3405-B	100	<b>_</b>
<u>S</u> etup				<u>S</u> etup		

- 3. Click the Setup Button for Memory Block 1.
- 4. Set the area, start word, and the number of words for the block.

Area :	I/O Relav
Start Word :	3200 Range : 0000 - 6143
Display Words	100 Range : 1 - 500 Ch

- 5. Set Memory Block 2 in the same way.
- 6. Select the I/O Allocation (IN) Tab, and make the memory block settings in the same way as for the I/O Allocation (OUT).

Note

- te 1. If a block is not to be used, select **Not Used** for the area setting.
  - Display Words refers to the number of words in the block displayed on the Configurator. This value is not downloaded to the master.
     If the occupied area of the block is less than 100 words when uploaded, Display Words will be set to 100 and displayed as such.

## 4-3-2 Adding Slaves

## **Dragging and Dropping**

The following three methods are supported for adding slaves.

Select the slave from the Network Configuration Window on the right and drag and drop it on the master icon.

When slaves are registered with the master, I/O words will be automatically allocated to the slaves in the order they are registered.



#### After Registration



When the slave is registered with the master, the master's node address will be displayed

#### Master Node Address Displayed after Registration



When the slaves are registered, the master's node address will be displayed (00 in the above case) with the prefix # after the symbol  $\blacksquare$  .

#### Clicking the Right Mouse Button and Selecting Register to other Device

A slave can also be added by selecting the slave in the Network Configuration Window, clicking the right mouse button, selecting **Register to other Device** or selecting **Register to other Device** from the Device Menu, and then selecting the master from the menu. The menu will appear on the right-hand side.

At the time of I/O allocation, the registered slave can be deleted or settings can be made for the registered slave by using the Parameter Wizard or by editing the parameters of the master.

- **Note** To select a device from the menu and register it, use the following procedure.
- **1,2,3...** 1. Select the device.
  - 2. Select Device.
  - 3. Select Register to other Device.

The following master candidates will be displayed in the **Register to other Device** Menu.

Parameter	
💒 Monitor	
<u>R</u> eset	
Register to other Device 🕨	#00 C200HW-DRM21-V1
	#03 C200HW-DRM21-V1
Export	#04 C200HW-DRM21-V1
Import	#05 C200HW-DRM21-V1
Ж Cu <u>t</u>	
Ва ⊆ору	
🗙 <u>D</u> elete	
Change Device <u>I</u> D	
Change Device Comment	
擅 Edit I/O <u>C</u> omment	
Property	

4. Select the device with which the slave is to be registered.

The slave registered will be automatically added to the scan list of the master and I/O will be allocated.

If the user attempts to register a slave that has already been registered with another device, the following confirmation window will be displayed. By clicking the **Yes** Button, the duplicated registration of the slave will be possible.

However, if the registration is duplicated, the slave will only be able to communicate with one device. If the same slave is registered to multiple masters, it will lead to communications errors.

DeviceNet	Configurat	or	×
	Already regi Will register?	ster with other (	device.
	Yes	No	

Registering in the Edit Device Parameters Window Slaves can be added and I/O areas allocated in the Edit Device Parameters Window. Refer to 4-7 *Master Parameter Editing Details (Tab Descriptions)* on page 80 for details.

## 4-4 Setting Master Properties

This section describes how to set properties for OMRON masters. The **Device – Property** Command is used.

## 4-4-1 CS1W-DRM21

To set the master and slave functions, use the following procedure.

- *1,2,3...* 1. Select the master from the Network Configuration Window on the right.
  - 2. Select Device and Property.

C51W-DRM21 Property		×
Master I/O Information	1/0 Information	PLC Information
General	On	it Fuction
🔽 Enable Master Func	tion	
🔲 Enabe Slave Function	on	
		Close

The following window will be displayed. Select the Unit Function Tab.

3. Select Enable Master Function and/or Enable Slave Function.

## 4-4-2 CVM1-DRM21-V1 or C200HW-DRM21-V1

To set the PLC model on which the Unit is mounted, use the following procedure.

- 1,2,3... 1. Select the device.
  - 2. Select Device.
  - 3. Select Property.

C200HW-DRM21-V1 Property	×
General 1/0 Information PLC Information	
PLC Model : C200HS Series	
Change Unit	
	Close

The following window will be displayed. Select the PLC Information Tab.

4. Click the **Change Unit** Button.

The following window will be displayed.

C	hange Unit	×
	Setup Model	
	Unit:	C200HW-DRM21-V1
	Corresponding PLC:	C200HS Series
		OK Cancel

- 5. Specify the PLC model. Click the **OK** Button.
- **Note** When the PLC model is changed, all the settings presently made will be cleared and default values will be set.

## 4-5 Editing Master Parameters

Device parameters are classified into master device parameters and slave device parameters. The following description provides information on how to edit master device parameters.

The master device parameters that can be edited include slave device I/O allocations to the master, communication cycle time settings, and connection settings. The following two methods can be used to edit parameters for the master.

**Parameter Wizard** The Parameter Wizard is an interactive interface that makes I/O allocations possible in order of node addresses.

**Editing Parameters** Parameters can be edited to allocate I/O and make settings as required, including the communication cycle time, connection, and device information check settings.

The Configurator allows settings, such as the following function settings, for the OMRON CS1W-DRM21 DeviceNet Unit, CJ1W-DRM21 DeviceNet Unit, CVM1-DRM21-V1 DeviceNet Master Unit, and C200HW-DRM21-V1 DeviceNet Master Unit.

Function	Setting	method	CS1W-DRM21	C200HW-DRM21-V1	
	Parameter Wizard ( <i>Device - Parameter -</i> <i>Wizard</i> )	Parameter editing ( <i>Device - Parameter -</i> <i>Edit</i> )		or CVM1-DRM21-V1	
I/O Allocations	Parameter Wizard	I/O Allocation (OUT) and I/ O Allocation (IN) Tabs	ОК	ОК	
Communication Cycle Time		Communication Cycle Time Tab	OK	ОК	
Connections		Advanced Setup Button in	ОК	No	
Device Information Compare		General Tab	ОК	No	
Message Timer		Message Timer Tab	ОК	No	
Slave Function		Slave Function Tab	ОК	No	
Startup Remote I/O Communications		General Tab	(See note 1.)	ОК	
Explicit Message Communications			(See note 2.)	ОК	

Note

1. The same function can be achieved with the setting to enable/disable the master function.

- 2. Explicit message communications is possible whether or not registration has been made on the scan list.
- 3. Before making device I/O allocations to the master, the device must be added to the Network Configuration Window, regardless of whether or not it has been registered as a slave with the master. Refer to *4-1 Adding Devices to the Virtual Network* to create the network configuration.

## 4-6 Parameter Wizard

Device - Parameter -Wizard

- The Parameter Wizard is an interactive function making it possible to allocate I/O with ease.
- This function ensures easy I/O allocations in order of node addresses beginning with block 1. Each block consists of a maximum of 100 words.

When block 1 is full, the remaining I/O will be allocated to block 2.

Note After I/O allocations are made with the Parameter Wizard, node addresses or I/O allocations can be changed by editing the parameters individually.

The following procedure uses the Parameter Wizard with the CS1W-DRM21 as an example. The same method can be used for the CVM1-DRM21-V1 and C200HW-DRM21-V1.

Specify the first address of each block (fixed to 100 words in size), the allocation method (i.e., allocation in blocks of words or minimum required number of words), and the registration or deletion of slaves. Note To make allocations in excess of 100 words per block, use *Parameter - Edit*.

To allocate I/O of slave devices to the master with the Parameter Wizard, use the following procedure.

- **1,2,3...** 1. Select the master.
  - 2. Select Device, Parameter, and Wizard.
  - 3. When settings are made with the Parameter Wizard, the present settings will be cleared and default values will be set. The following confirmation window will be displayed. Click the **Yes** Button.

DeviceNet Configurator 🛛 🕅				
$\underline{A}$	tup will be initial	ized.		
	Yes	No		

4. The following Scan List Wizard-Set Memory Block's Start Word Window will be displayed.

Set the memory area to be used and the start word and click the **Next** Button. I/O will be automatically allocated to block 1. When block 1 is full, I/O will be allocated to block 2. Each block consists of a maximum of 100 words.

Note It is not possible to go to the next step if there is any duplication between blocks or the first word is not within the permissible set range.

Scan list Wizard - Set Memory Block's St	art Word 🔀
	Set start Word of PLC Memory Block storing I/D data. Up to 100 words occupied for each block. A block must not be duplicated to other block. OUT Area Block Start Word Block 1 1/0 Relay 3200 Block 2 1/0 Relay 3400 IN Area Block Start Word Block 1 1/0 Relay 3300 Block 2 1/0 Relay 3500
	< Back. <u>N</u> ext > Cancel

5. The Scan list Wizard-Set How to Allocate I/O Date to PLC Memory Block Window will be displayed.







An allocation example is shown below. 

I/O Exa	ampie:
#00	1 byte
#01	2 bytes
#02	1 bytes
#03	4 bytes
#04	1 byte
#05	1 bytes

## Allocation in Units of Words

		High			Low	
	15		8	7		0
+0CH					#00	
+1CH			#	01		
+2CH					#02	
+3CH			#	03		
+4CH			#	03		
+5CH					#04	
+6CH					#05	

#### Allocation with Allocated Words Minimized

	High			Low	
	15	8	7		0
+0CH	#02			#00	
+1CH		#(	01		
+2CH	#03				
+3CH	#03				
+4CH	#05			#04	

6. Slave Registration/Deletion

The following Scan List Wizard - Set Memory Block's Start Word Window will be displayed.

Specify the slave to be registered with the master and click the **Next** Button.

Scan list Wizard - Register Device	X
	Set Slave Device to be registered in Scan list. Unregister Device List # Prod Out Size In Size
	Register Device List
	#         Product Na         Out Size         In S ▲           ♥ #01         C200HW-D         0 Byte         0 B!           ♥ #02         DRT1-COM         0 Byte         4 B           ♥ #03         DRT1-AD04         0 Byte         8 B           ♥ #04         DRT1-AD04H         0 Byte         8 B
	< <u>B</u> ack <u>N</u> ext > Cancel

The devices on the network are shown in the Registered Device List.

To cancel the registration of a device, click the following icon: \_\_\_\_\_\_ The user cannot go to the next step if no devices are registered.

7. The following Scan List Wizard-Allocated Result Window will be displayed after remote I/O allocations have been made as specified.

Click the **Finish** Button if the displayed details are OK. The Parameter Wizard will finish. The user can click the **Back** Button to return to the previous window.

The displayed details will be set as device parameters.

S	5can list Wizard - Allocated result 🛛 🛛 🛛 🛛 🛛						
	Specified slave device allocated as follows. If OK, press "Finish" Button.						
	#	Product Name	Out Size	Out Ch	In Size	In Ch	M
	<i></i> #01	C200HW-DRT21	2 Byte	050:Bit00	2 Byte	350:Bit00	
	🧼 #02	DRT1-COM	0 Byte		4 Byte	351:Bit00	
	🧼 #03	DRT1-AD04	0 Byte		8 Byte	353:Bit00	
	🧼 #04	DRT1-AD04H	0 Byte		8 Byte	357:Bit00	
	105 🏈	DRT1-HD16C	0 Byte		2 Byte	361:Bit00	
	106 🏈	DRT1-HD16S	0 Byte		2 Byte	362:Bit00	
				< <u>B</u> ack	Finish	Ca	incel

8. When the Configurator is online, the following window will be displayed.

DeviceNel	Configurator 🛛 🕅	1	
٩	Enable device to set the parameter. OK?		
C	Yes <u>N</u> o		

Click the **Yes** Button and edit the master parameters. Remote I/O communications according to the new settings will start.

**Note** Device parameters set with the Parameter Wizard can be edited by individually if necessary.

## 4-7 Master Parameter Editing Details (Tab Descriptions)

Device - Parameter -<br/>EditThe user can individually edit allocations for remote I/O communications and<br/>make advanced settings, such as the communication cycle time and connec-<br/>tion settings. Possible setting items vary with the master model.

The following description provides information on editing individual parameters. Refer to 4-8 Manual I/O Allocations on page 86 and 4-9 Advanced Settings (Connection, Communication Cycle Time, Slave Function Settings, Etc.) on page 93 for further information on editing parameters manually.

To edit the parameters, use the following procedure.

- *1,2,3...* 1. Select the device for which parameters are to be edited.
  - 2. Select Device, Parameter, and Edit.
  - 3. The Edit Device Parameters Window will be displayed.

Note a) If the I/O size of the device displayed in the Network Configuration Window does not coincide with the I/O data size of the device registered with the scan list, the following warning dialog will be displayed along with the Edit Device Parameters Window. In this case, the I/O size in the scan list will take priority.

DeviceNet Configurator 🛛 🛛 🗙			
	I/O data size mismatch detected in the registered device. I/O size in the Scan list will be used.		
	OK		

If an EDS file has not been installed in the slave, obtain the EDS file and install it. Also, set the correct I/O size in all slaves that require changes in the I/O size.

 b) If the slave device registered with the scan list has already been registered with another master, the following warning will be displayed along with the Edit Device Parameters Window.
 Revise the slaves registered in the scan list.

DeviceNet Configurator 🛛 🛛 🗙			
$\underline{\mathbf{A}}$	Found a device in the Scan list which is already registered to other device.		

## 4-7-1 Edit Device Parameters on CS1W-DRM21/CJ1W-DRM21

**Note** When the CS1W-DRM21/CJ1W-DRM21 DeviceNet Unit is used, select the device and then select *Device*, and *Property* and turn ON *Enable Master Function* in the Properties Dialog Box.

Edit Device Parameters х Communication Cycle Time **Slave Function** Message Timer General I/O Allocation(OUT) 1/0 Allocation(IN) Unregister Device List # Product Name Out Size In Size 2 Byte 🧼 #05 🛛 DRT1-ID16 0 Byte 🧼 #06 DRT1-ID16X 0 Byte 2 Byte ٠ Auto allocation as is registered. Register Device List # Product Name Out Size Out Ch In Size In Ch C I #01 DRT1-0D16 2 Byte 3200-Bit 0 Byte 🥏 #02 🛛 DRT1-MD1... 1 Byte 3201:Bit... 1 Byte 3300:Bit... 3301:Bit... 0 Byte 2 Byte 0 Byte 1 Byte 3302:Bit... Advanced Setup. Register/Unregisterd ΟК Cancel

The following window will be displayed when the CS1W-DRM21/CJ1W-DRM21 is selected.

#### The Edit Device Parameters Window consists of the following six tabs.

Tab	Description
General	Registers the device with the scan list and makes I/O allocations automatically.
I/O Allocation (OUT)	Allocates the output data and sets the output memory block for the CPU Unit with the advanced setup function.
I/O Allocation (IN)	Allocates the input data and sets the input memory block in the CPU Unit with the advanced setup function.
Communication Cycle Time	Sets the communication cycle time.
Slave Function	Makes necessary settings that enable the slave function.
Message Timer	Makes monitor timer settings for message communications (both explicit and FINS message communications).

#### General Tab of CS1W-DRM21/CJ1W-DRM21

Edit Device Parameters				
Communication Cycle Time Message Timer Slave Function General I/O Allocation(OUT) I/O Allocation(IN)				
Unregister Device List				
# Product Name Out Size In Size				
Image: Wight	Unregistered Device List			
	Device Register button			
	Device Unregister button			
Register Device List	Automatic allocation as registered			
the second	An asterisk (*) appears when the slave			
#01 DBT1-0D16 2 Byte 3200 Bit 0 Byte	is registered to allow explicit message only.			
🖉 #02 DRT1-MD1 1 Byte 3201:Bit 1 Byte 3300:Bit				
🔷 #03 DRT1-ID16-1 0 Byte 2 Byte 3301:Bit	Registered Device List			
n 🛷 #04 DRT1-ID08 0 Byte 1 Byte 3302:Bit;				
Advanced Setup Register/Unregisterd	Register/Unregistered Button			
Liplead Download Compare	Advanced Setup Button			
OK Cancel				
Compare parameters Usable only when online.				
Download parameters Usable only when online.				
Upload parameters Usable only when online.				

Item	Description
Unregistered Device List	Displays unregistered devices. Unregistered devices refer to slaves that are displayed in the Network Configuration Window but have not been registered with a master.
Registered Device List	Displays slaves that are presently registered with a master.
Device Register and Unregister Buttons	By pressing this button, the selected device will move from the Unregistered Device List to the Registered Device List.
	By pressing this button, the device will move from the Registered Device List to the Unregistered Device List.
Auto allocation as is registered	Check this box when registering the slaves with the master in the Edit Device Parameters Window so that the slaves will be allocated to an unused area in blocks of words in order of registration.
Register/ Unregistered Button	Click this button to unregister the I/O allocation of the selected slaves or to make the I/O reallocation of the slaves in the order of earlier addresses.
Advanced Setup Button	Click this button to set the connections or display the device information.
Upload	Uploads the parameters of the network device online.
Download	Downloads the parameters to the network device on line.
Compare	Compares the parameters of the selected network device with the parameters stored in the Configurator.

# 4-7-2 Editing Device Parameters on CVM1-DRM21-V1 and C200HW-DRM21-V1

The following window will be displayed when the CVM1-DRM21-V1 or C200HW-DRM21-V1 DeviceNet Master Unit is selected.

Edit Device P	dit Device Parameters 🛛 🛛 🗙					
General 1/(	General 1/0 Allocation(0UT) 1/0 Allocation(N) 1 Communication Cucle Time					
United at the						
Unregister				:		
#	Product Name		[ [	Jut Size	In Size	
🥏 #04	DRT1-AD04H		(	) Byte	8 Byte	
<i>t</i> #05	DRT1-ID08		(	) Byte	1 Byte	
<i>🛷</i> #06	DRT1-ID16		(	) Byte	2 Byte	
	-			to allocatio	n as is registered	
- Begister D	evice List		I <b>v</b> Au	to allocatio	n as is registered.	
+	Dreduct Marca	Out Size	L Dua Ch	Lin Cine	Luch [	
# #	CO1V/ DDM01	O Dut 5128		0.0.4		
₩00 ₩ #01	LSTW-DHMZI	0 Byte		U Byte	250.0300	
#UI		0 Byte 2 Bute	050.0300	4 Byte 2 Bute	350:BR00	
₩02 ₩ #02	C200HW-DRT2T	2 Byte 0 Bute	000:81000	2 Byte	352:BRUU	
<i>₩</i> #03	DHITAD04	орую		o byte	303.BILUU	
<u>E</u> xpli	cit Message Commu	nication Only	,	Allocat	e/Unallocated	
🔽 Start Re	Start Remote I/O Communication at Start-Up					
	_     _     _   _     _   _     _     _     _     _     _     _       _					
Upload Download Compare						
			Γ	OK	Cancel	
						_

The Edit Device Parameters Window consists of the following four tabs.

Tab	Description
General	Registers the device with the scan list and makes I/O allocations automatically.
I/O Allocation (OUT)	Allocates the output data and sets the output memory block the CPU Unit with the advanced setup function.
I/O Allocation (IN)	Allocates the input data and sets the input memory block in the CPU Unit with the advanced setup function.
Communication Cycle Time	Sets the communication cycle time.

#### General Tab of CVM1-DRM21-V1 or C200HW-DRM21-V1

Edit Device Parameters	
General 1/0 Allocation(0117) 1/0 Allocation(10) [Communication Cycle Time]	4
W HOS DRT1 ID00 0 Dute 1 Dute	
105 DHTI-D06 0 Byte 1 Byte 5	Unregistered Device List
	Dovice Register butten
	Device Register button
	Device Unregister button
Auto allocation as is registered.	Automatic allocation as registered
Register Device List	An astorick (*) appears when the slave
# Product Name Out Size Out Ch In Size In Ch <u>M</u>	in registered to allow explicit message only
#00 CS1W-DRM21 0 Byte 0 Byte *	is registered to allow explicit message only.
#UI DHIT-LUM U Byte 4 Byte 350:BitUU	
	Registered Device List
·	
Explicit Message Communication Only Allocate/Unallocated	Allocate/Unallocated Button
Start Remote 1/0 Communication at Start Lin	
	Explicit Message Communication Only Button
Lipland Rowpland Compare	
	Start Remote I/O Communication at Start-up
UK Cancel	
Compare parameters Usable	only when online.
Download parameters Usable only when o	online.
Upload parameters Usable only when online	

Item	Description					
Unregistered Device List	Displays unregistered devices. Unregistered devices refer to slaves that are displayed in the Network Configuration Window but have not been registered with the master.					
Registered Device List	Displays slaves that are presently registered with the master.					
Device Register and Unregister Buttons	By pressing this button, the selected device will move from the Unregistered Device List to the Registered Device List.					
	By pressing this button, the device will move from the Registered Device List to the Unregistered Device List.					
Auto allocation as is registered	Check this box when registering the slaves with the master in the Edit Device Parameters Window so that the slaves will be allocated to an unoccupied area in blocks of words in order of registration.					
Allocate/Unallocated Button	Click this button to unregister the I/O allocations or to make the I/O reallocations.					
Explicit Message Communication Only Button	Check this button to enable explicit message communications (transmission only) without remote I/O communications.					
Start Remote I/O Communication at Start-Up	Check this box to enable remote I/O communications automatically when the master starts up.					
Upload	Uploads the parameters of the network device online.					
Download	Downloads the parameters to the network device on line.					
Compare	Compares the parameters of the selected network device with the parameters stored in the					
	Configurator.					

## 4-7-3 Canceling Slave Registration with the Master

Device - Parameter - Edit - General Tab - Register/Unregistered (Allocate/Unallocated) Button

**Note** This procedure is the same for the CS1W-DRM21, CVM1-DRM21-V1, and C200HW-DRM21-V1.

The General Tab is used to register slaves with the master or unregister slaves.

To register or unregister a slave, use the following procedure.

- *1,2,3...* 1. Select the slave to be registered or unregistered.
  - 2. Click the Register or Unregister Button.

A unregistered slave will be displayed in the Unregistered Device List.

A registered slave will be displayed in the Registered Device List.

## 4-7-4 Automatic Allocation with Registration

## Device - Parameter - Edit - General Tab - Auto allocation as is registered - Register/Unregistered (Allocate/Unallocated) Button

- **Note** This procedure is the same for the CS1W-DRM21, CVM1-DRM21-V1, and C200HW-DRM21-V1.
  - If the Auto allocation as is registered Box is checked when the slaves are registered with the master in the Edit Device Parameters Window, the I/O allocation of the slaves will be made in units of words automatically in order of registration. This checkbox will be enabled in the Edit Device Parameters Window only. The automatic allocation of the slaves will be made in units of words automatically in order of registration beginning with the unoccupied area of block 1 of the corresponding memory block.
  - It is possible to unregister the I/O allocation of the selected slaves or to make the I/O reallocation of the slaves in the order of earlier addresses anytime by clicking on Register/Unregistered Button. The advanced setup function (explained later) will not be available is automatic allocation is specified.

## 4-8 Manual I/O Allocations

Device - Parameter - Edit - I/O Allocation Tab

Manual I/O allocation is possible with no restrictions on node address order or block size 1 or 2.

**Note** This procedure is the same for the CS1W-DRM21, CVM1-DRM21-V1, and C200HW-DRM21-V1. In the following setting example, the CS1W-DRM21 is used.

## 4-8-1 I/O Allocation Tab Page

Make the following settings on the I/O Allocation Tab Page.

**1,2,3...** 1. I/O allocation of each OUT/IN memory block (1/2) to the I/O memory area of the CPU Unit



## 2. Click the **I/O Allocation (OUT)** or **I/O Allocation (IN)** Tab. The following window will be displayed.

	ltem	Description
Registered Device List		Only devices with valid output or input data among all devices registered with the General Tab will be displayed.
Auto Bu	utton	Allocates in units of words the slaves selected from the Registered Device List to an unoccupied area in the order of earlier addresses.
Delete Button Unregisters the I/O allocations of the selected slaves in the Registered Delete Button		Unregisters the I/O allocations of the selected slaves in the Registered Device List.
Edit Button Allows manual allocation throu		Allows manual allocation through the Edit Window.
Informa	ation Button	Displays slave information items (i.e., the allocation area and I/O comment data).
Memor	y Blocks	Displays the allocation status of each slave (product name) of block 1 and block 2.
	Ch	Bit address where allocation starts. The first bit address will be displayed after the word address.
	Product Name	Block 1 and block 2
Setup E	Button	Sets the first address and first address size (i.e., the number of words) of block 1 or block 2.

Allocation Status of Blocks 1 and 2

The block allocation status list displays the words of the CPU Unit and the product names of devices allocated in the respective areas of the CPU Unit.

The start bit and the word address for each device are indicated in the *Ch* column.

Example:

3300:Bit00: Starts with bit 00 (LSB) of word 3300. 3300:Bit08: Starts with bit 08 (MSB) of word 3300.



No words in the CPU Unit will be displayed for any memory block that is not in use.

## 4-8-2 Changing the First Address of Output/Input Block

### Device - Parameter - Edit - I/O Allocation Tab - Setup Button

To change the output/input block allocation areas in the I/O memory of the CPU Unit, use the following procedure.

- *1,2,3...* 1. Click the **Setup** Button of the block to be changed.
  - 2. The following window will be displayed.

E	dit Memory Block 🛛 🔀
	Area : 1/0 Relay
	Start Word : 3300 Range : 0000 - 6143
	Words : 100 Range : 1 - 500 Ch
	OK Cancel

3. Set the area, start word, and the number of words for the block.

The number of words to be displayed in a block by the Configurator is set in the Words field. For the CS1W-DRM21, the maximum number of words that can be allocated to one block is 500. For the C200HW-DRM21-V1 and CVM1-DRM21-V1, the maximum number of words that can be allocated to one block is 100.

The ranges that can be set are shown below.

#### • CS1W-DRM21

PC model	Data area	Range
CS Series	CIO Area	0000 to 6143
	DM Area	DM 0000 to DM 8191
	Work Area	WR 000 to WR 511
	HR Area	HR 000 to HR 511
	EM Area	EM 00000 to EM 32767

Banks 0 to 12 of the EM area can be used.

#### • C200HW-DRM21-V1

PC model	Data area	Range
C200HS	IR/SR Area	000 to 511
	HR Area	HR 00 to HR 99
	LR Area	LR 00 to LR 63
	DM Area	DM 0000 to DM 5999
C200HE-CPU11	IR/SR Area	000 to 511
	HR Area	HR 00 to HR 99
	LR Area	LR 00 to LR 63
	DM Area	DM 0000 to DM 4095
C200HX/C200HG/	IR/SR Area	000 to 511
C200HE(-Z) other than the	HR Area	HR 00 to HR 99
above	LR Area	LR 00 to LR 63
	DM Area	DM 0000 to DM 5999

#### • CVM1-DRM21-V1

PC model	Data area	Range
CV500,CVM1-CPU01	IR/SR Area	0000 to 2555
	CPU Bus Link Area	G 000 to G 511
	DM Area	DM 0000 to DM 8191
CV Series and CVM1	IR/SR Area	0000 to 2555
other than the above	CPU Bus Link Area	G 000 to G 511
	DM Area	DM 00000 to DM 24575

Note a) *Words* refers to the number of words in the block displayed on the Configurator. This value is not downloaded to the master.

- b) If the occupied area of the block is less than 100 words when uploaded, *Words* will be set to 100 and displayed as such.
- 4. Click the OK Button. The memory block will be changed by clicking on OK. If a device has already been allocated, it will be allocated to a new memory block. If an area-over error occurs, however, the allocation of the device will be canceled. Reallocate the device, if necessary.

## 4-8-3 Allocating I/O

#### Device - Parameter - Edit - I/O Allocation Tab

The following three methods are supported for I/O allocation.

- 1. Manual Allocation through the Edit Window
   Select the slave from the Registered Device List, click the Edit Button and use the Edit Window to manually allocate the slave.
  - Drag-and-drop Allocation Drag the slave from the Registered Device List and drop the slave at the desired work position in the memory block.
  - Automatic Allocation Select the slaves from the Registered Device List and click the Auto Button. The slaves will be automatically allocated in units of words in the order of earlier addresses. If the user has made connections settings in the General Tab with the advanced setup function, automatic allocation will not be possible.

Note The I/O data size of devices will be displayed in the *Size* column of the Registered Device List as shown below if connections settings have been made for the devices.



To allocate the I/O data on the left-hand side with the mouse, drag and drop the I/O data with the left mouse button. To allocate the I/O data on the righthand side, drag and drop the I/O data with the right mouse button. If there is only a single connection, use the left mouse button.

#### Manual Allocation through the Edit Window

## I/O Allocation Tab - Edit Button

To make allocations through the Edit Window, use the following procedure.

- *1,2,3...* 1. Select the device to edit the I/O allocations of the device.
  - 2. Click the *Edit* Button.
  - 3. The following I/O Allocation Dialog Box will be displayed.

Specify block 1 or block 2, allocation word, start byte (LSB: Low or MSB: High), and the number of allocated bytes.

Edit I/O Allocate 🛛 🕅	Edit I/O Allocate
Block : Start Word : 3200	Poll Block : Start Word : 3300
Occupied: 2 Byte	Allocated : 5501 © Low C High Occupied : 2 Byte
OK Cancel	COS Block 1 Start Word : 3300
	Allocated : 3301 C Low C High Occupied : 2 Byte
	OK Cancel

Connections are specified in the General tab with the advanced setup function.

Specify the first word to allocate and the number of allocated words. The MSB (high) or LSB (low) position can be set for the first word to allocate. The user must set the byte position to LSB if the number of bytes allocated is 2 or more.

• Example: Device with 1 byte allocated in the LSB.



• Example: Device with 1 byte allocated in the MSB.



4. Click the **OK** Button to start I/O allocation.

#### **Drag and Drop**

## Drag and Drop on I/O Allocation Tab

- 1,2,3... 1. Display the Memory Block List.
  - 2. Select the slave from the Registered Device List.
  - 3. Drag the slave to the first byte of the desired position.

Communicatior General	n Cycle	Time I/O Alloc	Me: ation(Ol	ssage Time UT)	r   9 1/0 A	lave Function Ilocation(IN)
# Produ	ct	Size	Ch		C	Auto
🎾 #01 🛛 CS1W	'-D	2, 2 Byte			×	
🎾 #02 🛛 DRT1	-M	1 Byte	3300	:Bit00		<u>D</u> elete
🎾 #03 DRT1	-ID	2 Byte	3301	BitOO		<b>E</b> 10
🖗 #04 DRT1	-ID08	1 Byte	3302	:Bit00		<u>E</u> dit
						Information
Memory Block 1		_		Memory B	lock 2	
Ch	Prod	st Name 🛛	•	Ch	Pro	duct Name 🔺
IIII 3300:Bit00	#02	BT1		<b>IIII</b> 3500:	Bit00	
IIII 3300:Bit08				<b>IIII</b> 3500;	Bit08	
🗰 3301:Bit00	#03	BT1-I		3501:	Bit00	
🗰 3301:Bit08	#03	RT1-I		3501:	Bit08	
🗰 3302:Bit00	#04	DRT1-I		3502	Bit00	
IIII 3302:Bit08				3502	Bit08	
🗰 3303:Bit00				3503	Bit00	
🗰 3303:Bit08				3503:	Bit08	
🗰 3304:Bit00				3504:	Bit00	
IIII 3304:Bit08			_1	3504:	Bit08	
111 3305-BHOD				3505	Rittin	
<u>S</u> etup				<u>S</u> etup		

#### Memory Block List

In the Memory Block List, the *Ch* column displays each byte address (the word and first bit addresses). The *Product Name* column displays the product name for the slave.

#### **Registered Device List**

The # column of the Registered Device List displays the node address, the *Product Name* column displays the product name of the slave, and the *Size* column displays the number of allocated bytes for the slave. The *Ch* column displays the first byte (the word and first bit addresses) of the slave allocated.

To unregister the allocations of slaves, select the slaves in the Registered Device List, and click the **Delete** Button.

Note To register in units of words the slaves in the order of earlier addresses, select the slaves in the Registered Device List, and click the Auto Button. Automatic Allocation *I/O Allocation Tab - Register/Unregistered (Allocate/Unallocated) Button* Note This procedure is the same for the CS1W-DRM21,

CVM1-DRM21-V1, and C200HW-DRM21-V1.

- To allocate in units of words the selected slaves to an unoccupied area in the order of earlier addresses, click the **Auto** Button.
- To unregister the allocations of the selected slaves, click the **Delete** Button. The advanced setup function (to be explained later) is not possible while automatic allocation is specified.

## 4-8-4 Slave Information

#### I/O Allocation Tab - Information Button

Device information on registered slaves, such as I/O comments, can be checked with the I/O Allocation Tab Page. The I/O comment is set to the I/O data of a slave by selecting *Edit I/O Comment* from the Device Menu.

To display information on the slave, use the following procedure.

- **1,2,3...** 1. Select the device.
  - 2. Click the Information Button.

The following window will be displayed.

#03 C200H\	W-DRT21 I	nformation 🛛 🔀				
Description : MAC ID : Vendor : DeviceType Product Cod Product Nam Status : Poll	C200H #03 OMRO : Commu e: 51 ne: C200H Registe	W-DRT21 N Corporation Inications Adapter W-DRT21 Ired to #02.				
OUT Size :	2 Byte					
Area	Bit	Comment				
Ke 3201	Bit00	OUT Comment1				
Veii 3201	Bit06	OUT Comment2				
Xee 3201	Bit12	OUT Comment3				
IN Size :	IN Size : 2 Byte					
Area	Bit	Comment				
ات 🗺 📾	Bit02	IN Comment1				
KE 3301	Bit05	IN Comment2				
Veii 3301	Bit15	IN Comment3				
,						
Close						

When the registered device is selected in the above window, the slave information will be refreshed to information on the selected device.

# 4-9 Advanced Settings (Connection, Communication Cycle Time, Slave Function Settings, Etc.)

The following description provides information on the connection, device information display and check, communication cycle time, message timer, and slave function settings.

## 4-9-1 Advanced Settings

Select *Device*, *Parameter*, *Edit*, and the General Tab. Select the slave and click the Advanced Setup Button.

Make advanced setup settings (such as device information display/compare and connections settings) for remote I/O communications.

Note These settings are available for the CS1W-DRM21 only.

Device Information Display/Compare Settings

#### Device Information Tab

The following settings make it possible to display or compare device information on slaves.

Use the following procedure.

- *1,2,3...* 1. Select the slave from the Registered Device List.
  - 2. Click the Advanced Setup Button.
    - 3. The following window will be displayed. Click the **Device Information** Tab.

Advanced setting	
Device Information Device Information Vendor: 47 Device Type: 0 Product Code: 100 Check Device Type Product Code: 100 Check Product Code OK Cancel	<ul> <li>The device information on the selected slave will be displayed.</li> <li>If these boxes are checked, the device information will be compared with the corresponding data in the scan list during remote I/O communication. If the information does not coincide with the data, a verify error will result.</li> </ul>

The device information (consisting of vendor, device type, and product code data) on the selected slave is displayed.

By checking the above boxes, the device information will be compared with the corresponding data in the scan list when I/O communication connections are established.
Connections

#### Connection Tab

When the user specifies the connections for remote I/O communications, up to two connections can be set for each slave.

Use the following procedure.

- *1,2,3...* 1. Select the slave in the Registered Device List.
  - 2. Click the Advanced Setup Button.
    - 3. The following window will be displayed. Click the **Connection** Tab.

Advanced setting	X
Device Information Connection	
O Auto Connection	
OUT Size : 2 Byte	IN Size : 2 Byte
User Setup	
Use Poll Connection	
OUT Size: 2 Byte	IN Size : 2 Byte
Con. Path :	Con. Path :
Use Bit-Strobe Connection	
OUT Size : 0 Byte	IN Size : 0 Byte
Con. Path :	Con. Path :
☑ Use COS Connection	
OUT Size : 2 Byte	IN Size : 2 Byte
Con. Path :	Con. Path :
🔲 Use Cyclic Connection	
OUT Size : 0 Byte	IN Size : 0 Byte
Con. Path :	Con. Paih :
COS/Cyclic Heart Beat Timer : 1000	ms
	OK Cancel

Automatic connections are set by default.

To specify the connections, use the following procedures.

1,2,3... 1. Select User Setup.

Possible connections are ready for selection.

2. Select the connections.

Up to two connections can be selected.

Note Both COS and Cyclic cannot be specified at the same time.

- 3. Set the connection path as needed.
- 4. Set the COS/Cyclic heard beat timer.
- 5. Click the **OK** Button.

An asterisk (\*) will be displayed in the C column at the right edge of the Registered Device List. If the connections of a device already allocated with I/O are changed, the previous I/O allocations will be canceled. Make the I/O allocations again.

- **Note** 1. Both COS and Cyclic settings cannot be made at the same time.
  - 2. If *Poll* and *COS* settings or *Poll* and *Cyclic* settings are used in combination, check that the output settings of the both connections are the same.
  - 3. The automatic allocation function is not possible for devices if the user already made connections settings for the devices with the advanced setup.

Unregister and register the devices so that the automatic allocation function will be available for the devices.

### 4-9-2 Communication Cycle Time Settings

#### Device - Parameter - Edit - Communication Cycle Time Tab

Note This procedure is the same for the CS1W-DRM21, CVM1-DRM21-V1, and C200HW-DRM21-V1.

It is possible to set the communication cycle time on the Communication Cycle Time Tab Page. Furthermore, it is possible to check the present communication time calculated from the registered device information.

The following window will be displayed by clicking on the Communication Cycle Time Tab.

	I/O Allocation(OUT) I/O Allocation(IN)
Lommunication Lycle 1	ime Message Timer Slave Function
Communication Cycle 1	Time : 🚺 💌 ms
Setu	up Range: 0:Auto(Default),1 - 500 ms
Default Setup	
Communication Cycle Tir	me (Auto Setting)
Communication Cycle Tii Baud rate 125K Bit/s :	me (Auto Setting)
Communication Cycle Tir Baud rate 125K Bit/s : Baud rate 250K Bit/s :	me (Auto Setting) 2.942 ms 2.000 ms
Communication Cycle Tii Baud rate 125K Bit/s : Baud rate 250K Bit/s : Baud rate 500K Bit/s :	me (Auto Setting) 2.942 ms 2.000 ms 2.000 ms
Communication Cycle Tir Baud rate 125K Bit/s : Baud rate 250K Bit/s : Baud rate 500K Bit/s :	me ( Auto Setting ) 2.942 ms 2.000 ms 2.000 ms

Set the communication cycle time between 1 and 500 ms. To set the communication cycle time automatically, click the **Default Setup** Button or set the communication cycle time to 0 ms.

The communication cycle time set will be automatically calculated and displayed according to the registered device information and the baud rate.

**Note** Communication cycle time refers to the required time of remote I/O communications between the master and a slave. Communication cycle time settings properly made will prevent the fluctuation of the time of remote I/O communications with the slave. Furthermore, by setting the communication cycle time to a larger value, the prolonged processing operation of the slave will not be treated as a communications error. If the actual remote I/O communications time is shorter than the set communication cycle time, the remote I/O communications will keep pace with the communication cycle time. If actual I/O communications take longer than the set communication cycle time, the I/O communications will be continued regardless of the set communication cycle time. Advanced Settings (Connection, Communication Cycle Time, Slave Function Settings, Etc.) Section 4-9

### 4-9-3 Setting Message Timers

Device - Parameter - Edit - Message Timer Tab

Note This procedure is supported by the CS1W-DRM21 only.

General I/O Allo	ocation(OUT) I/O Allocation(IN)
Communication Cycle Time	Message Timer Slave Function
#	Message Timer
#00	2000 ms
#01	2000 ms
#02	2000 ms
#03	2000 ms
#04	2000 ms
#05	2000 ms
#06	2000 ms
#07	2000 ms
#08	2000 ms
#09	2000 ms
#10	2000 ms
#11	2000 ms
#12	2000 ms
#13	2000 ms
#14	2000 ms
#15	2000 ms
#16	2000 ms
#17	2000 ms
#18	2000 ms
🖗 #19	2000 ms
<u>E</u> dit	Copy to <u>A</u> ll Device

A message timer is by default set to 2 s (2000 ms). They can be set 1-ms increments to between 500 and 30,000 ms.

To change a message timer value, use the following procedure.

Double-click the node address (#) or select the node address and click the Edit Button. The following dialog box will be displayed.

Setup Message Timer
New Message Timer : 📴 📩 ms
Setup Range 500 - 30000 ms
OK Cancel

- 2. Set the value and click the **OK** Button.
- Note To set the same value for all the devices, select the node addresses and click the **Copy to All Device** Button.
- **Note** The message timer is used to monitor the time of message communications (explicit message communications time and FINS message communications time). The message timer can be set for each destination device independently. If the response of a destination device is slow, the timer value must be set to a larger value. The response may be slow for multilevel FINS message communications. The next message cannot be sent to the same device until the response is returned. The DeviceNet master monitors the time-out period of the message with this timer. The CPU Unit, however, is responsible for monitoring the response time with the CMND, SEND, and RECV instructions. Therefore, it is meaningless if only the timer value or monitor time is changed

for the CMND, SEND, RECV instructions. Set the response monitor time with the CMND, SEND, and RECV instructions to a value larger than the timer value. Set both of them to larger values but the value of response monitor time must be still longer than the timer value if a time-out error occurs frequently.

### 4-9-4 Slave Function Settings

### Device - Parameter - Edit - Slave Function Tab

Note This procedure is supported for the CS1W-DRM21 only.

The slave function is enabled through settings made in the Slave Function Tab.

**Note** To enable the slave function with the CS1W-DRM21, select the device first, and then select *Device – Property*, and select *Enable slave Function* in the CS1W-DRM21 Properties Dialog Box.

To set the slave function, use the following procedure.

#### *1,2,3...* 1. Click the **Slave Function** Tab.

The following window will be displayed.

dit Device Parameters	
General I/O Allo Communication Cycle Time	cation(OUT) I/O Allocation(IN) Message Timer Slave Function
<ul> <li>Auto Connection</li> </ul>	
_ OUT	IN
Area : 1/0 Relay	Area : 1/0 Relay
Allocated : 3370	Allocated : 3270
Occupied : 2 Byte	e Occupied: 2 Byte
Area : 1/0 Relay Allocated : 0	Area : 1/0 Relay
Allocated : 0	Allocated : 0
Occupied : U Byte	e Occupied : U Byte
	OK Cancel

2. Specify the connections.

Automatic settings are by default set. If the user specifies the connections, click the **User Setup** Button.

- Set the output and input areas used for remote I/O communications.
   Set each area type for *IN* (the slave to the master) and *OUT* (the master to the slave), start words, and allocated sizes.
   If the user's connections settings are used, make all connections settings.
   Up to two connections can be set.
- **Note** 1. Both COS and Cyclic settings cannot be made at the same time.
  - 2. If *Poll* and *COS* settings or *Poll* and *Cyclic* settings are used in combination, check that the *OUT* settings of the both connections are the same.

### 4-9-5 Setting/Canceling Explicit Message Communications

#### Device - Parameter - Edit - General Tab - Explicit Message Communications Box

**Note** This procedure is supported for the CVM1-DRM21-V1 and C200HW-DRM21-V1 only.

Specify whether or not to enable explicit message communications (transmissions only) without remote I/O communications. Check the box to enable explicit message communications only.

**Note** For the CS1W-DRM21 and CJ1W-DRM21, slaves with explicit message communications only do not have to be registered. Explicit message communications is possible regardless of the scan list registration.

### 4-9-6 Starting Remote I/O Communications

### Device - Parameter - Edit - General Tab - Start Remote I/O Communication at Start-Up Box

**Note** This procedure is supported for the CVM1-DRM21-V1 and C200HW-DRM21-V1 only.

Specify whether or not to enable remote I/O communications automatically. Check the box to start remote I/O communications automatically when the master is started.

**Note** For the CS1W-DRM21 and CJ1W-DRM21, remote I/O communications upon start-up can be disabled by disabling the master function. Select **Device**, then select the **Unit Function** Tab of the Property Window, and remove the check mark from Enable Master Function.

# 4-10 Creating and Editing I/O Comments

Device - Edit I/O Comment

The Configurator can be used to add comments to I/O data of slaves.

The I/O comments can be checked while making I/O allocations to the master. Note No I/O comments can be created or edited for a device if the device is not designed to have I/O data.

To edit I/O comments, use the following procedures.

- *1,2,3...* 1. Select the device.
  - 2. Select *Device* and *Edit I/O Comment* (or select the slave, click the right mouse button, and select *Edit I/O Comment*).

E

The following window will be displayed.

Area	Bit	Comment
ee 0000 ک	Bit00	OUT Comment1
ا≣ 0000 ک	Bit01	
ا≣ 0000 ک	Bit02	
0000 📾	Bit03	
0000 📾	Bit04	
COOO 🖃	Bit05	
VEE 0000	Bit06	OUT Comment2
NEE 0000	Bit07	
₩E 0000	Bit08	
€ 0000	Bit09	
₩E 0000	Bit10	
€E 0000	Bit11	
<del>کا (</del> 0000 🖂	Bit12	
VEE 0000	Bit13	
k≘ 0000	Bit14	
₩Ξ[0000	Bit15	OUT Comment3

An I/O comment can be created for each connection supported by the device. If areas have been already allocated to the master, they will be displayed.

- Note a) If an area is not allocated to a master, the display will show 0000, the default area.
  - b) Edited I/O comments can be exported in the CSV format by selecting Export and I/O Comment List. Refer to 6-5 Exporting Data Created in Configurator. Part of I/O comment data exported in the CSV format can be opened in spreadsheet software and used for CX-Programmer variable tables.
- 3. Select the bit where the I/O comment should be set and press the **Enter** Key (or click the bit position).

Data can be input into the comment area as shown below.

8	🖻 Poll OUT 🛛 😹	∋ Poll IN		
	Area	Bit	Comment	
	Xee 0000	Bit00	OUT Comment1	
	ا⊠ 0000	Bit01		
	ا⊠ 0000	Bit02		

4. Input the comment and press the Enter Key.

To cancel the input, press the **ESC** Key.

5. Set all comments and click the **OK** Button.

# 4-11 Displaying Device Properties

**Device - Property** 

This section explains device information on network devices.

Device properties are classified into those common to all devices and those inherent to each device.

To display the device properties, use the following procedure.

- **1,2,3...** 1. Select the device.
  - 2. Select Device and Property.
  - 3. The Property Window will be displayed.

### 4-11-1 Property Window Common to All Devices

The following window will be displayed for device properties.

C200HW-DRM21-	200HW-DRM21-¥1 Property 🛛 🗙				
General 1/0 Info	General 1/0 Information PLC Information				
C2(	00HW-DRM21-V1 Change <u>I</u> con <u>D</u> efault Icon				
Description :	C200HW-DRM21-V1				
MAC ID :	#14				
Vendor :	OMRON Corporation				
Device Type :	Communications Adapter				
Product Code :	1				
Revision :	1.02				
Serial No. :	0000000				

This Window has the name of the vendor of the device and the device type. By clicking the **Change Icon** Button, the icons displayed by the Configurator can be customized.

To change the icon to standard ones, click the **Default Icon** Button.

# 4-11-2 I/O Information Inherent to Each Slave Device

The following I/O Information Tab Page will be displayed.

20 0	General 1/O Information					
	OUT[Poll] allocated to #63. ( 3200 : Bit00 ) IN[Poll] allocated to #63. ( 3300 : Bit00 )					
	Connection	Out/In	Size	Help		
	9 Poll	Out	2 Bytes			
		In	2 Bytes			
	Bit-Strobe	Out	0 Bytes			
		In	0 Bytes			
	COS	Out	0 Bytes			
		In	0 Bytes			
	Cyclic	Out	0 Bytes			
		In	0 Bytes			
[	<u>E</u> dit					
					Clo	se

Supported I/O connection information defined by the EDS file will be displayed as slave I/O information properties. If the I/O allocations are made to another master, the I/O allocation information will be displayed together.

The following icon will be displayed next to I/O connection information items registered as default I/O connections in the EDS file.

Connection	Out/In	Size	Help
9 Poll	Out	2 Bytes	
	In	2 Bytes	
Bit-Strobe	Out	0 Bytes	
	In	0 Bytes	
COS	Out	0 Bytes	
	In	0 Bytes	
Cyclic	Out	0 Bytes	
	In	0 Bytes	

If there is no EDS file or the slave's I/O size is variable, the I/O size can be changed using the following procedure.

Edit I/O Size	2			×
Default     Poll	C Bit-Strobe C	COS	O Cyclic	
Poll OUT Size :	2 Byte	IN Size :	2	Byte
Bit-Strobe -	0 Byte	IN Size :	0	Byte
COS OUT Size :	0 Byte	IN Size :	0	Byte
Cyclic OUT Size :	0 Byte	IN Size :	0	Byte
	ОК	Cancel		

*1,2,3...* 1. Click the **Edit** Button. The Edit I/O Size Dialog Box will be displayed.

- Select the default connection typed.
   The size of each connection can be set by selecting *Poll*, *Bit-Strobe*, *COS*,
- or *Cyclic*. 3. Click the **OK** Button.

When connecting the Configurator to a network that is already operating, for example using OMRON Master Units with fixed allocation, the slave I/O size can be obtained from the master scan list for use in setting.

However, this is possible only with OMRON masters. Also, the slave I/O size cannot be obtained from the slave function of the CSW-DRM21.

Use the following procedure to obtain the I/O size from a master scan list.

- 1,2,3... 1. Select the desired slave, then select Device and Property.
  - 2. Select the I/O Information Tab.
  - 3. Click the Get I/O Size from the Scanlist Button.

Parameter Auto Update<br/>when I/O Size ChangedThe function to automatically update the I/O size registered on the master<br/>scan list when the I/O size is changed can be switched on or off.

To automatically update the I/O size registered on the master scan list, select *Option* and *Parameter Auto Update When I/O Size Changed*.

Obtaining I/O Size Information from the Master Scan List When this function is enabled, the v mark will be displayed in the menu. The default is set to not automatically update.



**Note** If the function to automatically update the I/O size is enabled and there is no EDS file, some operations will cause the I/O size in the master scan list to be updated to zero. For this reason, the function to automatically update the I/O size should normally be disabled.

### 4-11-3 Information for Master

OMRON's CVM1-DRM21-V1, C200HW-DRM21-V1, and CS1W-DRM21 have an Master I/O Information Tab Page and PLC Information Tab Page.

### CS1W-DRM21 DeviceNet Unit

Master I/O Information

C51W-DRM21 Property		×
General Master I/O Information	Uni 1/0 Information	t Fuction PLC Information
1/0 Communication Regist	er Counts : 2	
OUT Allocated Words :	1 Words (Include	es free area)
IN Allocated Words :	3 Words (Include	es free area)
OUT Bits :	16 Bits	
IN Bits :	48 Bits	
Communication Cycle Tin	ne	
Baud rate 125K Bit/s :	2.000 ms	
Baud rate 250K Bit/s :	2.000 ms	
Baud rate 500K Bit/s :	2.000 ms	
		Close

ltem	Description
I/O Communication Register Counts	The number of devices registered as remote I/O communications devices in the scan list.
OUT/IN Allocated Words	The number of allocated words including the unused words from the first memory block set in the scan list.
OUT/IN Bits	The number of actual I/O bits for remote I/O communications devices.
Communication Cycle Time	The communication cycle time based on the device information in the scan list. If the user has set the communication cycle time, the user's set value will be displayed.

The following master I/O information will be displayed.

### **PLC Information**

C51W-DRM21 Property		X
General	Ur	nit Fuction
Master I/O Information I/O	D Information	PLC Information
PLC Model : CS1H-CPUxx/	CS1G-CPUxx	
Unit No. : #0		
PLC Mode		
C Program O Monitor	O Run	Change
		Close

The name of the PLC model in use and the unit number of the master will be displayed as PLC Information. The operating mode of the PLC will be displayed online. To change the PLC mode, click the **Change** Button.

### CVM1-DRM21-V1 and C200HW-DRM21-V1

### Master I/O Information

200HW-DRM21-¥1 Prope	rty 🗵
General 1/0 Information	PLC Information
1/0 Communication Registe	er Counts : 2
Message Communication R	Register Counts : 0
OUT Allocated Words :	1 Words (Includes free area)
IN Allocated Words :	3 Words (Includes free area)
OUT Bits :	16 Bits
IN Bits :	48 Bits
Communication Cycle Tim	e
Baud rate 125K Bit/s :	2.000 ms
Baud rate 250K Bit/s :	2.000 ms
Baud rate 500K Bit/s :	2.000 ms
	Close

The following master I/O information will be displayed.

Item	Description
I/O Communication Register Counts	The number of devices registered as remote I/O communications devices in the scan list.
Message Communication Register Counts	The number of devices registered as message communications devices with the scan list.
OUT/IN Allocated Words	The number of allocated words including the unused words from the first memory block set in the scan list.
OUT/IN Bits	The number of actual I/O bits for remote I/O communications devices.
Communication Cycle Time	The communication cycle time based on the device information in the scan list. If the user has set the communication cycle time, the user's set value will be displayed.

### **PLC Information**

C200HW-DRM21-¥1 Property	×
General 1/0 Information PLC Information	
PLC Model : C200HX/HG/HE(-Z) Series	
Dhange Unit	
Close	

The name of the PLC model in use and the unit number of the master as PLC Information will be displayed. To change the PLC model, click the **Change Unit** Button. If the network configuration is read from the actual network, the PLC model cannot be changed.

# **SECTION 5 Online Operation**

This section explains the online operation of the DeviceNet Configurator, such as data downloading to or uploading from devices and device monitoring.

5-1	Switchi	ing between Online and Offline	108
	5-1-1	Specifying the Interface	108
	5-1-2	Going Online	108
	5-1-3	Going Offline	110
5-2	Downlo	bading the Network Configuration/Device Parameters to Devices	110
	5-2-1	Downloading the Network Configuration	111
	5-2-2	Downloading Device Parameters	112
	5-2-3	Resetting the Device	115
5-3	Upload	ing and Verifying Device Parameters	115
	5-3-1	Uploading the Network Configuration	115
	5-3-2	Uploading Device Parameters	116
	5-3-3	Verifying the Network Configuration	117
	5-3-4	Verifying Device Parameters	117
5-4	Monito	ring Devices	119
	5-4-1	Setting Monitor Refresh Timer	119
	5-4-2	Monitoring Devices	119

# 5-1 Switching between Online and Offline

This section explains how to connect the Configurator to the DeviceNet network.

The Configurator can be connected to the actual DeviceNet network, thus making it possible to exchange data with the devices on the network and monitor the operation of the devices online.

**Note** Online connections must be implemented together with the selection of the interface. Refer to 3-3-2 *Designating the Interface*.

### 5-1-1 Specifying the Interface

**Option - Select Interface** The following procedure is not required if the interface has already been specified. If the interface has not been specified, use the following procedure to select the interface.

The interface specified in the following procedure will be saved. The same interface will be selected the next time the Configurator is started.

### 1, 2, 3... 1. Select Option and Select Interface.

There will be a check mark on the current interface selection.

- 2. Select the interface from the menu.
  - Check DeviceNet IF Card if the dedicated Board or Card is used.
  - Check SYSMAC CS/CJ IF Port if a serial connection is used.
  - Check SYSMAC CS/CJ Ethernet Unit IF if an Ethernet connection is used.
- **Note** The interface cannot be changed if the Configurator is online. Select **Network** and **Disconnect** to change the interface offline.

### 5-1-2 Going Online

### **Network - Connect**

### 1, 2, 3... 1. Select Network and Connect.

The window corresponding to the interface designated for connection will be displayed.

### **DeviceNet IF Card**

2. If *DeviceNet IF Card* is specified, the following DeviceNet window will be displayed.

Select Interface	×
Select Interface Card.	
OMRON 3G8F5/E2-DRM21	<b>•</b>
	Cancel

3. Select the DeviceNet Card and click the **OK** Button.

The following Setup Interface Dialog Box will be displayed.

Setup Interface 🛛 🗙
Card Name
Driver250
MAC ID
Baudrate
O 125K Bit/s O 250K Bit/s ⊙ 500K Bit/s
OK Cancel

4. Click the **OK** Button.

A confirmation window will be displayed. Click the **OK** Button.

The DeviceNet network will be connected.

When the network is connected normally online, the color of the status indicator on the Status Bar will change to blue and *Online* will be displayed.

#### SYSMAC CS/CJ IF Port

(Continued from step 1 above.)

2. If SYSMAC CS/CJ IF Port is specified as the interface, the Setup Interface Dialog Box will be displayed.

Setup Interface	×
Interface :	Toolbus
Network Address :	0 -
Node Address :	0 +
Unit No. :	0 +
COM Port :	COM1 💌
Baud Rate :	115200 Bit/s 💌
Data Length :	8Bits 💌
Parity :	No
Stop Bit :	1Bit 💌
ОК	Cancel

Refer to Selecting SYSMAC CS/CJ IF Port in 3-3-2 Designating the Interface for each setting item.

3. Click the **OK** Button.

A confirmation window will be displayed. Click the **OK** Button.

The DeviceNet network will be connected.

When the network is connected normally online, the color of the status indicator on the Status Bar will change to blue and *Online* will be displayed. Refer to the SYSMAC CS/CJ-series Operation Manual for each setting item.

SYSMAC CS/CJ Ethernet Unit IF Port

- (Continued from step 1 above.)
- 2. If SYSMAC CS/CJ Ethernet Unit IF Port is specified as the interface, the Setup Interface Dialog Box will be displayed.

Setup Interface
Host(PC) Information
Host Name : PC1
IP Address : 10.3.74.142
Network Address : 000
Node Address : 142
- Remote Information
Registration Name :
DeviceNet Unit
Network Address : 000
Node Addres: 000
Unit No. : 01
Ethernet Unit
Port No. : 9600
IP Address : 10.3.74.37
<u>S</u> etup
0K Cancel

Refer to Selecting SYSMAC CS/CJ Ethernet Unit IF in 3-3-2 Designating the Interface for each setting item.

3. Click the **OK** Button.

A confirmation window will be displayed. Click the **OK** Button.

The DeviceNet network will be connected.

When the network is connected normally online, the color of the status indicator on the Status Bar will change to blue and *Online* will be displayed.

### 5-1-3 Going Offline

**Network - Disconnect** 

Use the following procedure to set the Configurator to offline.

#### Select Network and Disconnect.

The DeviceNet network will be disconnected.

# 5-2 Downloading the Network Configuration/Device Parameters to Devices

This section explains how to write the master and slave parameters that were created on the Configurator to the actual network devices. This process is called "downloading."

The following two methods are used to write parameters to network devices.

- Downloading the network configuration
- · Downloading parameters for specific devices

### 5-2-1 Downloading the Network Configuration

- **Network Download** The network configuration download function makes it possible to write and reset the device parameters in the order of node addresses, and enable the new settings.
  - **Note** When downloading the network configuration, each of the devices is reset. If the Master Unit is reset first, it may cause errors in writing parameters to the subsequent slaves. For that reason, this method (downloading the network configuration) should be used only when the Master Unit has been given the highest address.

To download the network configuration, use the following procedure.

- *1, 2, 3...* 1. Place the Configurator online.
  - 2. Select *Network* and *Download*.

The following conformation window will be displayed.

DeviceNe	et Configurator 🛛 🕅
	Enable all devices to set the parameter. OK?
	<u>Y</u> es <u>N</u> o

3. Click the **Yes** Button.

The following progress window will be displayed and the downloading of the network configuration will be started.

Downloading Device Parameter ( #63 )
Abort

- Note 1. The downloading can be canceled by clicking the Abort Button.
  - 2. If an error occurs while the network configuration is downloaded, the following confirmation window will be displayed.

Specified d	levice can not be accessed, or wrong device type. (
The proces	ss will be continuing after 15 minutes.
<u> </u>	Continue

Click the **Continue** Button to continue the process. Click the **Abort** Button to cancel the process.

The downloading of the network configuration will be automatically continued if the user does not click the **Abort** Button for 15 s after the above window is displayed. The following window with information on device errors or missing devices will be displayed after the network configuration has been downloaded.

E	rror device list		X
	The following devices skipp error occurred.	ed the process because of	
	Error Device	Description	
	#50 C200HW-DRT21	Specified device can n	
	#63 C200HW-DRM2	Specified device can n	
	Close		

The new settings will be valid when the network configuration is downloaded properly.

- Because the devices are reset in order, communications errors will temporarily occur in the master and slaves. For this reason, do not download the network configuration while the master-side PLC (CPU Unit) is operating.
  - Parameters cannot be downloaded to the master unless the CPU Unit is in Program Mode. If the message *Device state conflict. Going to change PLC Mode. OK*? is displayed, check the operation mode of the CPU Unit.

## 5-2-2 Downloading Device Parameters

Device parameters are downloaded through the Network Configuration Window or Edit Device Parameters Window.

- Downloaded device parameters will be valid only after the devices are reset unless they are the OMRON CVM1-DRM21-V1, C200HW-DRM21-V1, or CS1W-DRM21.
  - 2. The parameters in the Configurator will differ from the actual parameters if the **No** Button is clicked to quit the Edit Device Parameters Window after the parameters have been downloaded.

### Device - Parameter - Edit - Download

To download the parameters through the Edit Device Parameters Window, use the following procedure.

- *1, 2, 3...* 1. Place the Configurator online.
  - 2. Select the device.
  - 3. Select Device, Parameter, and Edit.
  - 4. Click the *Download* Button.

<u>U</u> pload	<u>D</u> ownload	<u>C</u> ompare

network configuration wh 2. Parameters cannot be do Program Mode. If the mes *Mode. OK*? is displayed

through the Edit Device Parameters Window

**Downloading Parameters** 

5. The following confirmation window will be displayed.



6. Click the Yes Button.

The progress window will be displayed while the parameters are being downloaded.

**Note** When downloading parameters with the Edit Device Parameters Window, the following dialog box will appear if the operation mode of the CPU Unit is set to anything other than Program Mode.

DeviceN	et Configurator	×
⚠	Device state conflict. Going to change PLC Mode. OK?	
	Yes <u>N</u> o	

Click the **Yes** Button, to change the operation mode of the CPU Unit and start downloading.

DeviceN	et Configurator 🛛 🛛 🔀
⚠	Going to change Original PLC Mode. OK?
<b>*</b>	Yes <u>N</u> o

After downloading, this dialog box makes it possible to return to the original operation mode.

Downloading through the Network Configuration Window

#### Device - Parameter - Download

To download the parameters through the Network Configuration Window, use the following procedure.

1, 2, 3... 1. F

- 1. Place the Configurator online.
  - 2. Select the device or devices.
  - 3. Select *Device*, *Parameter*, and *Download*.

The following confirmation window will be displayed.

DeviceNe	t Configurator 🛛 🕅
	Downloading parameters to selected devices will start. OK?
	<u>Y</u> es <u>N</u> o

4. Click the Yes Button.

The following progress window will be displayed and the downloading of the parameters will start.

Download	ling Device Parameter ( #00 )
	Abort

To cancel the downloading, click the **Abort** Button.

The following confirmation window will be displayed if there is an error while the parameters are downloaded.

Specified device can not be a #63 1 The process will be continuing	accessed, or wrong device type. ( g after 15 seconds.
Continue	Abort

Click the **Continue** Button to continue the process. Click the **Abort** Button to cancel the process.

The downloading of the network configuration will be automatically continued if the user does not click the **Abort** Button for 15 s after the above window is displayed.

The following window with information on device errors or missing devices will be displayed after the parameters have been downloaded.

Error device list 🛛 🗶			
	The following devices skipp error occurred.	ed the process because of	
	Error Device	Description	
	#00 CS1W-DRM21	Specified device can n	
Close			

**Note** Parameters cannot be downloaded to the master through the Network Configuration Window unless the CPU Unit is in Program Mode. If the message *Device state conflict. Going to change PLC Mode. OK?* is displayed, check the operation mode of the CPU Unit.

### 5-2-3 Resetting the Device

Device - Reset

To enable the new settings, use the following procedure to reset the network device.

- *1, 2, 3...* 1. Place the Configurator online.
  - 2. Select the device in the Network Configuration Window.
  - 3. Select *Device* and *Reset*.

The following confirmation window will be displayed.

DeviceNe	Configurator 🛛 🕅
$\underline{\mathbb{A}}$	Selected devices will be reset. OK?
	Yes <u>N</u> o

4. Click the Yes Button.

The progress window will be displayed and the devices will be reset.

The new settings will be valid after the devices have been reset.

**Note** When the devices are reset, communications errors will temporarily occur. For this reason, do not reset the devices while the master-side PLC (CPU Unit) is operating

# 5-3 Uploading and Verifying Device Parameters

Uploading device parameters refers to writing the parameters from the actual network devices to the virtual network in the Configurator. Verifying device parameters refers to comparing the parameters in the actual network devices with those in the Configurator.

This section explains how to do both of these operations.

**Note** When the network configuration is uploaded, the parameters for each device will all be read.

### 5-3-1 Uploading the Network Configuration

Network - Upload

- To read the actual network configuration, use the following procedure.
- *1, 2, 3...* 1. Place the Configurator online.
  - 2. Select Network and Upload.

The uploading of the network configuration will start and the following progress window will be displayed.

Uploading Device Information ( #04 )
Abort

The uploading can be canceled by clicking the **Abort** Button.

3. Uploaded devices will be displayed in sequence in the Network Configuration Window.

#### **Uploading Device Parameters** 5-3-2

The following two methods can be used to read parameters from network devices.

- · Reading parameters from the selected network devices
- Reading parameters from the Edit Device Parameters Window

Using Network **Configuration Window** 

### Device - Parameter - Upload

To upload the parameters through the Network Configuration Window, use the following procedure.

- 1. Place the Configurator online. 1, 2, 3...
  - 2. Select the device.
    - 3. Select Device, Parameter, and Upload.

DeviceNet Configurator

OK?

The following confirmation window will be displayed.

Yes

4.	Click the <b>Yes</b> Button.
	The progress window will be displayed and the parameters will be upload
	ed.

Uploading parameters from selected devices will start.

No

Using Parameter Edit Window

### Device - Parameter - Edit - Upload Button

To upload the parameters through the Edit Device Parameters Window, use the following procedure.

- 1, 2, 3... 1. Place the Configurator online.
  - 2. Select the device.
  - 3. Select Device, Parameter, and Edit.
  - 4. Click the Upload Button.

	opioda	Compare

Upload Download Comp

The following confirmation window will be displayed.

DeviceNe	t Configurator 🛛 🕅
	Uploading parameters from device will start. OK?
	Yes No

5. Click the Yes Button.

The progress window will be displayed and the parameters will be uploaded.

### Section 5-3

**Note** The uploaded parameters will be discarded if the **No** Button is clicked to quit the Edit Device Parameters Window after the parameters have been uploaded.

# 5-3-3 Verifying the Network Configuration

Use the following procedure to compare the actual network configuration with the network configuration created on the Configurator.

- 1, 2, 3... 1. Place the Configurator online
  - 2. Select **Network** and **Compare**. The progress window will be displayed and the comparison will begin.
  - 3. If there are any comparison errors, the following window with the details of the errors will be displayed.

Comparison Result			X
Description	Local	Network	
😡 Wrong device type. ( #22 )	C200HW-DR	Not exist.	
😡 Wrong device type. ( #29 )	DRT1-ID16X	DRT1-ID16	
😡 Wrong device type. ( #30 )	DRT1-0D16	DRT1-ID16	
1			
Close			

**Note** This function compares only the network configurations. It does not compare the parameters of each device.

### 5-3-4 Verifying Device Parameters

The following two methods can be used to compare the parameters of network devices for verification with corresponding parameters in the Configurator.

- Comparing the parameters of devices selected from the Network Configuration Window
- Comparing the parameters through the Edit Device Parameters Window

#### Using the Network Configuration Window

#### Device - Parameter - Compare

To compare the parameters through the Network Configuration Window, use the following procedure.

1, 2, 3...

- 1. Place the Configurator online.
  - 2. Select the device.
  - 3. Select Device, Parameter, and Compare.

The following confirmation window will be displayed.



4. Click the **Yes** Button.

The progress window will be displayed and the parameters will be compared for verification.

If there are any comparison errors, the following window with the details of the errors will be displayed.

Comparison Result				
Description	Local	Device		
😡 Wrong Register Status ( #02 ).	Unregister	Register		
Wrong Register Status ( #04 ).	Unregister	Register		
Wrong OUT Allocation ( #28 ).	3200:Bit00 - 2	3400:Bit00 - 2		
Wrong IN Allocation ( #31 ).	3300:Bit00 - 2	3500:Bit00 - 2		
Wrong OUT Allocation ( #50 ).	3201:Bit00 - 2	3401:Bit00 - 2		
Wrong IN Allocation ( #50 ).	3301:Bit00 - 2	3501:Bit00 - 2		
1				
Close				

#### Using the Edit Device Parameters Window

#### Device - Parameter - Edit-Compare Button

To compare the parameters through the Edit Device Parameters Window, use the following procedure.

- *1, 2, 3...* 1. Place the Configurator online.
  - 2. Select the device.
  - 3. Edit *Device*, *Parameter*, and *Edit*.
  - 4. Click the Compare Button.



5. The following confirmation window will be displayed.



6. Click the Yes Button.

The progress window will be displayed and the comparison results will be displayed.

## 5-4 Monitoring Devices

The following description explains how to use monitor the device or communications.

**Note** The device to be monitored through the Network Configuration Window must exist on the actual network. Before monitoring the status of the monitor, upload the network configuration.

### 5-4-1 Setting Monitor Refresh Timer

#### **Option - Setup Monitor Refresh Timer**

Set the interval to refresh the device monitor display. This setting will be applied to the monitoring of any network device.

Note If the refresh timer is set to too small a value, the DeviceNet network will be overloaded and the Configurator may experience timeout error (device access error).

To set the monitor refresh timer, use the following procedure.

#### 1, 2, 3... 1. Select Option and Setup Monitor Refresh Timer.

The following window will be displayed.

Setup Monitor Refresh Timer	×
New Refresh Timer : <mark>3 - s</mark> s Range 1 - 60 s	
OK Cancel	

2. Select the new timer value and click the **OK** Button.

The set value will be enabled for the next device monitor operation.

### 5-4-2 Monitoring Devices

Device - Monitor

To monitor the device, use the following procedure.

- 1, 2, 3... 1. Place the Configurator online.
  - 2. Select the device.
  - 3. Select Device and Monitor.

A monitor window for the device will be displayed.

- Note a) A Slave can be monitored only if the slave has parameters that can be monitored in the EDS file.
  - b) Monitor cannot be selected if the device selected has no monitoring function.
- 4. To quit the monitor function, click the **Close** Button.

#### CS1W-DRM21 DeviceNet Unit

The monitor window for the CS1W-DRM21 makes it possible to monitor the following items.

Tab	Monitor item
Status	Status of the master and slaves
Unit Status	Status of the functions of the Unit
Communication Cycle Time	Communication cycle time
Error History	Error history

#### Status Tab

The Status Tab Page displays the status of the master and slaves in remote I/ O communications.

Ionitor Device		×	
Status Unit Status Communication Cycle Time	e Error History		
Master Status Remote I/O Communication Running Error In Registered Scan list Invalid Mode Message Communication Permitted I/O Data Communication not Running Comparison Error	Communication Error     Sending Error     Structure Error     Configuration Error     Node Address duplicated/Bus0ificee     Unit Memory Error	Maste Starts comm	r status remote I/O unications.
Start Remote I/D Communication Stop	Remote I/D Communication	Stops comm	remote I/O unications.
00 01 02 03 04 05 06 07 08 09 10 11 12 13 1 20 21 22 23 24 25 26 27 28 29 30 31 32 33 3 40 41 42 43 44 45 46 47 48 49 50 51 52 53 5 0 01 52 52	4 15 16 17 18 19 4 35 36 37 38 39 4 55 56 57 58 59	Slave	status
Remote I/O Communication Running     Fernote I/O Communication Error     Structure Error (Unsupported Slave)	Invalid Product Code Invalid Product Code Invalid Verdor Invalid Verdor Invalid Verdor	Detaile	ed slave status
Invalid Connection Path Invalid I/O Size Unsupported Connection	<ul> <li>Slave not Exist</li> <li>Comparison Error</li> </ul>	Node tailed	address for de- slave status
		Close	

Remote I/O communications between the master and slaves can be started or stopped by clicking the **Start Remote I/O Communication** or **Stop Remote I/O Communication** Button.

The slave status indicators change according to the status of the slaves as shown below.

Indicator	Status
Gray	Device not registered
Green	I/O communications stopped
Blue	In normal communications
Red	Communications error

To monitor the detailed status of another slave device, select the corresponding node address.

### Unit Status Tab

The Unit Status Tab Page displays the status of the Unit, master, and slave functions.

In Registered Scan list Invalid Mode  Enable Slave Function  Slave Connection Type (Automatic)  File Read/Write Error  Fice Read/Write Error  Invalid Message Timer List	Network Power Error Node Address duplicated BusOff occurred Unit Memory Error Slave Function Error occurred Master Function Error occurred
Master Function Status         VD Data Communication Running         Failed to set Communication Cycle Time         Failed to register/clear Scan list         Failed to set Fixed Allocation Area         Failed to set User Allocation Area         Failed to enable/disable Master Function	Remote I/O Refresh Error Invalid Scan list Data Remote I/O Communication Error Structure Error Comparison Error
I/O Allocation Status : Configurator Setup         Slave Function Status         Remote I/O Communication Running (0UT1/IN1)         Remote I/O Communication Running (0UT2/IN2)         Connection1 Established.         Connection2 Established.         Failed to send COS         Failed to set Fixed Allocation Area	Failed to set User Allocation Area Failed to enable/disable Slave Function Remote I/0 Beresh Error Invalid Setup Data Remote I/0 Communication Error (DUT1/IN1) Remote I/0 Communication Error (DUT2/IN2)

### **Communication Cycle Time Tab**

The Communication Cycle Time Tab Page displays the present, maximum, and minimum values of communication cycle time.

Monitor Device			×
Status Unit Status Comm	unication Cycle Time	Error History	1
Current Value :	8 ms		
Maximum Value :	14 ms		<u>C</u> lear
Minimum Value :	8 ms		<u>C</u> lear
			Close

The maximum or minimum value can be cleared by clicking the **Clear** Button.

### Error History Tab

The Error History Tab Page displays the error history recorded in the master.

Monitor Device				×
Status Unit Status Co	ommunication (	Cycle Time E	rror History	
Time of Error	Error Infor	Detailed I	Content	
01/09/29 13:59:	0345	0105	Remote I/O Communication Error.	
01/09/29 14:00:	0345	0107	Remote I/O Communication Error.	
01/09/29 14:11:	0341	0000	Network power error.	
01/09/29 14:12:	0345	013C	Remote I/O Communication Error.	
<u>U</u> pdate <u>C</u>	lear	<u>S</u> ave		
				Close

The following error history items will be displayed.

ltem	Description
Time of Error	The time the error occurred.
Error Information	Error code
Detailed Information	Detailed error information
Content	Contents of error

To refresh the display, click the **Update** Button.

To clear the error history from the master, click the **Clear** Button.

To save the file in CSV file format, click the Save Button.

#### CVM1-DRM21-V1 or C200HW-DRM21-V1

The monitor window of the CVM1-DRM21-V1 or C200HW-DRM21-V1 makes it possible to monitor the following items.

Tab	Monitor item
Status	Status of the master and slaves
Communication Cycle Time	Communication cycle time
Error history	Error history

#### Status Tab

The Status Tab Page displays the status of the master and slaves in remote I/ O communications.

Monitor Device 🛛 🔀	
Status Communication Cycle Time Error History	Master status
Master Status  Remote I/O Communication Running Communication Error Error In Registered Master Parameter Invalid Mode Structure Error Remote I/O Communication not Running Configuration Error	Starts remote I/O communications.
Comparison Error  Start Remote I/D Communication  Slave Status	Stops remote I/O communications.
00 01 02 03 04 05 06 07 08 09 10 11 12 13 14 15 16 17 18 19 00 01 02 03 04 05 06 07 08 09 10 11 12 13 14 15 16 17 18 19 00 21 22 23 24 25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57 58 59 00 55 52 52	Slave status
	Detailed slave status
Communication Error	Node address for detailed slave status

Remote I/O communications between the master and slaves can be started or stopped by clicking the **Start Remote I/O Communication** or **Stop Remote I/O Communication** Button.

The slave status indicators change according to the status of the slave as shown below.

Indicator	Status
Gray	Device not registered
Blue	In normal communications
Red	Communications error

To monitor the detailed status of another slave device, select the corresponding node address.

### **Communication Cycle Time Tab**

The Communication Cycle Time Tab Page displays the present, maximum, and minimum values of communication cycle time.

Monitor Device	×
Status Communication Cycle Time	Error History
Current Value : 3	ms
Maximum Value : 13	ms Dear
Minimum Value : 3	ms <u>C</u> lear
	Liose

The maximum or minimum value can be cleared by clicking the **Clear** Button.

### Error History Tab

The Error History Tab Page displays the error history recorded in the master.



The following error history items will be displayed.

ltem	Description
Time of Error	The time the error occurred.
Error Information	Error code
Detailed Information	Detailed error information
Content	Contents of error

To refresh the display, click the **Update** Button.

To clear the error history from the master, click the **Clear** Button.

To save the file in CSV file format, click the **Save** Button.

The monitor window of the DRT1-COM makes it possible to monitor the status and configuration.

Monitor Device	
Unit Status UNIT Hardware Error VO Refresh Running UNIT Hardware Error Error VO Res Error Structure Error Application Error Application Error	Unit status
Unit Configuration	
Unit #1: GT1-DA04MX         Unit #5: GT1-AD04           Unit #2: GT1-TS04T         Unit #6: GT1-DA04           Unit #3: GT1-TS04P         Unit #7:           Unit #4: GT1-AD08MX         Unit #8:	I/O Unit configuration
Close	

The configuration indicates the connected I/O Units. I/O Units with errors will be indicated in red.

#### Fiber Amplifier Communications Unit

With the E3X-DRT21 Fiber Amplifier Communications Unit, the following window makes it possible to monitor the status of connected sensors.

onitor Device			×	
Status				
Number of conne	ecting sensor : 6	6 Counts		
Number of communica	atable sensor : 6	6 Counts		
Sensor status (OK)	🗖 Sensor stat	us (NG)		
Number of conectting	g sensor : Massa	age enable		Monitors the light amount of each senso
	ON/OFF Light	t nt	ON/OFF Light	
Sensor #1 : E3X-DA6-P	12	Sensor #9 :		
Sensor #2 : E3X-DATW	0 14	4 Sensor #10		
Sensor #3 : E3X-DATW	0 14	Sensor #11		
Sensor #4 : E3X-DA6	0 🔽	3 Sensor #12		
Sensor #5 : E3X-DA6	0 10	Sensor #13		
Sensor #6 : E3X-DA6	0	Sensor #14		
Sensor #7 :		- Sensor #15		- Monitors ON/OFF
Sensor #8 :		- Sensor #16		status. Black is ON, yellow is OFF.
			Close	

# MULTIPLE I/O TERMINAL

### **Other Slaves**

Devices can be monitored if the parameters of the devices are specified in EDS files to allow monitoring.

Parameters	
Parameter Name	Value
🧐 0008 Drive Running Forward	
🧐 0009 Drive Running Reverse	
🕼 0010 Drive Ready	
🧐 0011 Drive Fault	
🕼 0012 Drive Warning	
🧐 0015 At Reference	
🧐 0017 Actual Speed	
🧐 0018 Actual Current	
🧐 0020 Actual Power	
🕼 0022 Output Voltage	
🧐 0261 U1-01 Frequency Ref	
9 0262 U1-02 Output Frequency	
🕼 0263 U1-03 Output Current	
🧐 0264 U1-04 Control Method	
🧐 0265 U1-05 Motor Speed	
🕼 0266 U1-06 Output Voltage	
🧐 0267 U1-07 DC Bus Voltage	
🕼 0268 U1-08 Output kWatts	
@ 0269111-09 Torque Beference	
	Close

# **SECTION 6** Manipulating Files

This section provides information on manipulating files, including saving and reading procedures.

6-1	Saving	and Reading Files	130
	6-1-1	Saving Network Configuration Files	130
	6-1-2	Reading Network Configuration Files	130
	6-1-3	Saving Device Parameter Files	130
	6-1-4	Reading Device Parameter Files	131
6-2	EDS Fi	le Management	131
	6-2-1	Installing EDS Files	131
	6-2-2	Creating EDS Files	132
	6-2-3	Deleting EDS Files	133
	6-2-4	Saving EDS Files	133
	6-2-5	Searching EDS Files	134
	6-2-6	Displaying EDS File Properties	134
6-3	Printing	g	135
6-4	Using C	General-purpose Tools to Set Devices	135
	6-4-1	Setting Device Parameters with Class Instances	135
	6-4-2	Setting Node Addresses and Baud Rates through the Network	138
6-5	Exporting Data Created in Configurator		139
	6-5-1	Exporting the Network Configuration List	139
	6-5-2	Exporting I/O Comments	139
	6-5-3	Exporting Device Parameters	140
6-6	Importi	ng Files Created in Version 1 of the Configurator	140
	6-6-1	Importing Network Configuration Files	140
	6-6-2	Importing Device Parameter Files	141
6-7	Optiona	al Functions	141
	6-7-1	Installing Expansion Modules	141
	6-7-2	Adding Vendors and Device Types	141
# 6-1 Saving and Reading Files

This section explains the network configuration of the Configurator and how to save or read device parameter files.

Note By saving the network configuration or device parameters in files, the device parameters can be read and set to new devices that will replace faulty devices, to make it easier to restore the network configuration.

The following files can be saved or read.

Files	Description	Extension
Network configuration files	A parameter file for all devices (master and slaves) on a single virtual network in the Network Configuration Window	.npf
Device parameter files	Details of the parameters for each device (master or slave)	.d∨f

# 6-1-1 Saving Network Configuration Files

#### Network - Save (or Save As)

To save the Network Configuration in the Configurator, use the following procedure.

- 1, 2, 3... 1. Select Network.
  - 2. Select Save (or Save As).

The first time *Save* is selected, a dialog box will be displayed to specify the file name. If the file has been saved before, the file will be saved immediately, overwriting the previously saved file.

If *Save as* is selected, a dialog box will be displayed to specify the file name.

3. Input the file name and click the **OK** Button to save the network configuration.

Note When the network configuration is saved, the parameters of the registered devices on the network will be saved simultaneously.

## 6-1-2 Reading Network Configuration Files

Network - Open

To read the network configuration file, use the following procedure.

1, 2, 3... 1. Select Network.

2. Select Open.

A window to specify the network configuration file will be displayed.

3. Input the file name and click the **Open** Button.

If the displayed network configuration file has not been saved yet, a confirmation window will be displayed; click the **Yes** Button.

## 6-1-3 Saving Device Parameter Files

**Device - Parameter - Save** To save the parameters of the device, use the following procedure.

- 1, 2, 3... 1. Select the device.
  - 2. Select *Device*, *Parameter*, and *Save*.

A window to specify the name of the folder in which to save the parameters and the name of the parameter file will be displayed.

3. Input the folder and file names and click the **Save** Button. The parameters will be saved in the file.

#### 6-1-4 Reading Device Parameter Files

Device - Parameter - Open

To load the parameters of the device, use the following procedure.

- 1, 2, 3... 1. Select the device.
  - 2. Select Device, Parameter, and Open.

A window to specify the file name of the device parameters to be read will be displayed.

3. Input the file name and click the **Open** Button.

If the device parameters have been modified, a confirmation window will be displayed. Click the **Yes** Button.

# 6-2 EDS File Management

This section explains the EDS file management function of the Configurator.

#### 6-2-1 Installing EDS Files

EDS File - Install

The Configurator will support new devices if proper EDS files are installed. To install the EDS file, use the following procedure.

#### 1, 2, 3... 1. Select EDS File and Install.

The following window will be displayed.

iii 0930DSL009.eds           iii C200HW-DRM21.eds           iii C200HW-DRM21.eds             iii 20000816101447.EDS           iii C200HW-DRT21.eds           iii C200HW-DRT21.eds           iii C200HW-DRT21.eds             iii 3G3FV-PDRT1-SIN.eds           iii CPM2B-DRT.eds           iii DRT1-232C2.eds             iii 3G3MV-PDRT1-SINV.eds           iii CPM2C-S100C-DRT.eds           iii DRT1-AD04.eds             iii 3G8F7-DRM21.eds           iii CPM2C-S110C-DRT.eds           iii DRT1-AD04H.eds             iii 3G3FV-PDRT1-SINV.eds           iiii CPM2C-S110C-DRT.eds           iii DRT1-AD04H.eds             iii 3G515-A003.EDS           iii CQM1-DRT21.eds           iii DRT1-B7AC.eds
File <u>n</u> ame: DRT1-232C2.eds Qpen
Files of type: Electronic Data Sheet(*.eds)  Cancel
Device Information Vendor : OMRON Corporation Device Type : Generic Device Product Name : DRT1-232C2 Revision : 1.04

The device information will be displayed on the bottom of the window when the EDS file is selected.

2. Select the EDS file to be installed and click the **Open** Button.

The EDS file will be added to the Hardware List.

If the EDS file already exists, the new EDS file will overwrite the previous one.

If the EDS files are different to each other in version, the new EDS file will be added to the Hardware List as shown below.



### 6-2-2 Creating EDS Files

EDS File - Create

The EDS files are required by the Configurator to create a network configuration. To create a EDS file, use the following procedure.

#### 1, 2, 3... 1. Select EDS File and Create.

The following window will be displayed.

reate EDS File				×
Device Information				
Vendor ID : 47	Vendor Name :			
Device ID : 12	Device Type :			
Product Code 51	Product Name :	C200HW-D	RT21	
Major Rev. : 1	Catalog :			
Minor Rev. : 3			Upload from	Device
Default I/O ✓ Poll ■ Bit-Strobe ■ COS ■ Cyclic				
	· · ·	-IN		
Size : 64 By	lte	Size :	64 Byte	
Valid Bits : 0 (0	) :Valid all Bits )	Valid Bits :	0 (0:Val	id all Bits )
Name : OUT Data		Name :	IN Data	
Path :		Path :		
Help :		Help :		
			OK	Cancel

- Set the device information and I/O information.
   The device information can be obtained from the device on the network if the network is online.
- 3. To read the information, click the **Upload from Device** Button. The following window will be displayed.

Target Device	×
Target Node Address	: 🚺 💻 Setup Range 0 - 63
ОК	Cancel

- Set the node address of the device and click the **OK** Button.
   For the I/O connections and I/O size of the device, refer to the operation manual of the device.
- 5. Click the **OK** Button.

The device will be added to the Hardware List.

Note Device parameters cannot be set with the EDS file creation function of the Configurator. Obtain a proper EDS file from the manufacturer of the device to make device parameter settings for the device.

## 6-2-3 Deleting EDS Files

#### EDS File - Delete

To delete the EDS file, use the following procedure.

- *1, 2, 3...* 1. Select the device from the Hardware List.
  - 2. Select *EDS File* and *Delete*.

The following confirmation window will be displayed.



3. Click the **Yes** Button.

The device will be deleted from the Hardware List together with the EDS file.

#### 6-2-4 Saving EDS Files

EDS File - Save

To save the EDS file, use the following procedure.

1, 2, 3...

- 1. Select the device from the Hardware List.
- 2. Select EDS file and Save.

The following window will be displayed to specify the name of the folder where the EDS file will be saved and the name of the EDS file.

Save EDS File 🛛 😵 🔀			
Save in: 🔄 Eds	•	• 🗈 💣 🎟 •	
O930D5L009.eds     20000816101447.ED5     3G3FV-PDRT1-SIN.eds     3G3MV-PDRT1-SINV.eds     3G8F7-DRM21.eds     515-A003.ED5	C200HW-DRM21.eds     C200HW-DRT21.eds     C200HW-DRT21.eds     CPM2B-DRT.eds     CPM2C-S100C-DRT.eds     CPM2C-S110C-DRT.eds     CQM1-DRT21.eds	CS1W-DRM21.eds CVM1-DRM21.eds CVM1-DRM21.eds CVM1-DRM21.eds CVM1-232C2.eds CVM1-232C2.eds CVM1-AD04.eds CVM1-AD04.eds CVM1-AD04H.eds CVM1-B7AC.eds CVM1-CVM1-CVM1-CVM1-CVM1-CVM1-CVM1-CVM1-	
File name:	- Cl 1(x - 1.)		
Device Information Vendor : Device Type : Product Name : Revision :	ta Sheel(".eds)		

3. Input the folder and file names and click the **Save** Button The EDS file will be saved.

#### 6-2-5 Searching EDS Files

EDS File - Search

To search the device (EDS file) displayed in the Hardware List, use the following procedure.

#### 1, 2, 3... 1. Select the EDS File and Search.

The following window will be displayed.

Find EDS File	? ×
Find what:	Eind Next
	Cancel
Match <u>c</u> ase	

2. Input the character string and click the Find Next Button

The cursor will move to the position of the corresponding device closest to the present cursor.

3. To quit the search operation, click the Cancel Button

1. Select the hardware (device) from the Hardware List.

- Note a) The device will be found if it is located below the present cursor position.
  - b) Select *Hardware* in the Hardware List before using the above procedure to search all the devices.

#### 6-2-6 Displaying EDS File Properties

**EDS File - Property** 

To display the properties of the EDS file, use the following procedure.

1, 2, 3...

2. Select EDS File and Property.

The following window will be displayed.

C	51W-DRM21 Rev	1 Property	×
	General		
	CS1W-	DRM21	
	Description :	CSW1-DRM21 EDS File	
	Create Date :	05-11-2000 12:00:00	
	Modify Date :	05-11-2000	
	Revision :	1.0	
	Vendor :	OMRON Corporation	
	Device Type :	Communications Adapter	
	Product Name :	2	
	Revision :	1.01	
	Catalog :		
			Close

The time and date of the creation of the EDS file will be displayed along with device information on the file.

# 6-3 Printing

**Network - Print** 

The following description explains how to print the device parameters of the network configuration or I/O comment data presently displayed. To print them, use the following procedures.

- 1, 2, 3... 1. To print a partial list of device parameters, select the desired devices.
  - 2. Select Network and Print.

The following window will be displayed.

Print	X
Printer Model : Canon LASER SHOT LBP-840	Property
Range of Printed Device	Parameters to be Printed Device <u>P</u> aramete
O Selected Devices only	C 1/0 <u>C</u> ommei
	OK Cancel

3. Select the devices from the following ones.

All devices: The device parameter or I/O comment data of all the devices of the network configuration will be printed.

Selected devices: Only the device parameter or I/O comment data of the selected devices will be printed.

Both items cannot be selected.

4. Select the printing items from the following ones.

Device parameters: The device parameter data will be printed.

I/O comment: The I/O comment data of the devices will be printed. Both items cannot be selected.

- 5. If necessary, click the **Property** Button to change the printer settings. The printer settings depend on the printer being used.
- Click the OK Button to print the parameters or I/O comment data. When the I/O comment data is selected, the data on the following devices will be printed.

Device type	I/O comment data	
Master device	Master I/O comment data, if it exists or the I/O comment data of the devices registered in the scan list will be printed. If the registered slave devices support a number of connections, the I/O comment data on the connection used by the master will be printed.	
Slave devices	Slave I/O comment data. If the Slave devices support a number of connections, the I/O comment data on all devices including those not registered with the master will be printed.	

# 6-4 Using General-purpose Tools to Set Devices

This section explains how to set the device parameters with no corresponding EDS files or set the node addresses or baud rates through the network.

# 6-4-1 Setting Device Parameters with Class Instances

#### **Tools - Setup Parameters**

Device parameters for devices that have no corresponding EDS files can be set by specifying the following items.

- · Service code
- Class (object class), instance (class instance) and attribute (instance attribute)

Before setting the device parameters, it is also necessary to obtain the data setting information on attributes other than the above from the manufacturer of the device.

Unless all the information is known, the device parameters cannot be set. Use the following procedure to set the device parameters.

- *1, 2, 3...* 1. Place the Configurator online.
  - 2. Select *Tools* and *Setup Parameters*.

The following window will be displayed.

Setup Parameters 🛛 🗙			
Target Node Ad	Idress		
	Setup Range 0 - 63		
Service			
Generic	Apply Attributes		
C Custom	Service code set in HEX format string.		
Parameter			
Class : 0	All parameters set in HEX format string.		
Instance : 1	Attribute data set in Data field.		
Data :			
Result :			
<u>S</u> end	Close		

- 3. Set the *Target Node Address* to the node address of the device.
- 4. Designate the service.

There are two ways to designate the service code, using the generic service code defined by DeviceNet or designating it directly. To use the generic service code defined by DeviceNet, check the **Generic** option, then select the desired service from the drop-down list. To designate the service code directly, check the **Custom** option, then input the service code in a HEX format string.

Setup Parameters 🛛 🗙				
Target Node Address				
0 -	Setup Range 0 - 63			
Service				
<ul> <li>Generic</li> </ul>	Apply Attributes			
C Custom	Apply Attributes			
Parameter	Find Next Object Instance			
Class : 0	Get Attribute List Get Attribute Single			
Instance : 1	Get Attributes All Get Member Insert Member			
Data :	No Operation (NOP) Remove Member Reset			
Result :				
<u>S</u> end	Close			

#### **Designating the Service Code Directly**

Setup Parameters 🛛 🗙					
Target Node Ac	ldress				
0 -	Setup Range 0 - 63				
Service					
O Generic	Apply Attributes				
Custom	Service code set in HEX format string.				
- Parameter					
Class : 0	All parameters set in HEX format string.				
Instance : 1	Attribute data set in Data field.				
Data :					
Result :					
<u>S</u> end	Close				

- 5. Designate the class and instance parameters for reading and writing data.
- 6. Input the data corresponding to the designated service.
- 7. After all of the items have been input, click the **Send** Button. The device response will be displayed in the *Result* area.
- 8. Click the **Close** Button to quit device parameter setting. This completes the device parameter setting.
- 1, 2, 3... 1. Parameter reading example
  - a) Check the **Generic** option in *Service*, then select *Get Attribute Single* from the drop-down list.
  - b) Designate the class and instance of the parameter to be read.

- c) Input *Data Area* for the *Attribute* of the parameter to be read.
- d) Click the **Send** Button. The value that was read will be displayed in the *Result* area.
- 2. Parameter setting example
  - a) Check the **Generic** option in *Service*, then select *Set Attribute Single* from the drop-down list.
  - b) Designate the class and instance of the parameter to be set.
  - c) Input Data Area for the Attribute of the parameter to be set.
  - d) Input the attribute followed by the value to be set for the parameter in the Data area.
  - e) Click the Send Button.

#### 6-4-2 Setting Node Addresses and Baud Rates through the Network

#### Tools - Setup Node Address/Baud Rate

To set the node address and baud rate for the device through the network, use the following procedure.

- Set the DeviceNet network so that only the objective device and the Configurator are in operation on the network. Refer to the operation manual of the device for the factory-set node address and baud rate of the device. Connect the Configurator to the network at the same baud date.
  - 2. Place the Configurator online.
  - 3. Select Tools and Setup Node Address/Baud Rate.

The following window will be displayed.

Setup Node Address/Baud rate
Target Node Address
Setup Range 0 - 63
Change Node Address
New Node Address : 0 Setup Range 0 - 63
<u>C</u> hange
Change Baud rate
Current Setup :
New Setup : 125K Bit/s 💌 Change
Close

- 4. Set the *Target Node Address* to the present node address of the objective device.
- 5. To change the node address, set the *New Node Address* to the new node address and click the **Change** Button.

The node address will be changed.

6. To change the baud rate, set the *New Setup* to the new baud rate and click the **Change** Button.

The baud rate will be changed.

Note The node address and baud rate can be set to the device through the network only if the device is compatible with this function.

# 6-5 Exporting Data Created in Configurator

This section explains exporting data created in the Configurator in files for other software applications.

## 6-5-1 Exporting the Network Configuration List

**Network - Export** 

Use this command to export the content of the Detailed Display in CSV format.

X	Microsoft Excel	- Device	List.csv						- 🗆 ×
	Eile Edit View	Insert i	F <u>o</u> rmat <u>T</u> ool	s <u>D</u> ata <u>W</u> ini	low <u>H</u> elp				_ 8 ×
Ĩ	- ) 🛩 🖪 🚑	👌 💞	<u>∦</u> ⊑∎ E	. 🛷 🗠 🗸	α - 🍓 😤 Σ	f* 👌 🟹 🛍 🥥 🚜 1	00% - 🧭		
Π A	Arial	• 9	• B .		≣≣፼\$%	, <u>*</u> , *,% ∰ ∰ ₩ +	ð - <u>A</u> -	-	
	A7	•	=						
	Α		В	С	D	E	F	G	H
1	Product Name		#	##	Vendor	Device Type	Rev	Comment	
2	DRT1-ID16		#00	#63	OMRON Corporation	Generic Device	1.03	Node A	
3	DRT1-0D16		#01	#63	OMRON Corporation	Generic Device	1.03	Node B	
4	CPM2C-S1100	C-DRT	#10	#63	OMRON Corporation	Communications Adapter	1.01	Node C	
5	E3X-DRT21		#20	#63	OMRON Corporation	Communications Adapter	1.01	Node D	
6	CJ1W-DRM21		#63		OMRON Corporation	Communications Adapter	1.01	Scanner	
7			1						
I I I DeviceList / I I I I I I I I I I I I I I I I I I							► I		
Re	eady								

# 6-5-2 Exporting I/O Comments

Device - Export - I/O Comment List

It is possible to export device I/O comment data in CSV format.

The following window is an example of CSV-formatted I/O comment files read in Excel.

X	dicrosoft Ex	cel - 140te:	kt.csv								
12	] <u>F</u> ile <u>E</u> dit !	/iew Insert	Format Too	ıls <u>D</u> ata <u>W</u> i	ndow <u>H</u> elp						_ & ×
	) 😅 🖬 🛛	i 🕹 🗟 🗳	አ 🖻 f	l 💅 🗠	• ca - 🌘	δ 🍪 Σ	f <sub>×</sub> <mark>A</mark> ↓ Z↓	۰ 🔮 🛍	100% 🗖	· 🖗	
Ar	ial	• 1	0 • B	I <u>U</u> ≣		\$ %	00. 0.+ 0.+ 00. <b>t</b>	tje tje	🔛 • 🔕 •	<u>A</u> -	
	G2	-	<b>=</b> #01								
	A	В	С	D	E	F	G	H		J	К
1	Area	MAC ID	Comment	Bit No.	Bit Comm	ent	CX-P Valia	able Table			
2	3200.00	#01	DRT1-OD1	Bit00	OUT Com	mnet 1	#01	3200.00	OUT Comr	nnet 1	
3	3200.00	#01	DRT1-OD1	Bit01	OUT Comr	mnet 2	<b>#</b> 01	3200.01	OUT Comr	nnet 2	
4	3200.00	#01	DRT1-OD1	Bit02	OUT Com	mnet 3	<b>#</b> 01	3200.02	OUT Comr	nnet 3	
5											
6											
Rea	ady					Sum	=6400.03		NUN	N	

Use the following procedure to export the I/O comment data.

- 1, 2, 3... 1. Select the device.
  - 2. Select Export and I/O Comment List.

- 3. If valid I/O comments exist, a window will be displayed to specify the name of the folder where the I/O comment data is to be exported and the name of the file.
- 4. Input the folder and file names and click the **Save** Button.

The I/O comment data will be exported.

The following types of I/O comment data will be exported according to the device selected.

Device	I/O comment data
Master device	Master I/O comment data, if it exists, or the I/O comment data of the devices registered in the scan list. If the registered slaves support a number of connections, the I/O comment data on the connection used by the master will be exported.
Slave devices	Slave I/O comment data. If a slave supports a number of connections, the I/O comment data on all devices including those not registered with the master will be exported.

Note Part of I/O comment data exported in the CSV format can be opened in spreadsheet software and used for CX-Programmer variable tables.

# 6-5-3 Exporting Device Parameters

*Device - Export -* (Export type)

Device parameters of OMRON'S CVM1-DRM21-V1, C200HW-DRM21-V1, and CS1W-DRM21 can be exported in the DRM\_UNIT file format for open network controllers or the NX-Server file format for the DeviceNet.

Use the following procedure to export the device parameters of the Master

- 1, 2, 3... 1. Select the device.
  - 2. Select Export.
  - 3. Select the export type.

The following three export types are supported.

Export type	Description
NX-Server DDE file	Scan list information is output in the NX-Server DDE Edition file format.
NX-Server ONC file	Scan list information is output in the NX-Server ONC Edition file format.
ONC Master file	Scan list information is output in the ONC Master parameter file format.

If valid devices are registered with the scan list, a window will be displayed to specify the name of the folder, where the device parameters are exported, and the name of the file.

4. Enter the folder and file names and click the **Save** Button

The device parameters will be exported.

To use the exported file, refer to the operation manual on the software or hardware of the device.

# 6-6 Importing Files Created in Version 1 of the Configurator

This section explains the import function to read network configuration files and device parameter files created in version 1 of the Configurator (Ver.  $1.\square\square$ ).

# 6-6-1 Importing Network Configuration Files

Network - Import

To read the network configuration file (.ntf) created in version  $1.\Box\Box$  of the Configurator, use the following procedure.

1, 2, 3... 1. Select Network and Import.

A window to specify the name of the network configuration file to be imported will be displayed.

2. Input the file name and click the **OK** Button.

When the displayed network configuration file has not been saved yet, a confirmation window will be displayed; click the **Yes** Button.

#### 6-6-2 Importing Device Parameter Files

**Device - Import** 

Use the following procedure to read the master parameter file (.dsf) or slave parameter file (.dpf) created in version 1 of the Configurator.

- *1, 2, 3...* 1. Select the device.
  - 2. Select *Device* and *Import*.

A window to specify the name of the device parameter file to be imported will be displayed.

3. Input the file name and click the **Open** Button.

The device parameters will be imported.

# 6-7 Optional Functions

The optional functions make it possible to set Expansion Modules in the Configurator for new devices, add DeviceNet device vendors, and add new device types.

## 6-7-1 Installing Expansion Modules

#### **Option - Install Plugin Module**

To install the expansion module, use the following procedure.

1, 2, 3... 1. Select Option and Install Plugin Module.

A window to specify the name of expansion module set file will be displayed.

2. Input the file name and click the **Open** Button.

The Expansion Module will be added to the Configurator.

## 6-7-2 Adding Vendors and Device Types

#### **Option - Edit Configuration File**

Use the following procedure to add a new device vendor or device type.

- 1, 2, 3... 1. Select Option and Edit Configuration File.
  - 2. Select Vendor ID/Device Type List.

ID	Vendor Name			
( <b>0</b> ,1	Allen-Bradley Company, Inc.			
<u>a</u> 2	Namco Controls Corp.			
ā 4	Parker Hannifin Corp.			
	Rockwell Automation/Reliance Electric			
<b>.</b> 6	Schrader Bellows			
Q7	SMC Corp.			
<u>@</u> 8	SST (S-S Technologies, Inc.)			
Q 9	Western Reserve Controls Corp.			
ig 10	Advanced Micro Controls Inc. (AMCI)			
<b>@</b> 11	ASCO Pneumatic Controls			
<b>@</b> 12	Banner Engineering Corp.			
ig 13 🛛	Belden Wire & Cable Company			
<b>@</b> 14	Crouse-Hinds Molded Products			
ا 📴 📴	Daniel Woodhead Co.			
ing 17 👘 📴	Dearborn Group Technology, Inc.			
iiiiiii 20	Huron Net Works			
iiiiii 21	Lumberg, Inc.			
i 📴 22	Automation Value LLC (Online Dev)			
	<u>N</u> ew <u>E</u> dit <u>D</u> elete			

The following window will be displayed.

- To add or modify the vendor information, click on the Vendor Tab. To add new vendor information, use the following procedure.
  - a) Click the **New** Button.

The following window will be displayed.

Edit 🛛	×
ID : 10	
Description :	
OK Cancel	

- b) Input the new vendor ID and vendor name and click the **OK** Button. The new vendor will be added to the list.
- To edit existing vendor information, use the following procedure.
- a) Select the vendor.
- b) Click the **Edit** Button.

The following window will be displayed.

Edit 🛛 🗙
ID : 47
Description : OMRON Corporation
OK Cancel

- c) Input the new vendor name and click the **OK** Button.
- To add or change the device type, click on the **Device Type** Tab.
   The method of adding or changing the device type is the same as that of adding or changing vendor information.
- 5. When all the settings are completed, click the **OK** Button.

# Appendix A Error Messages

The following table lists the error messages that might be displayed during operation.

Error message	Meaning
"Printing can not be started."	The printing process could not be started. Check the printer and start
	printing again.
"Instance not specified."	The specified instance data is incorrect. Set correct data.
"Interface can not be opened."	The specified interface could not be opened. If the interface is being used by another application, the user should quit the application and open the interface again.
"Interface can not be go on-line."	The interface could not be placed online.
"Specified Interface can not be gone to On–line."	The interface could not be placed online. Check the settings and set the interface to online again.
"Interface can not be closed."	The interface could not be closed.
"Object can not be created."	The internal object could not be created.
"Can not execute due to on-line."	Operate offline.
"Wrong Start Words."	The start word is incorrect. Set the start word within the permissible range.
"2 bytes or more can not be allocated to odd address."	Two bytes or more cannot be allocated to odd addresses. Set to even addresses.
"Class not specified."	The specified class data is set incorrectly. Set the data correctly.
"Connection type not specified."	Valid connections are not set. Set the connections correctly.
"Wrong Connection Path."	The connection path length is incorrect. Set the connection path length correctly.
"Connection can not be opened."	The connection could not be opened.
"Connection can not be closed."	The connection could not be closed.
"Not support operation."	The operation is not supported by the specified device.
"Invalid File Type."	The specified file cannot be accessed. Set the correct file.
"This device can not register."	This device cannot be registered.
"Sending command failed."	The command could not be transmitted. Transmit the command again.
"No more character strings specified found."	No more corresponding character strings exist.
"Not able to add any more device to the list."	No more devices can be added.
"Server can not be started."	The server function could not be started.
"Exceeds the number of connection s permitted."	The number of connections cannot be supported. Limit the maximum num- ber of connections to two.
"Part or all of specified area Allocated to other device."	Part or all of the specified area has already been allocated to another device. Either allocate a different area or change the other device's I/O allocation.
"Part or all of specified area Allocated to other block."	All or part of the specified memory block area is already allocated to another memory block area. Either specify a different area or change the settings of the other memory block area.
"All or part of assigned area are conflicting with I/O data already allocated for other device.\nOr, it will exceed the permitted channel size."	All or part of the specified area is allocated to another I/O area or in excess of the permissible number of words. Either specify a different area or change the settings of the other I/O area.
"Specified file can not use this Device."	The specified parameter file was created for a different model. Set the correct file.
"Specified unit can not be accessed."	The specified CPU Bus Unit could not be accessed. Check the unit number and access the specified CPU Bus Unit again.
"Specified unit not CS1W-DRM21."	The specified CPU Bus Unit is not the CS1W-DRM21. Set the unit number for the CS1W-DRM21.

Error message	Meaning
"Specified ID already used."	The specified node address has already been used by another device. Set another node address.
"Specified MAC ID already used with other device."	The specified node address has already been used by another device. Set another node address.
"Specified interface already opened."	The specified interface has already been opened by another application. Quit the application and set the interface again.
"Newly Plugin module is not able to be configured because is currently being used."	The specified Expansion Module cannot be refreshed because the device presently displayed is in use. Restart the Configurator and set the Expansion Module again.
"Can not register to own device."	The master device cannot be registered with itself.
"Receive thread can not be created."	The internal reception thread could not be created.
"No usable interface."	A usable interface has not been set. Register the interface.
"Usable printer has not been set."	A usable printer has not been set. Register the printer.
"Status can not be got."	The interface status could not be obtained.
"Already set."	The specified data has already been set.
"Already register."	The specified device has already been registered.
"Slave's total channel size over."	The size per slave is in excess of the maximum IN/OUT size that can be registered in the scan list. Set the IN/OUT size correctly.
"Slave's total channel size over."	The number of words per slave is in excess of the maximum number of per- missible words that can be registered with the scan list. Set the number of words within the permissible range.
"Out of setup memory block."	The number of bytes occupied from the start word is in excess of the mem- ory block. Set the number of bytes occupied within the permissible range.
"Wrong Setup file."	The setup file is incorrect. Set the correct file.
"Setup file can not be copied."	The setup file could not be copied. Check the empty space of the disk and copy the file again.
"Setup possible parameters not exist."	The specified device has no parameters that can be set.
"Out of Setup Range."	The specified area is in excess of the possible setting range. Set the number of occupied words within the permissible range.
"Invalid parameter."	The set value is incorrect. Set the correct value.
"Wrong Occupied Bytes."	The number of allocated bytes is incorrect. Set the size correctly.
"Wrong Channel Counts."	The number of specified words is incorrect. Set the number of words within the permissible range.
"Register Slave to be tested not exist."	No slaves to be tested exist. Set the slaves correctly.
"Specified Device ID being used by Configurator."	The node address coincides with that of the Configurator. Set a different node address.
"Device not selected."	The target device has not been selected. Select the target device.
"Specified device can not be accessed, or wrong device type."	Either the specified device does not exist or the device in the Configurator is a different type to the actual network device. Check and set the device again.
"Device can not be accessed."	The specified device could not be accessed. Check and set the device again.
"Invalid device type."	The device in the Configurator does not coincide with the actual network device. Check and set the device.
"Default I/O not specified."	The default connections have not been specified. Set the default connections.
"Found registered device which is not yet assigned to allocate to memory."	I/O allocations to all registered devices have not been completed. Complete I/O allocations.
"Out of register possible I/O size."	The I/O size is in excess of the permissible I/O size that can be registered. Set the I/O size correctly.
"I/O data size mismatch detected in the reg- istered device. I/O size in the Scan list will	The I/O size of the actual slave is different from the I/O size registered in the parameters.
be used."	Set the correct I/O size for the slave.
	If no EDS file is installed (in which case the I/O size will be zero), install the EDS file.

Error message	Meaning
"Filename not specified."	The file name has not been specified. Set the file name.
"Save folder not specified."	The save folder has not been specified. Set the folder.
"Invalid EDS File."	The EDS file is not valid. Set the correct EDS file.
"Not enough memory."	The command cannot be executed because of insufficient memory. Quit the other application and execute the command again.
"Unit mode can not be changed."	The mode could not be changed. Check the device and execute again.
"Valid I/O Comment not exist."	The specified command cannot be executed because no valid I/O comment exists.
"Failed to find valid I/O data."	Valid I/O data does not exist. Set the I/O size correctly.
"No valid connection exists, or exceeding the number of connections permitted."	No valid connection exists or the specified number of connections is excessive. Set the connections correctly or limit the number of connections to two.
"No valid data."	There is no valid data. Check the parameters.
"Valid Device not exist."	There is no valid data in the scan list. Register the device.
"Valid Slave not exist."	There is no valid slave.
"Wrong Valid Bits."	There is no valid bit. Set the correct bit.
"Wrong Unit Configuration."	The Configurator and network device are different to each other in unit configuration. Check and amend the difference.
"Wrong Unit Type."	The type of the master specified in the file is wrong. Specify the correct file.
"Specified Parameter read only."	The data is read-only. Check the parameter.
"Receiving response failed."	The response could not be received. Try again.
"No valid memory area for allocation found."	The memory block has no possible allocation area. Check the size of the memory block.
"Out of allocate possible maximum number of words."	The number of words is in excess of the possible number of words that can be allocated to the master. Set the number of words within the possible range.
"Wrong Allocated Words."	The number of words is incorrect. Set the number of words within the possible range.
"Bit-Strobe has no valid I/O data."	There is no valid I/O data for bit-strobe connections. Set the size correctly.
"Wrong Bit-Strobe's IN data size."	The bit-strobe connection I/O size is incorrect. Set the size within eight bytes.
"Wrong Bit–Strobe Connection I/O Size."	The bit-strobe connection I/O size is incorrect. Set the size within eight bytes.
"Bus Off occurred."	A bus-off error occurred in the network. Make an online connection again.
"COS has no valid I/O data."	There is no valid I/O data for COS connections. Set the size correctly.
"COS and Cyclic can not be set simultaneously."	Both COS and cyclic connections cannot be set together.
"Cyclic has no valid I/O data."	There is no valid I/O data for cyclic connections. Set the size correctly.
"Wrong I/O Size."	The I/O size is incorrect. Set the I/O size correctly.
"This device does not have I/O data."	The device cannot be registered with the scan list because the device has no data.
"Input IN/OUT Size in even-numbered bytes."	The number of IN/OUT bytes is not an even number. Set an even number.
"Wrong PLC Type."	The mounted PC is wrong. Set the correct PC.
"Mismatch detected between POLL OUT data and COS/Cyclic OUT data."	The poll and COC/cyclic connections are different to each other in OUT size. Set the same OUT size.
"Poll has no valid I/O data."	There is no valid I/O data for poll connections. Set the size correctly.
"Set in HEX format string."	Input a hexadecimal string.

# Appendix B Troubleshooting

Problem	Remedy
The DeviceNet Configurator will not go online when the PCMCIA Card is used.	<ol> <li>The following causes are probable. Take the necessary countermeasures as provided below.</li> <li>The PCMCIA card may not be recognized correctly. Select <i>Control Panel, System</i>, and <i>Device Manager</i>, and check that 5136-DN-PCM(n), where n is the PCMCIA slot number, is in the SST devices. Select <i>Property</i> and <i>Resource</i> and check that the resource settings are made properly in an unallocated area. If the PCMCIA Card is not recognized correctly, dismount and mount the Card again. If the Card is still not recognized, install the PCMCIA Card again.</li> <li>The above resource set value and Card driver set value may be different to each other. Set the correct value in the 3G8xx-DRM21 Configuration Software</li> </ol>
Not all devices are displayed when the network is uploaded.	<ul> <li>The following causes are probable. Take the necessary countermeasures as provided below.</li> <li>1. The master may have a configuration error or comparison error. If the master has a configuration error or comparison error, data cannot be read from the slave correctly. Turn OFF the master, create and write parameters to the Master, and execute <i>List Devices</i>.</li> </ul>
	<ol> <li>The network may be unstable. Proper communications will not be available if the network is unstable due to noise, line disconnection, improper contact loose connectors, or loose connector screws. As a result, all the devices may not be displayed. Check the above items.</li> <li>The network load may be too high. The network load will be excessively high in the following cases. As a re- sult, the Configurator may have a timeout error and all the devices may not be displayed.</li> <li>More than one master is connected to the network.</li> <li>High-density slaves are connected to the network.</li> </ol>

# Appendix C Error History Information

The following table shows error information recorded in the DeviceNet Master Unit or DeviceNet Unit and messages that are displayed by selecting **Device Monitor** and **Error History Display** in the DeviceNet Configurator.

# CVM1-DRM21-V1 and C200HW-DRM21-V1

Error	Error details		Error		Meaning	7-
code (Hex)	Upper 2 digits	Lower 2 digits				segment display on Unit
0002	Code from	Error node	PC interface e	error	Error occurred in PC interface.	F6
0006	7-segment display on	address	Initialization e	error with PC	Error occurred in initialization with PC.	F5
000B	Offic		Routing table error		Error in routing table data.	E5
0101	Destination node address	Frame discriminator (automatically	Send response message	Not in network	The local node is not in network; attempted to send response mes- sage, but message was destroyed.	
0103		set by system when FINS command is	destroyed	Local node not participating	Send error occurred; attempted to send response message, but message was destroyed.	
0109		56()		Remote node busy	Remote mode was busy; attempted to send response message, but message was destroyed.	
0112				Illegal header	An illegal header was detected; attempted to send response mes- sage, but message was destroyed.	
0117	Source node address		Receive response message destroyed	Reception buffer full	The internal reception buffer in the local node was full; attempted to receive response message, but message was destroyed.	
0118				Illegal message	An illegal message was received and destroyed.	
0119	Destination node address		Send response message destroyed	Local node busy	Local node was busy; attempted to send response message, but mes- sage was destroyed.	

Error Error detai		details	Error	Meaning	7-
code (Hex)	Upper 2 digits	Lower 2 digits			segment display on Unit
0701	Code from	Error node	Configuration error	A configuration data error occurred.	E8
0702	7-segment display on	address	I/O area overlap	Words in the Slave I/O areas are overlapping.	d0
0703	Onit		I/O area range exceeded	An I/O area is outside the valid areas.	d1
0704			Unsupported Slave	An unsupported Slave is connected.	d2
0705			Verification error: Slave missing	A Slave registered in the scan list is not connected to the network.	d5
0706			Verification error: Slave I/O size differs	The I/O capacity of a Slave does not agree with the information in the scan list.	d5
0707			Communications error	An error occurred in remote I/O communications.	d9
0708			Scan list operation failed	It was not possible to perform a scan list operation.	C0 to C5
0709			PC mounting error	A PC mounting error has occurred.	E4
0781			Node address duplication	The same node address is allocated to two nodes.	F0
0782			Bus Off detected	A Bus Off status was detected.	F1
0783			No communications power supply	The communications power is not being supplied.	E0
0784			Send timeout	A send timeout occurred.	E2

# CS1W-DRM21

Error	Error	Error data		EEPROM
code (Hex)		Upper 2 digits	Lower 2 digits	
0001	CPU Unit WDT Error	00 Hex	00 Hex	Yes
0002	CPU Unit service monitor error (No service from the CPU Unit for a specified time.)	Monitor time (ms)		Yes
0006	Other CPU Unit error	Bit 14: Unit number duplicated. Bit 11: The Unit is not listed in the registered I/O tables.		Yes
		The other bits are reserved by the system.		
000F	CPU Unit initial processing error	00 Hex	00 Hex	Yes
0011	Event timeout	MRC	SRC	Yes
0012	CPU Unit memory error	01 Hex: Read error 02 Hex: Write error	03 Hex: Routing table 04 Hex: CPU Bus Unit Parameter Area 05 Hex: CIO/DM words allocated to CPU Bus Unit	Yes

### Appendix C

Error	Error	Error data		EEPROM
code (Hex)		Upper 2 digits	Lower 2 digits	
0101	No transmission is possible because the Master node is not in the network.	FINS message communications Command		No
0105	No transmission is possible because the node address setting is wrong.	Bit 15: OFF Bit 14 to 8: SNA Bit 7 to 0: SA1		No
0106	No transmission is possible because the node address is duplicated.	Response Bit 15: ON	Bit 7 to 0: SA1 Response Bit 15: ON	
0107	No transmission is possible because the destination node is not in the network.	Bit 14 to 8: DNA Bit 7 to 0: DA1		No
0108	No transmission is possible because there is no Unit corresponding to the unit number.	Explicit message trans	mission	No
0109	No transmission is possible because the destination node is busy.	Bit 15: OFF Bit 14 to 8: 0 Bit 7: ON		No
010B	No transmission is possible because the CPU Unit has an error	Bit 6 to 0: Master noc Response	le address	No
010D	No transmission is possible because the destination address is not registered in the routing table.	Bit 15: ON Bit 14 to 8: 0 Bit 7: ON Bit 6 to 0: Master noc	le address	No
010E	No transmission is possible because the routing tables are not registered.	FINS message commu Command	nications	No
010F	No transmission was possible because the routing tables are wrong.	Bit 15: OFF Bit 14 to 8: SNA Bit 7 to 0: SA1		No
0110	No transmission was possible because the number of relay times is excessive.	Response Bit 15: ON		No
0111	No transmission is possible because the command is in excess of the maximum permissible command length.	Bit 14 to 8: DNA Bit 7 to 0: DA1		No
0112	No transmission is possible because the header is wrong.	Command Bit 15: OFF		No
0117	The packet was discarded because the internal reception buffer was full.	Bit 14 to 8: 0 Bit 7: ON Bit 6 to 0: Master pade address		No
0118	The packet was illegal and discarded.	Bit 6 to 0: Master noc Response Bit 15: ON Bit 14 to 8: 0 Bit 7: ON Bit 6 to 0: Master noc	le address	No
0120	An unexpected routing error.	FINS message commu Command Bit 15: OFF Bit 14 to 8: SNA Bit 7 to 0: SA1	nication	No
0123	The packet was discarded because the internal transmission buffer was full.	Response Bit 15: ON Bit 14 to 8: DNA Bit 7 to 0: DA1		No
0124	Routing failed because the frame length was in excess of the maximum permissible length.	Explicit message comn Command Bit 15: OFF Bit 14 to 8: 0 Bit 7: ON	nunication	No
0125	The packet was discarded due to a response time- out error.	Bit 6 to 0: Master noc Response Bit 15: ON Bit 14 to 8: 0 Bit 7: ON Bit 6 to 0: Master noc	de address de address	No

Error	Error Error data		data	EEPROM
code (Hex)		Upper 2 digits	Lower 2 digits	
021A	Set table logic error	00 Hex	0A Hex: Master scan list 0B Hex: Slave scan list 0C Hex: Message timer list	Yes
0211	Node address overlap	00 Hex	Master node address	No
0300	The packet was discarded because a parameter error occurred.	The same as the error Hex, used for discardir discarding.	code, such as 0101 ng FINS/explicit packet	No
0340	Bus OFF error	00 Hex	00 Hex	No
0341	Network power supply error	00 Hex	00 Hex	No
0342	Transmission timeout	00 Hex	00 Hex	No
0343	Configuration error	01 Hex: I/O area overlap 02 Hex: I/O area range over 03 Hex: Unsupported slave	Slave node address	No
0344	Comparison error	01 Hex: No Slave 02 Hex: Illegal vendor 03 Hex: Illegal device type 04 Hex: Illegal product code 05 Hex: Unsupported connection 06 Hex: I/O size non- conformance 07 Hex: Illegal connection pass	Slave node address	No
0345	Remote I/O communications error	01 Hex: Master function 02 Hex: Slave function	Master function: Slave node address Slave function: Master node address	No
0346	A remote I/O communications error occurred and the remote I/O communications stopped.	01 Hex: Remote I/O communications error 02 Hex: Network power supply error 03 Hex: Transmission timeout error	Remote I/O communications error: Slave node address Power supply error: Master node address Transmission timeout error: Master node address	No
0347	I/O refresh error	01 Hex: Master function 02 Hex: Slave function	00 Hex	No
0348	The message was discarded in order to receive a new request.	The same as the error Hex, used for discardin discarding.	code, such as 0101 ng FINS/explicit packet	No

Error	Error	Error data		EEPROM
code (Hex)		Upper 2 digits	Lower 2 digits	
0601	CPU bus Unit error	Indefinite value		Yes
0602	CPU bus Unit memory error	01 Hex: Read error 02 Hex: Write error	06 Hex: Error history 09 Hex: Identity data 0A Hex: Master scan list (including Master function enable information) 0B Hex: Slave scan list (including Slave function enable information) 0C Hex: Message timer list 0D Hex: Communication cycle time set value	Yes (see note)

**Note** If a memory error occurs for the error history area in the EEPROM, the error history will not be recorded in the EEPROM.

# Appendix D Unit Replacement

By saving device parameter files in the DeviceNet Configurator, the saved parameters can be written to new master and slave devices that may replace present ones in case of device failures. This will save time required for setting up the new devices.

# **Backup Device Parameters**

While the DeviceNet network is in normal operation, use the following procedure to upload online and back up the device parameters of the network devices.

- Connect the DeviceNet Configurator (Ver 2. ) to the DeviceNet network.
- Place the DeviceNet Configurator online, select the device, and then select *Device, Parameter*, and *Upload*.
- Click the Yes Button. The parameters will be uploaded.
- Select the device and then select *Device, Parameter,* and *Save* to save the device parameters as a file (.dvf).

# Writing Device Parameters to New Units

After a device is replaced, use the following procedure to read the device parameters in the file and write the parameters to the new device online.

- Connect the DeviceNet Configurator to the DeviceNet network.
- Select Device, Parameter, Open, to read the device parameters saved in the file.
- Select the device and then select Device, Parameter, and Download.
- Click the Yes Button. The parameters will be downloaded.
- **Note** The device parameters of the CS1W-DRM21 DeviceNet Unit can be backed up in the following methods as well.

Method 1: Writing a Device Parameter File Uploaded by the Configurator to the DeviceNet Unit through a Memory Card

The uploaded device parameter file (.dvf) can be saved as DNnnBKUP.dvf (nn: Unit number in 2 digits in hexadecimal) through the HMC-AP001 Memory Card Adapter in the Memory Card. Then the device parameters can be saved in the DeviceNet Unit by turning ON the Unit Setup File Restore Bit (bit 14 of word n) after the Memory Card is mounted to the CPU.

Method 2: Writing a Device Parameter File Backed Up in a Memory Card to the DeviceNet Unit without Using the Configurator

- Turn ON the Unit Setup File Backup Bit (bit 15 of word n) to save the setup data in the DeviceNet Unit as a Unit setup file in the Memory Card.
- Turn ON the Unit Setup File Restore Bit (bit 14 of word n) to write the Unit setup data file in the Memory Card to the DeviceNet Unit.

Note The data in the Unit setup files is compatible with the device parameter file prepared by the Configurator.

# Appendix E

# Replacing the C200HW-DRM21 on a CS-series CPU Rack with the CS1W-DRM21

With the following procedure, the CS1W-DRM DeviceNet Unit can be used to replace the C200HW-DRM21 DeviceNet Master Unit mounted to a CS-series PC in which the scan list is enabled. The device parameters of the C200HW-DRM21 will be transferred to the CS1W-DRM21.

- **Note** The present I/O allocation information of the C200HW-DRM21 is read regardless of whether it is fixed or used-allocated and written to the CS1W-DRM1. Therefore, no address changes in the ladder program are necessary. The addresses in the status area, however, must be changed in the ladder program according to the CS1W-DRM21 status area. Furthermore, the message communications functions require ladder program changes. Refer to information on differences with the C200HW-DRM21 in the CS1W-DRM21 DeviceNet Unit Operation Manual.
  - Place the present system online without changing the configuration (i.e., without removing the C200HW-DRM21), and then upload the network configuration by selecting **Network** and **Upload** (or import the network configuration file (.ntf) from version 1 of the Configurator by selecting **Network** and **Import**).
  - Save the device parameters of the C200HW-DRM21 in a device parameter file (.dvf).
  - Delete the C200HW-DRM21 from the Network Configuration Window (i.e., the virtual network) in the Configurator.
  - Add the CS1W-DRM21 to the Network Configuration Window.
  - Set the node address CS1W-DRM21 to the previous node address of the C200HW-DRM21.
  - Select the CS1W-DRM21 in the Network Configuration Window, and import the device parameter file (.dvf) for the C200HW-DRM21 by selecting *Device, Parameter,* and *Import*. The CS1W-DRM21's Master function will be enabled automatically.
  - Turn OFF the Programmable Controller and replace the C200HW-DRM21 with the CS1W-DRM21.
  - Turn ON the Programmable Controller.
  - Place version 2 of the Configurator online.
  - Select the CS1W-DRM21 in the Network Configuration Window and set the device parameters to the CS1W-DRM21 by selecting *Device, Parameter,* and *Download*.

The above procedure will transfer the device parameters of the C200HW-DRM21 to the CS1W-DRM21.

The IR, HR, and DM areas allocated by version 1 of the Configurator will be allocated to the same addresses in the CS1. Link Area words LR 00 through LR 63 will be, however, changed to CIO 1000 through CIO 1063.

# Appendix F Replacing the C200HW-DRT21 with the CS1W-DRM21's Slave Functions

With the following procedure, the CS1W-DRM21 with slave functions can be used to replace the C200HW-DRT21 C200H I/O Link Unit mounted to the CS-series PC in a system using the read/write areas for Special I/ O Units or read/write areas set with Configurator version 1.11 or later. The device parameters of the C200HW-DRT21 will be transferred to the CS1W-DRM21 as a slave device.

- Place the present system online without changing the configuration (i.e., without removing the C200HW-DRT21), and then upload the network configuration by selecting **Network** and **Upload** (or import the network configuration file (.ntf) from version 1 of the Configurator by selecting **Network** and **Import**).
- Save the device parameters of the C200HW-DRT21 in a device parameter file (.dvf).
- Delete the C200HW-DRT21 from the Network Configuration Window (i.e., the virtual network) in the Configurator.
- Add the CS1W-DRM21 to the Network Configuration Window.
- Set the node address CS1W-DRM21 to the previous node address of the C200HW-DRT21.
- Select the CS1W-DRM21 in the Network Configuration Window, and import the device parameter file (.dvf) for the C200HW-DRT21 by selecting **Device, Parameter,** and **Import**. The CS1W-DRM21's slave function will be enabled automatically.

Note Select Property and Unit Function to disable the Master function of the CS1W-DRM21.

- Turn OFF the Programmable Controller and replace the C200HW-DRT21 with the CS1W-DRM21.
- Turn ON the Programmable Controller.
- Place version 2 of the Configurator online.
- Select the CS1W-DRM21 in the Network Configuration Window and set the device parameters to the CS1W-DRM21 by selecting *Device, Parameter*, and *Download*.

The above procedure will transfer the device parameters of the C200HW-DRT21 to the CS1W-DRM21 for the slave functions of the CS1W-DRM21.

The IR, HR, and DM areas allocated as the Read/Write areas by version 1 of the Configurator will be allocated to the same addresses in the CS-series CPU Unit. Link Area words LR 00 through LR 63 will be, however, changed to CIO 1000 through CIO 1063 and AR 0 through AR 27 will be allocated to H100 through H127.

**Note** If the TIM/CNT area is specified as a Read/Write Area for the C200HW-DRT21, importing will not be possible.

# Appendix G Dimensions of PCMCIA Card and Unit





# Index

# Α

applications precautions, xiv

# В

baud rate setting through network, 135, 138

# С

C200HW-DRM21 replacing with a CS1W-DRM21, 159 C200HW-DRM21-V1 cycle time, 105, 123 editing parameters, 84 error history, 123, 151 exporting parameters, 140 general tab, 85 I/O allocations, 84, 89 PLC information, 106 settings, 74 status, 123 C200HW-DRT21 replacing with CS1W-DRM21, 161 commands main menu device, 43 edit, 42 EDS files, 43 help, 44 network, 42 options, 43 tools, 43 view, 42 menus, 42 communications cycle time, 8 See also cycle time explicit message See also explicit messages remote I/O See also remote I/O comparing device information, 93 previous versions, 11 components ISA Board, 19 PCMCIA Card, 25 connections online, 108 settings, 8, 76 contents product, 12 CPU Units I/O memory editing memory block, 88 CS1W-DRM21, 152 cycle time, 104, 120 monitoring, 120 editing parameters, 81 error history, 120 exporting parameters, 140

I/O allocations, 82, 88, 103 monitoring, 120 PLC information, 104 replacing a C200HW-DRM21, 159 replacing C200HW-DRT21, 161 selecting SYSMAC CS1 I/F port, 50, 109, 110 settings, 73 status, 120 Unit status, 120 CVM1-DRM21-V1 cycle time, 105, 123 editing parameters, 84 error history, 123, 151 exporting parameters, 140 general tab, 85 I/O allocations, 84, 89 PLC information, 106 settings, 74 status, 123 cycle time, 8, 9, 76 See also CS1W-DRM21 See also CVM1-DRM21-V1 settings, 95, 122 tab page, 95, 122

# D

DeviceNet IF Card, 108 devices adding, 62 classification, 39 commands, 43 comments changing, 65 copying, 64 deleting, 64 editing, 64, 66 I/O information, 101 information comparing, 93 displaying, 93, 100 Property Window, 100 uploading, 63 monitoring, 119 procedure, 119 new vendors adding, 141 pasting, 65 positions changing, 65 Property Window, 100 resetting, 115 See also masters See also slaves settings, 8 changing node addresses, 63 comparing information, 76 types, 39 adding, 141 DIP switch settings ISA Board, 19 display, 7 Driver Setup Software
installation, 20, 28

## Ε

Edit Device Parameters Window downloading device parameters, 112 reading device parameters, 116 verifying device parameters, 118 editing commands, 42 parameters, 61, 75 slaves, 66 EDS files See also files errors error history, 151 monitoring, 120 tab page, 123 error logs, 9 error messages, 145 Expansion Modules adding, 141 explicit messages, 76 canceling, 98 resetting, 9 setting, 9, 98 exporting data, 139 parameters, 140

#### F

files, 7 EDS files, 10 commands, 43 creating, 132 deleting, 133 displaying properties, 134 installing, 131 saving, 133 searching, 134 exporting, 5 importing version 1 Configurator files, 7, 140 manipulating, 129 network configuration files, 10 importing, 140 reading, 130 saving, 130 parameter files, 10 importing, 141 reading, 130 saving, 130 functions, 7 CS1W-DRM21, 73 exporting files, 5 monitoring, 5 optional, 141 other, 5, 10 setup, 5 storage, 5

# Η

hardware management, 10 hardware tree, 39 adding devices to Network Configuration Window, 62 help menu, 44

I

I/O allocations, 8, 72, 89 automatic, 86, 92 manual, 86 See also C200HW-DRM21-V1 See also CS1W-DRM21 See also CVM1-DRM21-V1 See also Parameter Wizard tab page, 86 I/O comments creating, 98 editing, 98 exporting, 139 printing, 135 I/O memory changing allocation areas, 88 I/O port address ISA Board, 19 icons tool bar, 45 indicators slave status, 121 input block changing first address, 88 installation ISA Board, 16 Driver Setup Software, 20, 28 Windows NT 4.0, 18 PCMCIA Card, 24 Driver Setup Software, 20, 28 procedure, 16, 32 interface selecting, 108 DeviceNet IF Card, 108 SYSMAC CS1 I/F Port, 109, 110 selection procedure, 48 settings, 50 **IP** communications IP addresses subnet masks, 57 ISA Board components, 19 installation, 16, 20

#### Μ

main window, 38 hardware tree, 39 Network Configuration Window, 40 masters adding, 62 adding slaves, 70

#### Index

editing parameters, 75 I/O comments exporting, 140 printing, 135 I/O information, 103 monitoring status, 9, 120, 121 parameter files, 7 See also devices setting properties, 73 settings, 8 memory block list, 91 menus main, 42 device, 43 edit, 42 EDS file, 43 help, 44 network, 42 option, 43 tools, 43 view, 42 message timers, 76 settings, 96 models, 3 3G8F5-DRM21 or 3G8E2-DRM21 See also ISA Board See also PCMCIA Card settings, 49 PLC, 74 monitoring, 9 devices, 119 procedure, 119 error logs, 9 functions, 5 master status, 9 setting monitor refresh timer, 119 slave status, 9 MULTIPLE I/O TERMINALs monitoring, 126

## Ν

network commands, 42 configuration files, 7 downloading, 110 importing, 140 reading, 6, 63, 115, 117, 130 saving, 130 connecting, 108 creating virtual network, 62 disconnecting, 110 displaying, 40 information reading, 63, 115 managing, 8 reading network configuration, 63, 115 status display, 46 offline, 7, 59 online, 7 virtual

creating, 62 Network Configuration Window, 40 adding devices, 62 downloading device parameters, 113 reading device parameters, 116 verifying device parameters, 117 node addresses changing, 63 setting, 135, 138

# 0

offline going offline, 110 online connecting, 108 operating environment, 13 precautions, xiii operating system, 7 operation, 37 flowcharts, 59 options commands, 43 output block changing first address, 88 overview, 2, 3

#### Ρ

Parameter Wizard, 76 parameters backing up, 157 DeviceNet parameter files, 7 importing, 141 reading, 131 writing to new Units, 157 downloading, 110, 112 editing, 61, 75, 80 See also Parameter Wizard slaves, 66 exporting, 140 printing, 135 reading, 115, 116 setting with class instances, 135 setting with general-purpose tools, 135 verifying, 115, 117 writing to new Units, 157 PCMCIA Card components, 25 dimensions, 163 installation, 24 precautions, 13 precautions, xi applications, xiv general, xii, 13 operating environment, xiii PCMCIA Card, 13 safety, xii previous versions comparison, 11 printing, 135

# R

registered device list, 91 remote I/O starting, 9, 76, 98, 120 status, 120 stopping, 120

# S

safety precautions, xii settings, 8 advanced device information, 93 baud rate, 135, 138 cycle time, 95 enabling, 115 interface, 50 master properties, 73 message timers, 96 monitor refresh timer, 119 node addresses, 135, 138 slave functions, 97 using general-purpose tools, 135 without EDS files, 135 setup functions, 5 slaves adding, 62, 70 checking device data, 8 deleting, 79, 86 functions, 97 I/O allocations, 8, 92 I/O Information Tab Page, 101 I/O comments, 92, 98 exporting, 140 printing, 135 monitoring status, 9, 120, 121, 127 parameter files, 7 editing, 66 registered device list, 91 registering, 72, 79 I/O allocation, 86 See also devices settings, 8 unregistering, 86 specifications, 4 starting, 38 status bar, 46 display, 46 monitoring, 9, 120 tab page, 120 storage functions, 5 subnet masks address conversion, 57 switches DIP switch settings, 19 SYSMAC CS1 I/F Port, 109, 110 SYSMAC CS1 I/F ports settings, 50, 109, 110

system configuration, 2

# T-U

tab pages cycle time, 95, 122 error history, 123 general, 86 CVM1-DRM21-V1 and C200HW-DRM21-V1, 85 I/O allocation, 86, 89, 92, 101 slave functions, 97 status, 120 Unit status, 121 timers monitor refresh timer setting, 119 tool bar icons, 45 tools commands, 43 troubleshooting, 149 Units dimensions, 163 replacing, 157 status, 121 upgrading, 32 uploading network information, 63

#### V

view commands, 42 virtual network creating, 62

# **Revision History**

A manual revision code appears as a suffix to the catalog number on the front cover of the manual.



- Revision code

The following table outlines the changes made to the manual during each revision. Page numbers refer to the previous version.

Revision code	Date	Revised content
01	September 2000	Original production
02	November 2001	Revised to reflect Configurator version upgrade as outlined in the new section 1-1-5.

#### **OMRON CORPORATION**

FA Systems Division H.Q. 66 Matsumoto Mishima-city, Shizuoka 411-8511 Japan Tel: (81)559-77-9181/Fax: (81)559-77-9045

#### Regional Headquarters

OMRON EUROPE B.V. Wegalaan 67-69, NL-2132 JD Hoofddorp The Netherlands Tel: (31)2356-81-300/Fax: (31)2356-81-388

#### **OMRON ELECTRONICS LLC**

1 East Commerce Drive, Schaumburg, IL 60173 U.S.A. Tel: (1)847-843-7900/Fax: (1)847-843-8568

#### OMRON ASIA PACIFIC PTE. LTD.

83 Clemenceau Avenue, #11-01, UE Square, Singapore 239920 Tel: (65)835-3011/Fax: (65)835-2711

# OMRON

Authorized Distributor:

Cat. No. W382-E1-02

Note: Specifications subject to change without notice. 0675130-4B

Printed in Japan

OMRON

**OPERATION MANUAL** 

Cat. No. W382-E1-02 DeviceNet Configurator Ver. 2.□

AUDIN - 8, avenue de la malle - 51370 Saint Brice Courcelles Tel : 03.26.04.20.21 - Fax : 03.26.04.28.20 - Web : http://www.audin.fr - Email : info@audin.fr